# **Land Use Code**

## Town of Norwood, Colorado

## Prepared By:

**Town of Norwood Planning and Zoning Commission Town of Norwood Board of Trustees** 

With the assistance from Amy Levek, Town Planner Jennifer Russell, Town Attorney and Sandra Stuller, Esq.

Adopted July 11, 1994

Recodified June 14, 1995

**Amended March 1997** 

Amended April 14, 2004

Amended August 10, 2005

**Amended October 2006** 

Amended November 12, 2008

Amended June 9, 2010

Amended December 14, 2011

Amended May 14, 2014

Amended November 12, 2014

Amended December 10, 2014

Amended October 12, 2016

Amended May 10, 2018

Amended September 12, 2018

Amended November 13, 2019

Amended February 5, 2020

## TABLE OF CONTENTS TITLE 7: LAND USE CODE

Chapter 1	.00 GENERAL PROVISIONS	1
§ 1.01	Enactment and Repeals	1
§ 1.02	Purpose	1
§ 1.03	Jurisdiction	
§ 1.04	Consistency with Comprehensive Plan	2
§ 1.05	Municipal Services Outside of Town Boundaries.	2
§ 1.06	Nonconforming Provisions	3
§ 1.07	Enforcement and Penalties	6
§ 1.08	Fees	7
§ 1.09	Severability	7
§ 1.10	Void Permits	7
§ 1.11	General Waiver Authority	
Chapter 2	.00 RULES OF CONSTRUCTION, DEFINITIONS AND INTERPRETATIONS	
§ 2.01	Rules of Construction	9
§ 2.02	Definitions	.10
§ 2.03	Interpretations	
Chapter 3	.00 ZONING DISTRICT REGULATIONS	.28
§ 3.01	Scope and Applicability	.28
§ 3.02	Districts Established	.28
§3.03	Official Zoning Map	.28
§ 3.04	Temporary Zoning of Annexed Territory	.29
§ 3.05	Use Regulations	.31
§ 3.06	Area Regulations	
§ 3.07	A, Agricultural District	.56
§ 3.08	R-1, Residential District	
§ 3.09	MH, Mobile Home District	
§ 3.10	B-1, Business District	.63
§ 3.11	I, Industrial District	.67
§ 3.12	P, Public District	
§ 3.13	PUD, Planned Unit Development Overlay District	.72
§ 3.14	Medium Density Residential District	
Chapter 4	.00 SUBDIVISION STANDARDS	.79
§ 4.01	Scope and Applicability	.79
§ 4.02	Building Lots	
§ 4.03	Streets and Alleys	.80
§ 4.04	Easements	.84
§ 4.05	Park Land Dedication	
§ 4.06	Drainage	
§ 4.07	Water Supply	
§ 4.08	Sanitation	
§ 4.09	Underground Utilities	
§ 4.10	Impact Fees (Reserved)	
§ 4.11	Prominent Site Features	
-	.00 SITE DEVELOPMENT STANDARDS	
§ 5.01	Scope and Applicability	.89

§ 5.02	Parking, loading and access	89
§ 5.03	Sidewalks and Trails	93
§ 5.04	Fences	95
§ 5.05	Landscaping and Screening	96
§ 5.06	Signs	
§ 5.07	Performance Standards	102
§ 5.08	Exterior Lighting Standards	105
Chapter 6	5.00 ADMINISTRATION AND PROCEDURES	110
§ 6.01	Planning and Zoning Commission	110
§ 6.02	Zoning Map and Land Use Code Amendments	112
§ 6.03	Master Plan Amendments.	
§ 6.04	Preliminary Plat Procedures	117
§ 6.05	Final Plat Procedures	122
§ 6.05A	Small Scale Subdivision Procedures	127
§ 6.05B	Condominium Subdivision and Conversion	134
§ 6.06	Acceptance of Subdivision Improvements	137
§ 6.07	Improvements Agreements and Performance Guarantees	138
§ 6.08	Mandatory Homeowners' Associations	139
§ 6.09	Replats and Plat Amendments	
§ 6.10	Conditional Use Permits	146
§ 6.11	Vested Property Rights	149
§ 6.12	Annexations	152
§ 6.13	Board of Adjustments	158
§ 6.14	Appeals	160
§ 6.15	Variances	161
§ 6.16	Special Exceptions	162
§ 6.17	Temporary Use Permits	162
§ 6.18	Sign Permits	163
§ 6.19	Zoning Development Permits	164
§ 6.20	Certificates of Occupancy	164
§ 6.21	Administrative Officials	164
Chapter 7	7.00 RESIDENTIAL USE, GROWING, CULTIVATING AND	166
PROCES	SING OF MARIJUANA	166
§7.01	Findings and Purpose	166
§7.02	Definitions	166
§7.03	Regulations For The Growing, Cultivating, And Processing Of Marijuana In A	
Residenti	al Structure	
§7.04	Inspection; Inspection Warrant	
§7.05	Violations And Enforcement Remedies.	172
APPEND	NX "A"	174

#### TITLE 7

#### LAND USE CODE

## **Chapter 1.00 GENERAL PROVISIONS**

## § 1.01 Enactment and Repeals

- (a) Upon the adoption of this Title on July 11, 1994, as recodified June 14, 1995, the following are hereby repealed in their entirety: Norwood Town Code (1993) Title 7: Zoning Regulations of the Town of Norwood, Colorado: passed and approved with Ordinance 483, Series 1974 on Nov. 6, 1974, together with all amendments thereto; Ordinance #239, Series 1974, creating the Planning and Zoning Commission, passed and approved in 1983, together with all amendments thereto; Town of Norwood Annexation Policies and Requirements Fourth Draft, passed and approved in March, 1983, together with all amendments thereto; and Subdivision Regulations for the Town of Norwood, Colorado, passed and approved in October, 1985 together with all amendments thereto; and any other ordinance or regulation inconsistent with this Code except as this Code is amended from time to time.
- (b) The following is hereby adopted as Title 7: Land Use Code of the Town Code of Norwood, Colorado. It may be referred to herein as "this Title" or the "Land Use Code". It may also be referred to as Title 7, Town Code of Norwood.
- (c) By the adoption of this Title, no presently illegal use shall be deemed to have been legalized unless specifically such use falls within a zoning district where the actual use is a conforming use. Otherwise, such uses shall remain nonconforming uses where recognized, or an illegal use, as the case may be. It is further the intent and declared purpose of this Title that no offense committed, and no liability, penalty or forfeiture, either civil or criminal, incurred prior to the time the existing Zoning Regulations were repealed and this Zoning Ordinance adopted, shall be discharged or affected by such repeal; but prosecutions and suits for such offenses, liability, penalties or forfeitures may be instituted or causes presently pending proceeded with in all respects as if such prior Ordinance had not been repealed.

#### § 1.02 Purpose

The regulations of this Title have been established in accordance with a Comprehensive Plan for the purpose of promoting the health, safety and general welfare of the Town of Norwood. They have been designed to lessen the congestion in the streets, to secure safety from fire, panic and other dangers, to provide adequate light and air, to prevent the overcrowding of land, to avoid undue concentration of population, to promote energy conservation and to facilitate the adequate provision of transportation, water, sewerage, schools, parks and other public requirements. They have been made with reasonable consideration, among other things, for the character of the districts, and their peculiar suitability for the particular uses specified; and with a view of conserving the value of buildings and encouraging the most appropriate use of land throughout the Town consistent with a Comprehensive Plan.

#### § 1.03 Jurisdiction

All development of land within the corporate limits of the Town of Norwood, and all land outside the corporate limits of the Town of Norwood that the Board of Trustees may be petitioned to include within the corporate limits of the Town, by an extension of said corporate limits, shall conform to the following rules and regulations unless specifically exempted herein or by law.

## § 1.04 Consistency with Comprehensive Plan

The regulations of this chapter are intended to implement and be consistent with the Town's official Comprehensive Plan, as such may be amended from time to time. A copy of the plan shall be kept in the office of the Town Clerk with the original ordinance adopting it and it shall be available for public inspection during regular business hours.

## § 1.05 Municipal Services Outside of Town Boundaries.

The provision of municipal services to development outside of the incorporated limits of the Town shall be subject to the property owner's execution of a pre-annexation agreement pursuant to C.R.S. 31-12-121, including but not limited to, the following terms and conditions:

- (a) Use. The use of the area to be served shall be consistent with the adopted Norwood Major Streets and Future Land Use Plan, in harmony with the intent of Town zoning and policies of the Town, and compatible with adjacent neighborhoods;
- (b) Municipal water services outside Urban Growth Boundary. Extension of municipal water service to development outside the Urban Growth Boundary shall be subject to the Rules and Regulations of the Norwood Water Commission.
- (c) Municipal water service outside of Town and inside Urban Growth Boundary. Extension of municipal water service to development inside the Urban Growth Boundary shall be subject to the Rules and Regulations of the Norwood Water Commission, and all land uses shall be consistent with the Norwood Major Streets and Future Land Use Plan, and the property owners execution of a preannexation agreement.
- (d) Non-complying uses and substandard-sized parcels. Municipal water service and any other municipal services shall not be extended to lots or parcels or uses that are not consistent with the Comprehensive Plan and in compliance with the Norwood Major Streets and Future Land Use Plan as implemented by Sec. 3.05, Use regulations, and Sec. 3.06, Area regulations, of this Chapter.
- **Consent to Annexation**. As a condition precedent to the provision of municipal services by the Town to land within the Urban Growth Boundary, the Town may require that the owners of such land apply for or consent to the annexation of such land to the Town at such future date as such land becomes eligible for annexation.

## § 1.06 Nonconforming Provisions

- (a) Nonconforming uses and structures.
  - (1) Nonconforming status. The use of land, use of a structure, or a structure itself shall be deemed to have nonconforming status when each of the following conditions are satisfied:
    - a. The use or structure does not conform to the regulations prescribed in the district in which such use or structure is located and was in existence and lawfully constructed, located and operating prior to, and at the time of, the event that made such use or structure nonconforming.
    - b. The event that made such use or structure nonconforming was one of the following: annexation into the Town of Norwood, adoption of this Title or a previous zoning ordinance, or amendment of this Title or a previous zoning ordinance.
    - c. The nonconforming use or the use occupying the nonconforming structure has been operating since the time that the use or structure first became nonconforming without abandonment, as abandonment is defined in subsection (a)(4) below.
  - (2) Expansion. No nonconforming use may be expanded or increased.
  - (3) Change of use.
    - a. Any nonconforming use may be changed to a conforming use and once such change is made, the use shall not thereafter be changed back to a nonconforming use.
    - b. Where a conforming use is located in a nonconforming structure, the use may be changed to another conforming use by securing a Zoning Development Permit.
    - c. A change from one nonconforming use to another nonconforming use may be made by securing a Zoning Development Permit provided such change is to a permitted use in a more restrictive zoning district classification. For the purpose of interpreting these provisions, the zoning districts shall be considered to be arranged in a hierarchy as they are listed in Sec. 3.02, Districts established, from the most restrictive A, Agricultural District to the least restrictive I, Industrial. The P and PUD Districts shall not be considered part of the hierarchy. In addition, for the purpose of interpreting this paragraph (c), a use that is authorized in a district with a conditional use permit shall not be considered a permitted use in such district.
  - (4) Abandonment. Whenever a nonconforming use or a conforming use in a nonconforming structure is abandoned, all nonconforming rights shall cease and the use of the premises shall henceforth conform to this Title. Abandonment shall involve the actual act of discontinuance, regardless of the intent of the user or owner to discontinue a nonconforming operation. Any nonconforming use that is discontinued for, or that

remains vacant for a period of six (6) months, shall be considered to have been abandoned. Any nonconforming use that is moved from the premises shall be considered to have been abandoned.

(5) Destruction. If a nonconforming structure or a structure occupied by a nonconforming use is destroyed by fire, the elements or other cause, it may not be rebuilt except to conform to the provisions of this Title. In the case of partial destruction of a structure occupied by a nonconforming use not exceeding sixty percent (60%) of its reasonable value, reconstruction may be permitted after a hearing and favorable action by the Board of Adjustment, but the size and function of the nonconforming use shall not be expanded.

## **(b)** Nonconforming lots.

- (1) General. A single family dwelling and customary accessory buildings may be developed on a lot that has less area than the minimum required by the underlying zone district and was an official "lot of record" prior to the adoption of the Town's original Zoning Ordinance (Nov. 6, 1974), if:
  - a. The "lot of record" is in separate ownership or contiguous to lots in the same ownership; and the proposed single family dwelling can be located on the lot so that the yard setback, height, and other dimensional requirements of the underlying zone district can be met, or a variance is obtained from said dimensional requirements pursuant to Sec. 6.16 of this Title.
  - b. Undivided lot. If two or more lots or combinations of contiguous lots in a single ownership (including husband and wife as, in all cases, a single owner) are of record at the effective date of the adoption or amendment of this Title, regardless of time of acquisition, or all or parts of the lots do not meet the requirements established for lot width and area, the lots shall be considered an undivided parcel, and no portion shall be used or occupied which does not meet the width and area requirements of this Title.

#### (2) Lot Reduction.

- a. No lot or interest therein shall be transferred, conveyed, sold or subdivided so as to create a new nonconforming lot, to avoid, circumvent or subvert any provision of this Title, or to leave remaining any lot in violation of the dimensional requirements of this Title.
- b. No lot or portion of a lot required as a building site under this Title shall be used as a portion of a lot required as a site for another structure.
- c. No building permit shall be issued for any lot or parcel of land which has been conveyed, sold, or subdivided in violation of this subsection. Any transferee who acquires a lot in violation of this subsection without knowledge of such violation, and any subsequent transferee, shall have the right pursuant to Colorado law to rescind and/or receive damages from any transferor who violates the provisions of this paragraph.

(c) Nonconforming signs. Any sign that is permitted to remain in place as a nonconforming use may be continued in use until the sign, or a substantial part of it, is blown down or otherwise destroyed or dismantled for any purpose other than maintenance operations or for changing the letters, symbols, or other matter on the sign. A sign, or a substantial part of it, shall be considered to have been destroyed or dismantled if the cost of repairing the sign is more than sixty percent (60%) of the cost of erecting a new sign of the same type at the same location.

- (d) Nonconforming mobile homes and mobile home sites. An individual mobile home in a mobile home park or mobile home subdivision shall not be replaced unless the individual mobile home site is in compliance with the following provisions:
  - (1) Each mobile home site in mobile home parks and mobile home subdivisions shall be clearly designated and arranged so that all mobile home have access to the internal driveway.
  - (2) There shall be a minimum of twenty feet (20') between mobile homes.
  - (3) Each trailer site shall be improved with a four inch (4") concrete or gravel pad. No trailer with a footprint larger than the pad shall be placed on site.
  - (4) Exposed ground surfaces in all part of a mobile home park or mobile home subdivision shall be paved or covered with stone screening or other solid material, or protected with a vegetative growth that is capable of preventing soil erosion and of eliminating objectionable dust.
  - (5) A detached, accessory storage building or buildings shall be provided within a mobile home park or mobile subdivision containing a minimum storage capacity of ten (10) cubic yards per mobile home site.
  - (6) Mobile home dwelling units shall be "skirted" within (60) sixty days of its placement in a Mobile Home Park or Subdivision by affixing thereto of a solid, nonporous screening or skirt between the underside of the dwelling unit at its outer edge around ground level.
- (e) Nonconforming Formula Business Establishment. No businesses that exist within the Town as of the February 5, 2020 meet the definition of a Formula Business Establishment, and furthermore, if it could be construed that any business that exists within the Town as of Norwood did meet the definition of a Formula Business, any such business in operation prior to February 5, 2020 is exempt from these regulations, and may relocate, expand, sell or otherwise modify same exempt from these regulations and otherwise in compliance with the Land Use Ordinance.

#### § 1.07 Enforcement and Penalties

(a) Any person, firm or corporation who shall violate any of the provisions of this Title or who shall fail to comply with any provisions hereof within the corporate limits of the Town of Norwood shall be guilty of a misdemeanor and upon conviction shall be subject to a fine. Any person violating any of the provisions of this Title shall be fined upon conviction not less than one dollar (\$1.00) nor more than one thousand dollars (\$1,000.00), or imprisoned for a period of up to ninety (90) days, or both such fine and imprisonment. Each day any violation or noncompliance continues shall constitute a separate and distinct offense.

- (b) Any person, being the owner or agent of the owner of any land located within a subdivision, who transfers or sells, agrees to sell, or negotiates to sell any land by reference to or exhibition of or by use of a plat of a subdivision before such plat has been approved by the Board of Trustees and recorded or filed in the office of the San Miguel County Clerk shall pay a penalty of one hundred dollars to the Town for each lot or parcel so transferred, or sold, or agreed or negotiated to be sold.
- (c) The penalties provided herein shall be cumulative of other remedies provided by state law as provided in Colorado Revised Statutes 31-23-216.5 or 31-23-308 and the power of injunction may be exercised in enforcing this Title whether or not there has been a criminal complaint filed.
- (d) In addition to the penalty provided for, the right is hereby conferred and extended to any property owner owning property in any district where such property may be affected by a violation of the terms of this Title, to bring suit in such court or courts having jurisdiction thereof and obtain such remedies as may be available at law or in equity for the protection of the rights of such property owners.
- (e) When it is determined that there has been a violation of any provision of the Land Use Code, written legal notice of violation shall be served in the following manner:
  - (1) Determine and include a list of violations, refer to the section or sections of the Code violated;
  - (2) Determine and specify a time for compliance with relevant Land Use Code provisions thirty (30) days from the service of the notice; and
  - (3) Serve the notice on the owner, occupant, operator, lessee, agent or other responsible party in person, provided that such notice and requirement shall be deemed to be properly served on such responsible party if a copy thereof is delivered to, posted on, or sent by registered or certified mail to his/her last known mailing address, residence or place of business.

#### § 1.08 Fees

(a) Fees for the processing of land use applications for proposed developments shall be set by ordinance of the Board of Trustees commensurate with the level of service. Such fees may include all costs occasioned to the Town, including publication of notices, public hearing and review costs, planning, engineering, legal and other professional review costs.

(b) No person or entity owing money to the Town, in any amount or for any purpose, including delinquent taxes certified by the County Treasurer or any land use application fees, may be granted any development permit or any other development approval and the Town and any of its boards, commissions, departments, officers or agents will take no action on a zoning development permit or other land use application until all monies owed the Town by an applicant are paid. This provision shall not prohibit the Town or any of its designees from conducting a pre-application conference or determining application completeness.

## § 1.09 Severability

It is hereby declared to be the intention of Board of Trustees of the Town of Norwood that the sections, paragraphs, sentences, clauses and phrases of this Title are severable, and if any phrase, clause, sentence, paragraph or section of this Title shall be declared unconstitutional or invalid for any reason, such unconstitutionality or invalidity shall not affect any of the remaining phrases, clauses, sentences, paragraphs or sections of this Title, since the same would have been enacted by the Board of Trustees without the incorporation in this Title of any such unconstitutional or invalid phrase, clause, sentence, paragraph or section.

#### § 1.10 Void Permits

All persons are presumed to know the terms and requirements of this Land Use Code and the extent of the legal authority of the Town and its employees, boards, and commissions to issue development approvals or permits. Any permit or approval issued in error or otherwise not in conformity with the requirements of this Code shall be void. Similarly, any permit or approval issued in reliance upon, or as a result of, a materially false statement or representation made in the process of obtaining the permit or development approval shall be void. Any person having received a void or voidable permit or approval shall not be relieved from having to comply with all applicable terms and conditions of this Code and the Town shall not be stopped from fully enforcing the same.

## § 1.11 General Waiver Authority

The Board of Trustees may on its own motion, or upon request of an applicant, a reviewing body, or the Planner, and after prior public notice and hearing, waive any substantive or procedural requirement of this Title as applied to a specific application.

- (a) A waiver request may be approved only if the Board of Trustees finds that the waiver:
  - (1) Will substantially promote the public health, safety of welfare; and

(2) Is consistent with the goals and objectives of the Norwood Major Streets Plan and Future Land Use Plan and the purposes of the Land Use Code; and

- (3) Will not cause substantial injury or harm to adjoining property owners and residents of the neighborhood which may be affected by the waiver; and
- (4) Is justified because compliance with a requirement would cause undue hardship to the applicant and serve no useful public purpose.
- (5) The waiver request has been reviewed by the Planner and the Planning and Zoning Commission, and a recommendation thereon has been provided.
- (b) A waiver shall be approved by resolution and only upon an affirmative vote of an extraordinary majority of the Board of Trustees (i.e., four (4) Trustees), except that in the event of disqualification of one (1) or more Trustees from acting on the request, the number of votes necessary to grant the waiver shall be reduced by the same number.

#### (c) Procedures

- (1) Review by Planning and Zoning Commission. A public hearing is required. The Planning and Zoning Commission shall hold a public hearing on any waiver prior to making its recommendation to the Board of Trustees. The Town Clerk shall cause notice of the public hearing of the Planning and Zoning Commission to be given by one publication in a newspaper of general circulation in the Town of Norwood without the necessity of notifying property owners by mail. Such notice shall state the time and place of such hearing and the nature of the subject to be considered, which time shall not be earlier than fifteen (15) days from the date of publication.
- (2) Action by Board of Trustees. A public hearing shall be held by the governing body before deciding whether or not to waive a provision of the Land Use Code. The Town Clerk shall cause notice of the public hearing of the Board of Trustees to be given by publication in a newspaper of general circulation in the Town of Norwood without the necessity of notifying property owners by mail. Such notice shall state the time and place of such hearing and the nature of the waiver to be considered, which time shall not be earlier than fifteen (15) days from the date of publication.

## Chapter 2.00 RULES OF CONSTRUCTION, DEFINITIONS AND INTERPRETATIONS

#### § 2.01 Rules of Construction

- (a) Meaning and intent. All provisions, terms, phrases and expressions contained in this chapter shall be liberally construed in order to accomplish the purposes declared in subsection 1.02.
- **Text**. In case of any difference of meaning or implication between the text of this Zoning Code and any illustration or figure, the text shall control.
- (c) Computation of time. The time within which an act is to be done shall be computed by excluding the first and including the last day; if the last day is a Saturday, Sunday or legal holiday declared by the Town, that day shall be excluded. In the computation of time for public hearing notice, both the first day (day of the advertisement) and the last day (day of the hearing) shall be excluded. The following time-related words shall have the meanings ascribed below.
  - (1) "Day" means a calendar day unless working day is specified.
  - (2) "Week" means seven (7) calendar days.
  - (3) "Month" means a calendar month.
  - (4) "Year" means a calendar year, unless a fiscal year is indicated.
- (d) Conjunctions. Unless the context clearly indicates to the contrary, conjunctions shall be interpreted as follows:
  - (1) "And" indicates that all connected items, conditions, provisions or events shall apply; and
  - (2) "Or" indicates that one or more of the connected items, conditions, provisions or events shall apply.
- **(e) Delegation of authority**. Whenever a provision appears requiring the head of a department or some other officer or employee to do some act or perform some duty, it is to be construed to authorize the head of the department or other officer to designate, delegate and authorize professional-level subordinates to perform the required act or duty unless the terms of the provision or section specify otherwise.
- (f) Non-technical and technical words. Words and phrases shall be construed according to the common and approved usage of the language, but technical words and phrases and such others as may have acquired a peculiar and appropriate meaning in law shall be construed and understood according to such meaning.
- **(g) Public officials, bodies and agencies**. All public officials, bodies, and agencies to which reference is made are those of the Town of Norwood, Colorado, unless otherwise indicated.
- **(h) Mandatory and discretionary terms**. The word "shall" is always mandatory. The word "may" is permissive.

(i) Tense, number and gender. Words used in the past or present tense include the future as well as the past or present, unless the context clearly indicates the contrary. The singular number shall include the plural and the plural shall include the singular, as the context and application of this Code may reasonably suggest. Words of one gender shall apply to persons, natural or fictitious, regardless of gender as the context and application of this Code may reasonably suggest.

## § 2.02 Definitions

Words defined in this section shall be given the meanings set forth here. All other words shall be given their common, ordinary meanings, as the context may reasonably suggest. In case of dispute over the meaning of a term not defined here or over the application of a definition set forth here, the Town Planner shall give a written interpretation in accordance with Chapter 2.03.

Accessory Use or Structure: A use or structure naturally and normally incidental to, subordinate to, and devoted primarily to the principal use or structure of the premises. See; Sec. 3.05(d)(1)

**Agriculture:** The use of land for the production of food and fiber, including the growing of crops and/or the grazing of animals.

**Antique Shop:** An establishment offering for sale, within a building, articles such as glass, china, furniture or similar furnishings and decorations that have value and significance as a result of age, design or sentiment.

**Alley:** A space or thoroughfare that affords only secondary means of access to property abutting thereon that has been dedicated or deeded to the public for public use.

**Alley House**: A separate dwelling unit located on a lot which includes a single family dwelling unit or double wide manufactured/mobile home. The alley house is accessory to the primary residential use on the property.

**Amusement, Commercial (Indoor):** An amusement enterprise wholly enclosed in a building offering entertainment or games of skill to the general public for a fee or charge and including, but not limited to, a bowling alley or billiard parlor.

**Asphalt or Concrete Batching Plant, Temporary:** A temporary facility for producing asphalt or concrete products used in construction activities on the same or nearby sites.

**Auto Repair Garage:** A building or place arranged, designed, used or intended to be used for the purpose of providing general repair and servicing of motor vehicles. Such repair or servicing may include reconditioning of engines, air conditioning systems and transmissions; wrecker service; collision services including body, frame or fender straightening or repair; painting, undercoating and rust proofing; replacement or repair of brakes, shock absorbers, tires, batteries, mufflers, or upholstery; and other similar services.

**Basement:** A building story having at least one-half (1/2) of its height below the average level of the adjoining ground. A basement shall not be counted as a story in computing building height.

**Bed and Breakfast:** An owner occupied, single-family dwelling unit converted into an establishment that provides lodging and breakfast for temporary overnight occupants in no more than five separate bedrooms for compensation. In addition to residential off-street parking requirements, a bed and breakfast shall provide one off-street parking space per bedroom offered for use for temporary overnight accommodations.

**Berm:** Mounded or elongated landscape hills constructed for many reasons such as blocking out unwanted or unsightly views, directing or redirecting foot traffic or drainage, creating subtle and natural-looking privacy, adding raised elements to the garden, or simply emphasizing a particular area or focal point.

**Billboard:** A large\_outdoor advertising sign placed along a right-of-way for which rent is charged that directs attention to a business, commodity, service offered existing elsewhere than upon the property where the billboard is located.

**Block:** An area enclosed by streets and occupied by or intended for buildings; or if said word is used as a term of measurement, it shall mean the distance along a side of a street between the nearest two (2) streets that intersect said street.

**Board of Adjustment:** The Town of Norwood Board of Adjustment as established in Chapter 6.13 of the Land Use Code.

Board of Trustees: The governing and legislative body of the Town of Norwood, Colorado.

**Boarding House or Rooming House:** A building, other than a hotel or motel, where lodging and/or meals for three (3) or more persons are provided for compensation.

Bridle Path: See "Trail."

**Building:** Any structure built for the support, shelter and enclosure of persons, animals, chattels or movable property of any kind. When subdivided in a manner sufficient to prevent the spread of fire, each portion so subdivided may be deemed a separate building. Building includes yurts, removable sheds, and similar uses, but does not include signs or fences.

**Building Material Yard and Construction Office, Temporary:** A yard or building for storing or housing building material and building equipment in conjunction with an approved construction or development project on a temporary basis.

**Building Official:** The Building Inspector charged with the responsibility of issuing permits and enforcing on behalf of the Town the International Building Code, or other building code adopted by the Town, and Land Use Code, including any amendments, or that person's designee.

**Building Permit:** A permit issued by the Building Official, after approval of a Zoning Development Permit, that allows a developer to erect, construct, reconstruct, excavate for a foundation, alter or change the use of a building or other structure or improvements of land.

**Campground**: A tract of land designed for use by temporary or short term occupancy for travel, recreation or vacation.

**Caretaker or Guard Residence:** Dwelling facilities located on a premise occupied by a permitted main use for the housing of persons and their families who are employed on the premises as guards, caretakers or in similar custodial capacity.

**Car Wash:** A facility designed and used for the primary purpose of washing and cleaning passenger automobiles.

Cellar: See "Basement."

**Certificate of Occupancy:** An official certificate issued by the Town through the enforcing official that indicates conformance with or approved conditional waiver from the zoning regulations and authorizes legal use of the premises for which it is issued.

**Character (of a developed area):** The density, height, coverage, setback, massing, fenestration, materials and scale of materials.

**Church or place of worship:** A site used primarily or exclusively for religious worship and related religious services operated by a bona fide religious group for religious activities.

**Cleaning Shop or Laundry Pick-up Station:** A custom laundry cleaning or pressing shop a pick-up station for laundry or cleaning where the work is performed other than on the premises.

**Clustering**: The act of consolidating and closely grouping development within a parcel for the purpose of protecting and/or preserving common open space areas, environmentally sensitive lands, open meadows, scenic vistas and wetlands, and to encourage the efficient provision of utility facilities and public transportation.

**Commercial Core:** Commercial Core means the Grand Avenue area of the B-1, Business District between Market Street and Pine Street.

**Commercial Development:** Includes, but is not limited to: expansion or construction of office, retail, wholesale, warehouse, manufacture, commercial recreation, restaurant/bar and/or service-commercial operations by the addition of square footage.

**Common Open Space:** A parcel of usable land, area of water, or a combination of land and water within the site designated for a planned unit development designated and intended primarily for the use or enjoyment of residents, occupants and owners of the planned unit development. Common open space may include, but is not limited to, areas devoted to recreation, courts, gardens, parks and walkways. The term shall not include space devoted to streets, parking and loading areas.

**Community Center:** A building and grounds operated by a governmental entity or non-profit corporation, for the social, recreational health or welfare of the community served.

**Compatible:** Consistent with, harmonious with, similar to and/or enhancing the mixture of uses, siting and/or complimentary architectural styles, either of an individual structure or the character of the surrounding structures.

**Comprehensive Plan:** The comprehensive plan of the Town and adjoining areas adopted by the Planning and Zoning Commission and Board of Trustees, including all of its revisions. The plan indicates the general locations recommended for various land uses, transportation routes, public and private buildings, streets, parks and other public and private developments and improvements.

**Condominium, Attached:** A non-residential or residential dwelling unit in a structure containing two or more units, the spaces of which are individually owned; the balance of the property (both land and building) is owned either in common by the owners of the individual units or by an association consisting of such owners.

**Condominium, Detached:** One of at least two individually owned, unconnected non-residential or residential dwelling units located on property owned either in common by the owners of such units or by an association consisting of such owners.

**Contiguous:** Sharing a common border at more than a single point of intersection and in such a manner that the shared boundaries are touching and not separated except by boundaries of public or private right-of-ways, watercourses or water bodies, running parallel and between the shared boundaries. Contiguity is not the mere touching of points at intersections.

**Convenience Store:** A retail establishment primarily offering for sale meats, fruits, vegetables, bakery products, dairy products, personal care items, cleaning products and similar household items to a localized or neighborhood market, for off-premise consumption.

**Cottage Dwelling Units:** A detached dwelling unit not greater than 850 square feet in lot coverage and a maximum floor area of 1500 square feet or 1.5 times the lot coverage, whichever is less, constructed as part of a comprehensive site plan in a single application for development.

**Cottage Housing Development:** A comprehensive development consisting of cottages on a single lot. The site plan includes individual dwelling units clustered around a shared court yard, off street parking and access. The cluster includes a minimum of four and maximum of nine units included in a single application. Cottage housing development will be accomplished through a condominium subdivision and include covenants.

**Cul-de-sac:** A street that terminates at one (1) end in a vehicular turnaround.

**Custom Personal Services:** Barber, beauty, tailor, dressmaker, footwear fitting or repair, fitness, or similar business offering custom services.

**Day Camp for Children:** A facility arranged and conducted for the organized recreation and instruction of children, including outdoor activities, on a daytime basis only.

Day Care Center: A facility licensed by the Colorado Department of Social Services (defined by that

department as "Child Care Center") that is maintained for the whole or part of a day for the care of five or more children under the age of 16 years who are not related to the owner, operator or manager thereof, whether such facility is operated with or without compensation for such care, and with or without stated educational purposes. A day care center must provide Town and State approved water and sewer systems and shall provide one off-street parking space per employee, a child loading/unloading area of adequate dimensions (preferably off-street) and adequately sized indoor and outdoor play areas.

**Day Care Home:** An accessory use facility licensed by the Colorado Department of Social Services that provides care for two or more children not related to each other, or for children from more than one family, not related to the care provider. Care shall be provided for each child for more than two consecutive full days (seven or more hours each) on a regular weekly basis. If established in conjunction with a residential use, a day care home shall meet the requirements of a home occupation. If established in conjunction with an institution or business, its use shall be limited to children of employees or guests of that institution or business and shall provide two off-street parking spaces.

**Decibel:** A unit of measurement of sound pressure.

**Dedication:** The turning over by an owner or developer of private land for public use, and the acceptance of land for such use by the governmental agency having jurisdiction over the public function for which it will be used. Dedications for roads, parks, school sites, or other public uses often constitute conditions for approval of a development.

**Demolition:** The act of demolishing a structure.

**Density:** The maximum number of dwelling units per gross acre of land permitted in a zone district, as set forth in Chapter 3.00.

**Development:** The physical extension and/or construction of land uses. Development activities include: subdivision of land; change in the intensity of use of land; construction, reconstruction, demolition or partial demolition or alteration of buildings, roads, utilities, and other facilities; commencement of drilling, mining, or excavation; grading; deposit of refuse, debris, or fill materials generated offsite; and clearing of natural vegetation cover.

**Developer:** Any public or private person, partnership, association or agency that prepares land for the construction of buildings or causes to be built physical building space for use primarily by others, during which preparation the land or the creation of the building space is in itself a business and is not incidental to another business or activity.

**Development Permit:** See "Zoning Development Permit."

**District:** A section of the Town of Norwood for which the regulations governing the height, area or use of the land and buildings are uniform.

**Driveway:** An area providing vehicular access between a street and an off-street parking or loading area.

**Dwelling unit:** A structure or portion of a structure, other than a mobile home, that is designed, occupied or intended to be occupied as living quarters and includes facilities for cooking, sleeping and sanitation.

**Duplex dwelling:** The use of a lot for two dwelling units, other than mobile homes, within a single building and under a single roof.

**Single-family dwelling:** The use of a lot for one dwelling unit that has no physical connection to a building located on any other lot.

**Multi-family dwelling:** The use of a lot for three or more dwelling units within a single building.

**Easement:** A property right given by an owner of land to another person or entity for specific limited use of that land.

**Educational Facilities** buildings, fixtures, and indoor or outdoor areas and equipment necessary for the effective and efficient operations of the programs of any publicly or privately operated primary, secondary or post-secondary institution of learning duly recognized and declared as such by the appropriate authority of the state but shall not include any institution whose offered programs are primarily vocational in nature.

**Engineer:** A person duly authorized under applicable provisions of Colorado Revised Statutes to practice the profession of engineering.

**Equestrian Center:** A recreational facility, the primary activity of which is the business of recreational horseback riding and instruction, boarding of horses and competitive riding activities.

**Essential Services:** The development or maintenance of Town approved public utilities, including but not limited to, underground, surface or overhead gas, electrical, communication, fuel or water transmission or distribution systems, including towers, poles, wires, mains, drains, sewers, pipes, conduits, cables, hydrants and similar equipment.

**Events Center:** A building or complex of buildings housing community recreation facilities owned, operated or leased for operation by a non-profit organization or governmental entity and may include equestrian facilities, swimming pools, tennis and other indoor or outdoor athletic facilities.

**Excavation:** The formation of a cavity formed by cutting, digging or scooping earth.

**Family:** Two or more persons related by blood or marriage, or between whom there is a legally recognized relationship, or not more than 5 unrelated persons occupying the same dwelling unit.

**Fence:** Any structure intended for the use of confinement, to control access, boundary identification, or screening of an activity encircling wholly or any portion of an area.

Field Office, Temporary: A structure or shelter used in connection with an approved development or

building project for housing on the site of temporary administrative and supervisory functions and for sheltering employees and equipment.

**Finished Grade:** The elevation of the ground surface, following development, next to the completed walls of a structure, prior to placement of any fill material.

Floor Area, Gross Floor Area: The sum of the gross horizontal areas of all floors of a building measured from the exterior faces of the exterior walls or from the centerline of walls separating buildings, plus the floor area projected over stair and elevator openings. Covered and enclosed areas, such as but not limited to covered decks and balconies, shall be included in calculating floor area. Floor area that is more than seventy five percent below grade shall not be considered in gross floor area, but shall be used to calculate allowable uses, parking and other mitigation.

Floor Area Ratio: The gross floor area of a building on a lot, divided by the lot area.

**Formula Business Establishment** means a business which is required by contractual or other arrangement: (1) to conform generally to a certain business model with standardized features that may include exterior façade, architecture, layout, signage, array of services, array of merchandise, trademarks, logos, servicemarks, symbols, color scheme, interior decor, uniforms, and/or centralized corporate control with minimal local autonomy; **and** (2) which causes it to be substantially identical to more than five other businesses regardless of ownership or location. Formula businesses can include, but are not limited to restaurants, retail stores, banks, real estate sales offices, spas, hair and nail salons, and hotel/motel/inn/B&B. For purposes of this definition, the following also apply:

- (i) Array of merchandise means fifty percent (50%) or more of the in-stock merchandise is from a single manufacturer or distributor bearing uniform markings, AND/OR the type, arrangement and pricing of the inventory is generally standardized.
- (ii) Array of services means a substantially similar set of services or food and beverage menus that are priced, prepared and performed in a consistent manner that distinguishes the source of a service of one party from those of others.
- (iii) Color scheme means the selection of colors used throughout the business, such as on the walls, furnishings, permanent fixtures or on the building façade.
- (iv) Décor means the interior design and furnishings that may include style of furnishings, shelving, display shelving/racks, wall coverings or other permanent fixtures.
- (v) Façade means the principal exterior face or front of a building, including awnings that are oriented towards a public area (including without limitation a street, alley or parking lot).
- (vi) Servicemark means a word, phrase, symbol or design or a combination of words, phrases, symbols or designs that identifies and distinguishes the source of a service of one party from those of others.
- (vii) Standardized means there is a basis for comparison against which items can be evaluated for similarity in order to assure consistency and regularity; to conform to a standard set by the business.
- (viii) Substantially identical means identical in most fundamental and material aspects. For example, a sign may be substantially identical to another if the color, font, and style are fundamentally identical even if the size is not identical; or a building may be substantially identical to another if the general construction methods, materials and architecture are fundamentally identical even if the exterior façade materials and/or size of the structure is not identical.
- (ix) Trademark means a word, phrase, symbol or design, or a combination of words, phrases, symbols or designs legally registered or established by use that distinguishes the source of the product of one party from those of others.

(x) Uniform apparel means standardized items of clothing, including but not limited to standardized aprons, pants, shirts, vests, smocks or dresses, and hat and pins (other than name tags) with standardized colors and fabrics.

**Frequency:** The number of times per second a vibration or sound wave oscillates.

**Free-standing Sign:** Any sign permanently affixed to the ground, supported by a standard or legs or other self-supporting structure used solely for that sign, and physically separated from any building or structure.

**Gasoline Service Station:** A place of business for the retail sales of gasoline, which may or may not exist in conjunction with an auto repair shop, based upon specific zoning requirements.

**Gross Area:** The total area of a site minus any dedicated rights-of-way and surface easements.

**Groundwater:** Subsurface water within and below the zone of continuous saturation.

**Group Home:** A facility providing 24-hour care in a protected living arrangement for not more than eight (8) residents per home sixty years of age or older, or for the developmentally disabled, limited to cerebral palsy, multiple sclerosis, mental retardation, autism, and epilepsy and not more than two (2) supervisory personnel. Such homes for the developmentally disabled must be state-licensed.

Height: The maximum possible upward distance to the top of a building, measured adjacent to a building at a right angle to the horizon line from each and every point on the finished grade, except that:

(1) The ridge of a gable, hip, gambrel or similar pitched roof may extend up to five feet above the specified maximum height limit; (2) Antennas, chimneys, flues, vents or similar structures may extend up to ten feet above the specified maximum height limit, and radio transmission antennas may extend above the specified maximum limit with the approval of the Planning and Zoning Commission; (3) Water towers and mechanical equipment may extend up to five feet above the specified maximum height limit; (4) Church spires, bell towers and like architectural features when associated with a church; (5) flag poles, may extend over the specified maximum height limit; (6) Schools, churches and public administration buildings in residential zone districts may exceed the established height limitation by fifty percent, provided they comply with all other requirements for the zone district and that the total floor area of the structure does not exceed the total area of the lot on which the building is located. See also Fences.

**Heliport:** A landing facility for rotary wing aircraft used for public purposes only.

**Home Occupation:** A business, occupation or trade conducted entirely within a residential building or accessory structure for gain or support by a resident of the dwelling as an accessory use to a dwelling unit subject to the conditions of  $\S 3.06 \text{ (d)}(5)$ .

**Hospital or Clinic:** A public or private facility that: (1) provides in- and out-patient care, medical and/or dental, ward, surgery, laboratory, physical therapy, diagnostic, radiology, emergency medical and/or pharmacy services; (2) may contain administration offices, gift shop, waiting rooms, kitchen (to service hospital patients, staff and visitors), and similar uses such as health service and prepaid health

maintenance offices, ambulance garage, heliport.

**Hotel or Motel:** A building or group of buildings designed and occupied as a temporary lodging place of individuals. To be classified as a hotel or motel, an establishment shall contain a minimum of six (6) individual guest rooms or units and shall furnish customary hotel or motel services.

**Impact:** The effect of any direct manmade actions or indirect repercussions of manmade actions on existing physical, social, or economic conditions.

**Impervious Surface or Cover:** Material placed over the surface of the ground (e.g. pavement, sidewalks, roofs, driveways) that reduces the infiltration of precipitation into the ground.

**Kennel:** Any lot, building, structure, enclosure or premises whereupon or wherein are kept seven (7) or more dogs, cats or similar small animals in any combination, whether such keeping is for pleasure, profit, breeding, or exhibiting, and including places where dogs or cats or similar small animals in any combination are boarded, pet for sale or kept for hire.

**Landscaping:** May include trees, shrubs, ground cover, vines, walkways, ponds, fountains, sculpture, and other organic and inorganic materials used for creating an attractive appearance. Smooth concrete or asphalt surfaces are not considered landscaping.

**Library:** A publicly operated facility housing a collection of books, magazines, audio and video tapes, or other material for use by the general public.

**Lot:** An undivided tract or parcel of land under one ownership having frontage on a public street and either occupied or to be occupied by a building or building group together with accessory buildings, which parcel of land is designated as a separate and distinct tract.

**Lot Area:** The net area of the lot, excluding portions of streets and alleys.

Lot Coverage: That portion of a lot that is covered by buildings, measured from where the exterior wall meets the foundation, excluding open decks, roof projections (extending less than three feet (3') from the exterior wall), stoops, sidewalks, pathways, patios and driveways. Portions of the building that project beyond the foundation wall and are covered with an impervious membrane (roof covering) such as bays, bay windows, porches, decks and basement spaces, shall be included in the site coverage calculations. Basements that are contained within the footprint of the first story shall not be counted towards site coverage. When an alley house is constructed on a lot, up to 250 square feet of lot coverage shall be excluded from lot coverage computations, provided that the structure is constructed in rear yards only.

**Lot Lines:** The lines bounding a lot as defined herein.

**Lot line, front:** Any lot line adjacent to a street.

**Lot line, rear:** A lot line opposite a front lot line and not adjacent to a street.

**Lot line, side:** Any lot line not defined as a front or rear lot line.

**Lot of Record:** A parcel of land, the deed for which is duly recorded, that is part of the original town site or an addition to the town site, the plat of which has been duly recorded in the office of the County Clerk of San Miguel County prior to the adoption of the Town's original Zoning Ordinance (Nov. 6, 1974).

**Lot Width:** The length of the minimum required front yard setback line between the two side lot lines. If the front yard setback line is an arc or a curve, the lot width shall be the length of said arc or curve.

**Manufacturing, general**: An establishment engaged in the basic processing and manufacturing of materials or products predominately from extracted or raw materials, excluding uses classified as manufacturing, hazardous or objectionable.

Manufacturing, hazardous or objectionable: A use engaged in storage of, or manufacturing processes utilizing, flammable or explosive materials, or storage or manufacturing processes that potentially involve hazardous or commonly recognized offensive conditions. Typical uses include chemical manufacturing and warehousing, dry ice manufacturing, fat rendering plants, fertilizer manufacturing, fireworks and explosives manufacturing and warehousing, petroleum refineries, pulp processing and paper products manufacturing, radioactive materials manufacture or use, steel works, slaughterhouses and tanneries.

**Manufacturing, light:** An establishment or use engaged in the manufacture, predominantly from previously prepared materials, of finished products or parts, including processing, fabrication, assembly, treatment and packaging of such products, incidental storage, distribution of such products, and sales, including on-site retail sales. Typical uses include production and manufacturing of apparel, electrical and electronic equipment, furniture and fixtures, one of-a-kind jewelry, pottery, arts, crafts, toiletries, leather products, sports and recreational equipment, toys, musical instruments, pharmaceutical and medical apparatus; monuments and grave markers, rubber and plastic products; appliance and computer assembly, bakeries engaged in wholesale distribution, meat cutting and wholesale storage, motion picture production lots, printing and publishing; and woodworking milling operations.

**Master Plan:** The Town of Norwood Comprehensive Development Plan.

**Master Plan Boundary:** See "Urban Growth Boundary".

**Mini Storage:** A structure containing separate, individual, and private storage spaces of varying sizes, leased or rented on individual leases for varying periods of time.

**Mobile Home Dwelling:** A structure designed to be transported after fabrication and equal to or exceeding eight feet in width and/or 32 feet in length, excluding towing gear and bumpers. Such a structure is suitable for human habitation on a year-round basis when equipped with required plumbing, heating and electrical facilities. Mobile homes have integral chassis or permanent hitches to allow future movement. Any substandard-size mobile homes occupied at the time of adoption of these regulations shall be allowed to continue.

**Mobile Home Subdivision:** A tract of land subdivided into lots, which are designed as sites for mobile or relocatable homes and which are served by separate utilities and dedicated streets on a legally filed

plat and are capable of being conveyed as separate lots.

**Mobile Home Park:** A tract of land designed or being used to accommodate two or more mobile home dwelling sites for rental.

**Mobile Home Park Accessory Building:** A building providing laundry facilities operated by the mobile home park management, recreational facilities, storage and all other activities, facilities and uses authorized or approved by the Board of Trustees under Chapter 6.10.

**Mobile Home Park Access way:** A way providing access for vehicular traffic from a mobile home stand to an abutting collector street, cul-de-sac street or a service street.

**Mobile Home Park Pad:** That portion of an individual mobile home space reserved for the placement of a mobile home, and structures or additions appurtenant to the mobile home.

**Modular Unit:** A International Building Code-approved, factory-fabricated, transportable building or major component designed for use by itself or for incorporation with similar units on-site into a structure for residential, commercial, educational, or industrial use. A modular unit must be placed on a permanent foundation.

**Natural Grade:** The highest natural elevation of the ground surface, prior to development, next to the proposed walls of a structure. If the natural grade has been disturbed immediately prior to development, the Building Official may establish the natural grade.

**Neighborhood:** The area adjacent to or surrounding existing or proposed development and characterized by common use or uses, density, style and age of structures and environmental characteristics.

**Net Leasable Area:** Those areas within a commercial or office building which are or which are designed to be, leased to a tenant and occupied for commercial or office purposes, exclusive of any area including, but not necessarily limited to, areas dedicated to bathrooms, lobbies, stairways, circulation corridors, mechanical areas, space related to the operation and maintenance of the building, and storage areas provided, however, that these areas are used solely by tenants on the site.

**Noise:** Any sound that interferes with speech and hearing, is intense enough to damage hearing or is otherwise annoying or unwanted. See also, §5.07.

**Nonconforming Lot:** A lot that fails to meet the minimum lot size or width standards for the zone district in which it is located.

**Nonconforming Structure:** Any structure that no longer conforms to the area and bulk requirements for the zone district in which it is located.

**Nonconforming Use:** Any use no longer permitted under the regulations for the zone district in which it is located.

**Nursing Home:** A facility for the care of elderly people that provides lodging and meals with or without skilled nursing care.

**Occupancy:** (1) residence or tenancy in a building; or (2) the use or intended use of the land or buildings by proprietors or tenants; as the context requires.

Office, Business or Professional: A use where business, professional, or governmental services are made available to the public, including: (1) Business Office - an office for use by persons such as realtors, travel, advertising or insurance agents and property managers providing both products and services, or the home office of a company that sells retail or wholesale products or provides professional services. (2) Professional Office - an office for use by persons such as physicians, dentists, lawyers, architects, engineers, accountants and other professionals who primarily provide services rather than products.

**Off-site Sign:** A type of directional sign generally prohibited in San Miguel County that directs attention to a business, commodity, service, entertainment, attraction or product sold, offered or existing elsewhere than upon the same lot where the sign is located, including billboards.

**Off-street Parking:** Land not within a public right-of-way used for parking, storing, or displaying motor vehicles and similar items. Off-street parking shall be paved with all-weather surfacing or covered with gravel and shall be maintained in a usable condition at all times and as otherwise required by the zone district standards.

**Open Space:** Area included in any side, rear or front yard or any unoccupied space on the lot or tract that is open and unobstructed to the sky except for the ordinary projections of cornices, eaves, porches and plant material.

**Owner:** Any part owner, joint owner, tenant in common, tenant in partnership, joint tenant or other Person with sole or with concurrent legal and/or beneficial title to the whole or to part of a building or land.

**Parcel:** Any quantity of land for which location and boundaries can be established.

**Park:** Publicly owned land used for active or passive recreational purposes.

**Person:** An individual, proprietorship, trust, partnership, corporation, association, or other legal entity.

**Planned Unit Development:** A development provided for by the Planned Unit Development Overlay District wherein certain yard setback, areas and related standards may be varied and a variety of land uses associated on a tract, the plan of which is subject to approval by the Planning and Zoning Commission and Board of Trustees.

**Planning and Zoning Commission:** The body politic appointed by the Board of Trustees as an advisory body to it with the duties and responsibilities delegated to it pursuant to the Norwood Land Use Code.

**Plat, Final:** A map of the subdivision showing accurate surveying by a registered surveyor. The map shall show all streets, alleys, blocks, lots and all other requirements listed in Section 6.05.

**Plat, Preliminary:** The plat of any lot, tract or parcel of land drawn and submitted in accordance with the requirements of adopted regulations listed in Section 6.04, that is not to be recorded of record, but is only a proposed division of land for review and study by the Town.

**Portable/Wheeled Sign:** A sign not permanently affixed to the ground, a building, or other permanently affixed structure, which may be moved from place to place and which may be mounted on wheels.

**Pre-application Conference:** An informal conference held with Town staff or Planning and Zoning Commission for the purpose of obtaining information and guidance prior to formal filing of a land use application.

**Principal Use:** The primary use for which a parcel and any buildings thereon are developed.

**Projecting Sign:** Any non-free-standing sign, affixed to or supported by a building or structure, that projects beyond the surface of that portion of the building or structure to which it is affixed or supported by more than six inches.

**Public Building, Shop or Yard:** Facilities such as office buildings, maintenance yards and shops required by branches of Local, State or Federal Government for service to an area such as Highway Department yard, Town Service Center or Experiment Station.

**Public Facilities/Services:** Major capital improvements, including but not limited to transportation, sewer, solid waste, drainage, water, education and parks and recreation facilities.

**Public Land:** Land or interests in land owned by a governmental entity or held in trust for the benefit of the public by a not-for-profit organization.

**Public Right-of-way:** The legal right established by usage or grant to pass along a specific route through ground or property by the general public.

**Public Use:** Land, buildings and/or facilities owned or operated by a government agency for the benefit of the general public.

**Radio, Television or Microwave Tower:** Structures supporting antennae for transmitting or receiving any portion of the radio spectrum, but excluding noncommercial antennae installations for home use of radio or television.

**Recreational Vehicle/Travel Trailer:** A vehicular portable unit designed for temporary occupancy for travel, recreational and vacation use which are either self-propelled, mounted on, or pulled by another vehicle that may not be used for a dwelling, accessory building, accessory use or home occupation.

**Recreational Vehicle/Travel Trailer Park:** A tract of land designed or being used to accommodate two or more Recreational Vehicles or Travel Trailers for travel, recreation or vacation purposes.

**Refreshment Stand, Temporary:** The use of a portable building or other temporary facilities for the retail sale of food and beverages in a ready-to-consume state. This use may be a principal use or an accessory use to a principal use located on the premises.

**Repair Services, Heavy Equipment:** An establishment engaged in the repair of trucks, buses, agricultural equipment, construction equipment or other heavy equipment.

**Repair Services, Limited:** An establishment engaged in the repair of personal apparel and household appliances, furniture, and similar items, excluding repair of motor vehicles. Typical uses include apparel repair and alterations, small appliance repair, bicycle repair, lawn mower repair, clock and watch repair, and shoe repair shops.

**Replatting:** The division, aggregation or reconfiguration of any previously platted lot, parcel or tract.

**Restaurant, General:** An establishment where the principal business is the sale of food and beverages in a ready-to-consume state and where the design or principal method of operation is where customers, normally provided with an individual menu, are generally served in non-disposable containers by a restaurant employee at the same table or counter at which said items are consumed. This use may include take-out service or service to a customer in a motor vehicle.

**Retail, General (Indoors):** A retail establishment that does not fit the definition of any other land use classification and that does not entail any outdoors sales, service, display, storage or other activity. Typical uses include but are not limited to apparel and accessory stores, camera and photographic supply stores, clothing rental stores, consumer electronics stores, gift, novelty and souvenir shops, liquor stores, luggage and leather goods stores, jewelry stores, music stores and video tape rental stores.

**Retail, General (Outdoors):** A retail establishment that does not fit the definition of any other land use classification and that entails some outdoors sales, service, display, storage or other activity. Typical uses include but are not limited to boat dealers, hot tub dealers, recreational vehicle dealers, and monument sales.

**Roof:** The top covering structure of a building.

**Rooming House:** See "Boarding House."

**Sales Trailer, Temporary:** A mobile office trailer used while model homes are being constructed.

**Safety services:** A facility for conduct of public safety and emergency services, including fire and police protection services and emergency medical and ambulance services.

**School, commercial trade:** An educational institution that provides technical skills required to perform the tasks of a particular type of work with focus on skilled trades rather than providing academic training for students pursuing careers in a professional discipline.

**School, Elementary or Secondary:** The use of a site for instructional purposes on an elementary or secondary level, approved under the regulations of the state.

**Secondhand Store or Pawnshop:** An establishment primarily engaged in the retail sale of previously owned merchandise.

**Setback**: Open space at grade between a structure and the property line of the lot on which the structure is located, measured by the horizontal distance between the lot line and the closest projection of the principal or accessory building. Each setback shall remain unoccupied and unobstructed from the ground upward, except for fences. Where no public dedication exists and the lot line extends to the centerline of the right-of-way, the required yard setback shall equal the distance specified under zone district regulations, plus an additional distance equal to one-half the right-of-way width.

**Sign:** Any letter, figure, character, mark, plane, point marquee sign, design poster, pictorial, picture, stroke, stripe, line, trademark, or reading matter of illuminated or non-illuminated surface that shall be so constructed, placed, attached, painted, erected, fastened, or manufactured in any manner whatsoever, so that the same shall be used for the attraction of the public to any place, subject, person, firm, corporation, public performance, article, machine, or merchandise whatsoever, that is displayed in any manner whatsoever out of doors.

**Small Scale Subdivision:** The division, aggregation or reconfiguration of any parcel of land in excess of 15,000 square feet but not more than 45,000 square feet, into no more than three lots or parcels.

**Smoke:** The visible discharge of particulate matter from a chimney, vent, exhaust or combustion process.

**Stable, Private:** An accessory building for quartering horses for private use without fee.

**Street:** A public way, other than an alley or driveway, which affords the principal means of access to abutting property.

**Street, collector:** A street which has the primary function of carrying through traffic, but which also provides access to abutting property.

**Street, local:** A street which has the primary function of providing access to abutting property, and which does not normally carry through traffic.

**Street Line:** A dividing line between a lot, tract or parcel of land and a contiguous street, the right-of-way line or easement line.

**Street Width:** The dimension of the shortest distance between the lines that delineate the right-of-way of a street, road or other way.

**Structural Alterations:** Any change in the supporting member of a building, such as a bearing wall, column, beam or girder.

**Structure:** That which is lawfully built or constructed, an edifice or building or any kind or any piece of work artificially built up or composed of parts joined together in some definitive manner.

**Subdivider:** A Person who causes land to be divided into a subdivision.

**Subdivision:** The division of any parcel of land into two or more parcels for the purpose, whether immediate or future, of sale or transfer of ownership, or for building or other development. Subdivision shall also mean the consolidation, aggregation and reconfiguration of lots. When appropriate to the context, subdivision relates to the process of subdividing land.

**Subdivision Improvements Agreement:** One or more security arrangement(s) that may be accepted by the Town to secure the construction of such public improvements as are required by this Code as a condition of the approval of a subdivision.

**Supermarket:** A retail establishment primarily offering for sale meats, fruits, vegetables, bakery products, dairy products, personal care items, cleaning products and similar household items for off-premise consumption.

**Surveyor:** A licensed surveyor as authorized by the State statutes to practice the profession of surveying.

**Town:** The Town of Norwood, San Miguel County, Colorado.

**Town Planner:** The person designated to receive and process plats, amendments to this chapter or the zoning map, and other land use applications and advise the Planning and Zoning Commission and Board of Trustees on matters of planning and/or engineering.

**Toxic and Noxious Matter:** Any solid, liquid, or gaseous matter that is present in sufficient quantities to endanger health, safety and comfort of persons in the vicinity or that may cause injury or damage to property.

**Trail:** A multi-purpose easement designed to provide safe and easy passage for non-motorized means of travel.

**Trail Clearing:** The vertical (v.) clearance from the trail tread to overhead obstructions, or the horizontal (h.) clearance from the edge of the trail tread to obstructions within the required vertical clearance. The matrix of Trail Construction Standards in Section 5.03 (c) shows the total required horizontal clearance for both sides combined. For nordic skiing trails, the vertical clearance shown indicates distance above maximum snow depth, and the horizontal clearance indicates the overall requirement including the trail tread width.

**Trail Cross Slope:** The slope measured perpendicular to the centerline of the trail.

**Trail Cross-slope (X-slope) Range:** The range of existing topographic slopes on which a particular type of trail may be constructed economically.

**Trail Maximum Profile:** The maximum slope measured along the centerline of a trail commonly accepted for a particular type of use.

**Trail, Multi-use:** Multi-use trails are designed for the most restrictive of several types of uses they can accommodate. They typically accommodate high-volume pedestrian, equestrian, hiking, and mountain biking.

**Trail Tread:** The useable surface of a trail.

**Urban Growth Boundary:** The Master Plan Boundary illustrated in the Norwood Major Streets and Future Land Use Plan and identifying an area surrounding the Town of Norwood that separates urbanizable land from rural land and within which the Plan anticipates that urban growth will be contained.

**Variance:** A modification in the application of certain zoning district regulations to a particular parcel of land granted by the Town through applicable provisions of this Title.

**Vibration:** Periodic or random oscillations about an equilibrium point caused by a mechanical phenomenon.

**Wall Sign:** Any sign painted, placed on, incorporated in or affixed to a building wall, window or canopy, or any sign consisting of cut-out letters or devices affixed to a building wall, window or canopy with no background defined on the building wall, window or canopy, with the exposed face of the sign located in a place substantially parallel to the wall, window or canopy surface on which it is placed.

**Warehouse:** A building or portion of a building for the storage of materials or goods.

Wrecking or Auto Salvage Yard: A yard or building where automobiles or parts of automobiles or machinery are stored, dismantled and /or offered for sale in the open as whole units, as salvaged parts or as processed metal.

**Yard:** The area between buildings and property lines generally defined by setbacks.

**Yard, front:** A yard adjacent to a front lot line and extending from the lot line more or less a uniform distance into the lot to the nearest projection of a structure.

**Yard, rear:** A yard adjacent to a rear lot line and extending from the lot line more or less a uniform distance into the lot to the nearest projection of a structure.

**Yard, street:** The area of a lot that lies between any street line and the facing wall line or lines of the principal building, extended by imaginary lines from the outer corners of the building and parallel to the street to the property lines. In the case of multiple principal buildings located on a building tract, imaginary lines shall connect the corners of facing walls of adjacent buildings.

**Zoning Development Permit:** A permit issued by the Town Planner that allows a developer to engage in development in compliance with all applicable sections of this Code and further enables the developer to seek a building permit that would allow the developer to commence actual development.

**Zoning Map:** The certified Official Zoning Map upon which the boundaries of the various zoning

districts are drawn.

## § 2.03 Interpretations

- (a) Authority. The Town Planner shall have the authority to make all interpretations of the text of this chapter, and the boundaries of the Official Zoning Map.
- **(b) Initiation**. An interpretation may be requested by any affected person, any resident or real property owner in the Town of Norwood, or any person having a contractual interest in real property in the Town of Norwood.
- (c) Procedures.
  - (1) **Submission of request for interpretation.** Before an interpretation shall be provided by the Town Planner, a Request for Interpretation shall be submitted to the Town Planner in a form established by the Town Planner.
  - (2) **Determination of completeness.** Within a reasonable amount of time after a Request for Interpretation has been received, the Town Planner shall determine whether the request is complete. If the Town Planner determines the request is not complete, he shall serve written notice on the applicant specifying the deficiencies. The Town Planner shall take no further action on the Request for Interpretation until the deficiencies are remedied.
  - (3) Rendering of Interpretation. After the Request for Interpretation has been determined complete, the Town Planner shall render an interpretation within a reasonable amount of time. The Town Planner shall consult with the Mayor and Town Attorney and review this chapter and the Official Zoning Map, whichever is applicable, before rendering an interpretation.
- (d) Form. The interpretation shall be in writing and shall be sent to the applicant by certified mail.
- (e) Official Record. The Town Planner shall maintain an official record of all interpretations in the Town Hall, which shall be available for public inspection during normal business hours.
- (f) Appeal. Any person who has made a Request for Interpretation may appeal interpretation of the Town Planner to the Board of Trustees by filing a petition within thirty (30) days of the Town Planner's decision. The date of the decision shall be the postmark date of the certified mail notifying the applicant of the interpretation. The petition shall be considered by the Board of Trustees within thirty (30) days of its filing, and the interpretation of the Town Planner affirmed or modified.

## **Chapter 3.00 ZONING DISTRICT REGULATIONS**

## § 3.01 Scope and Applicability

The regulations of this Sec. 3.00 shall apply to all lands located within the corporate limits of the Town of Norwood, Colorado. All land, buildings, structures or appurtenances thereon located within the Town of Norwood, Colorado, that are hereafter occupied, used, erected, altered, removed, placed, demolished or converted shall be occupied, used, erected, altered, removed, placed, demolished or converted in conformance with the zoning regulations prescribed for the zoning district in which such land or building is located as hereinafter provided.

## § 3.02 Districts Established

In order to implement the Comprehensive Plan and the other purposes and provisions of this Land Use Code, the Town of Norwood, Colorado, is hereby divided into seven (7) zoning districts. The regulations as set out herein are uniform throughout each district. The districts established herein shall be known as:

Abbreviated Designation	Zoning Districts
A	Agricultural District
R-1	Residential District
MH	Mobile Home District
B-1	<b>Business District</b>
P	Public District
I	Industrial District
MD	Medium Density District

## §3.03 Official Zoning Map

- (a) Map adoption. The boundaries of the zoning districts set out herein are delineated upon the Official Zoning Map of the Town of Norwood, Colorado, said map being hereby adopted as a part of this chapter as fully as if the same were set forth herein in detail.
- (b) Zoning map amendment. No changes or amendments to the district boundaries shown on the official zoning map shall be made except in compliance and conformity with all procedures set forth in Sec. 6.02, Zoning Map and Land Use Code Amendments including areas rezoned Planned Unit Development Overlay as provided in Section 3.13 (d). If, in accordance with these procedures, changes or amendments are made to district boundaries, such changes or amendments shall be made promptly after official adoption of the change or amendment as provided for herein. The Town Planner shall be responsible for the physical updating and amendment of the official zoning map.
- (c) Map replacement. In the event that the official zoning map becomes damaged, destroyed, lost or difficult to interpret because of the nature or number of changes and additions, the Board of

Trustees may adopt a new official zoning map that shall supersede the prior map. The new official zoning map may correct drafting and clerical errors or omissions in the prior official zoning map, but no such corrections shall have the effect of amending this chapter or any subsequent amendment thereto without a duly noticed public hearing as provided herein. Unless the prior official zoning map has been lost or totally destroyed, the prior map or any significant parts thereof remaining shall be preserved, together with all available records pertaining to its adoption or amendment.

- (d) Interpretation of district boundaries. The district boundary lines shown on the Official Zoning Map are usually along streets, alleys, property lines or extensions thereof. Where uncertainty exists as to the boundaries of districts as shown on the official zoning maps, the following rules shall apply.
  - (1) Boundaries indicated as approximately following streets, highways or alleys shall be construed to follow the centerline of such street, highway or alley.
  - (2) Boundaries indicated as approximately following platted lot lines shall be construed as following such lines.
  - (3) Boundaries indicated as approximately following town limits shall be construed as following town limits.
  - (4) Boundaries indicated as approximately following the centerline of irrigation ditches or drainage ways shall be construed to follow such centerline.
  - (5) Boundaries indicated as parallel to or extensions of features indicated in this subsection shall be so construed. Distances not specifically indicated on the original Zoning Map shall be determined from the graphic scale on the Map.
  - (6) Whenever any street, alley or other public way is vacated by official action of the Board of Trustees the zoning district line adjoining each side of such street, alley or other public way shall be automatically extended to the centerline of such vacated street, alley or way, and all area so involved shall then and henceforth be subject to all regulations of the extended districts.
  - (7) Where physical features of the ground are at variance with information shown on the Official Zoning Map, or when there arises a question as to how or whether a parcel of property is zoned and such question cannot be resolved by the application of subsections 3.03(d)(1) through (7), the property shall be considered as classified A, Agricultural District, temporarily, and subject to Sec. 3.04.

## § 3.04 Temporary Zoning of Annexed Territory

(a) All territory hereafter annexed to the Town of Norwood shall be temporarily classified as A, Agricultural District, until permanent zoning is established by the Board of Trustees of the Town of Norwood. The procedure for establishing permanent zoning on annexed territory shall conform to the procedure established by this Article for zoning amendments.

- (b) In an area temporarily classified as A, Agricultural District, the following regulations shall apply.
  - (1) No person shall erect, construct or proceed or continue with the erection or construction of any building or structure or add to any building or structure or cause the same to be done in any newly annexed territory to the Town of Norwood without first applying for and obtaining a Zoning Development Permit and a Building Permit therefore from the Building Official or the Board of Trustees as may be required herein.
  - (2) No permit for the construction of a building or use of land shall be issued by the Building Official other than a permit that will allow the construction of a building permitted in the A, Agricultural District, unless and until such territory has been classified in a zoning district other than the A, Agricultural District, by the Board of Trustees in the manner provided in section 6.02.

## § 3.05 Use Regulations

- (a) Use regulations generally. Land and buildings in each district may be used for any of the principal or accessory land uses authorized in the regulations set forth for that district in Sec. 3.07, A, Agricultural District through Sec. 3.12, P, Public District and Sec. 3.14, MD, Medium Density District, but no land shall hereafter be used and no building or structure shall hereafter be occupied, used, erected, altered, removed, placed, demolished or converted that is arranged or designed to be used or used for other than those uses specified for the district in which it is located, other than accessory uses incidental to a permitted principal use and complying with the provisions of Sec. 3.05(d)(1), Accessory use or structure, general.
- **(b) Use classification**. The use regulations of the zoning districts are based on the following use classification system.
  - (1) Purpose and intent. The purpose of these provisions is to classify uses of land into a number of specially defined land use categories on the basis of common functional characteristics and similar compatibility with other uses, thereby providing a basis for regulation of uses in accordance with criteria that are directly relevant to the public interest. These provisions shall apply throughout this chapter.
  - Classification of principal uses. The use categories listed on the following Use Regulations Schedule shall be considered to be principal uses. A single lot may contain more than one principal use, unless otherwise prohibited by the provisions of this Title. All existing and proposed uses shall be classified into the use category that most closely portrays the overall nature of such activity, consistent with the definitions of the use categories in Sec. 2.02, Definitions. Any use that cannot be so classified shall not be permitted in any district.
- (c) Schedule of use regulations. The following Use Regulations Schedule summarizes the use regulations of the districts. In the event of any conflict between the Use Regulations Schedule and the text of the zoning district regulations, the text shall control. Uses that were established prior to the adoption of this chapter or its predecessors, but that are now inconsistent with the requirements of this chapter, shall be permitted if such uses meet the requirements of Sec. 1.05, Nonconforming Provisions, for a valid nonconforming use. The Use Regulations Schedule shall be interpreted as follows.
  - (1) Permitted uses. Uses identified in a particular district column with a "P" shall be permitted in such district, subject to compliance with any applicable conditions and all other provisions of this Title. In the event that any use, regardless of classification, meets the definition of Formula Business Establishment it may only be permitted as a conditional use subject to special use conditions 13 and 21 of Section 3.05(d) in addition to any other special use condition applicable to that certain use.
  - (2) Conditional uses. Uses identified in a particular district column with an "C" shall be permitted in such district only upon approval of a conditional use permit by the Board of

Trustees in accordance with the procedures and standards of Sec. 6.10, Conditional use permits. In the event that any use, regardless of classification, meets the definition of Formula Business Establishment it may only be permitted as a conditional use subject to special use conditions 13 and 21 of Section 3.05(d) in addition to any other special use condition applicable to that certain use.

- (3) Temporary uses. Uses identified in a particular district column with a "T" shall be permitted in such district only upon approval of a temporary use permit in accordance with the procedures and standards of Sec. 6.17, Temporary use permits.
- (4) Not permitted. Uses not identified in a particular district column with a "P", "C" or "T" are not allowed in such district unless the P&Z determines that the use meets the criteria of Sec. 3.05 (b)(2). The P&Z shall also find that the proposed use can meet the standards of Sec. 6.10, Conditional use permits. Any additional use deemed by P&Z to be classified within the Schedule of Use Regulations shall be subject to the procedures and standards of Sec, 6.02, Zoning Map and Land Use Code Amendments.
- (5) Special use conditions. Numbers occurring in parenthesis after the names of selected use categories refer to conditions applicable to the use and set forth in Sec. 3.05(d), Special use conditions. In the event that any use, regardless of classification, meets the definition of Formula Business Establishment it may only be permitted as a conditional use subject to special use conditions 13 and 21 of Section 3.05(d) in addition to any other special use condition applicable to that certain use.

## SCHEDULE OF USE REGULATIONS

Use Type		<b>Zoning District</b>								
	A		M H	B 1	Ι	P	MD			
Residential uses										
Single-family dwelling	P	P	P	C	P		P			
Medium Density Residential (19)		C								
Cottage Dwelling Units (17)		C					C			
Double-wide manufactured/ mobile home (12)	P	P	P				P			
Duplex dwelling		P		С			P			
Multi-family dwelling (See Note 1)		С		P			С			
Mobile home park (6)			P							
Mobile home subdivision (6)			P							
Residential Use, Growing, Cultivating and Processing of Marijuana, subject to all requirements of Chapter 7, Title 7	P	P	P	P	P		P			

Use Type		<b>Zoning District</b>									
	A	R	M	В	I	P	MD				
		1	H	1							
Accessory and temporary uses			ı								
Accessory use or structure (1)	P	P	P	P	P	P	P				
Alley house (14)		C									
Asphalt or concrete batch plant, temporary (3)		T	T		T	T	T				
Building material yard/construction office, temporary	T	T	T	Т	T	T	T				
Caretaker or guard residence	C				P	P					
Field office, temporary	T	T	Т	Т	T	T	T				
Home occupation (5)	P	P	P	С			P				
Off-street parking, accessory	P	P	P	P	P	P					
Sales trailer, temporary (7)	T	T	Т		T		Т				
Stable, private (8)	P	С			С						
Educational, institutional and civic uses	<u>.</u>										
Church or place of worship	P	P	P	P	P	P					
Community center (18)		С	С	С	С	С	С				
Correctional facility or jail, public						С					
Day care center	P	С	С	P	P	P	С				
Day care home	P	P	P	P	P	P	P				
Educational Facilities	P	P	С		P	P	С				
Essential services	P	P	P	P	P	P					
Events center or fairgrounds	С					P					
Fraternal club or lodge	С			P	P	P					
Group home	P	P		P			Р				
Hospital or clinic (13)		С		С		С					
Library				P		P					
Museum				P	P	P					
Nursing home (13)	С	С		С	С		С				
School, elementary or secondary	P	P				P	P				
Fire station	P	P	P	P		P					
Gas line and regulating station (10)	P	С	С		1	С					
Private franchise Utility					С						
Private utility shop, yard or building	1	1			С						

Use Type	<b>Zoning District</b>									
	A	R	M	В	I	P	MD			
		1	H	1						
Public building, shop or yard (10)				P	P	P				
Radio, TV or microwave tower (10)	С				P	P				
School, commercial trade	С	C		C	P	C				
Sewage treatment plant (10)	C				C					
Sewage pump station (10)	P	P	P	C		P				
Water treatment, storage or pump station (10)	C	P	P	C		P				
Transportation-related uses										
Bus station or terminal (13)				P	P					
Heliport	С	С				P				
Parking lot, trucks or trailers (transport)				С						
Storage or hauling company					P					
Truck freight terminal					P					
Retail uses				•						
Antique shop (13)		С		P						
Art gallery or handicraft sales (13)				P						
Art supply store (13)				P						
Bakery or confectionery shop (13)				P	P					
Bar				P						
Book or stationary store or newsstand (13)				P						
Convenience store (13)				P	P					
Department or variety store (13)				P						
Drug store or pharmacy (13)				P						
Fabric, drapery and needlework store (13)				P						
Florist (13)				P						
Fruit or vegetable stand (13)	P			P	P					
Furniture or appliance store (13)				P	P					
Greenhouse or plant nursery (13)	С			P	P					
Hardware store and hobby shop (13)				P	P					
Liquor store (13)				P						
Optical sales (13)				P						
Paint and wallpaper store				P						

Use Type	Zoning District								
	A	R	M	В	I	P	MD		
		1	H	1					
Pet store (13)				P	P				
Photography shop				P					
Restaurant (13)				P					
Retail, general (indoors) (13)				P					
Retail, general (outdoors) (13)					P				
Second hand store or pawnshop (13)				P					
Sporting goods (13)				P					
Supermarket (13)				P	С				
Toy store or hobby shop (13)				P					
Personal service uses									
Bank or savings and loan (13)				P					
Bed and Breakfast (13)	P	P		P					
Boarding or rooming house (13)		С		С			С		
Cleaning shop or laundry pick-up station (13)				P	P				
Custom personal service (13)		С		P	P		С		
Hotel or motel (13)				P	С				
Household appliance service and repair (13)				P	P				
Key shop (13)				P	P				
Laundry, self-service (13)				P	P				
Office, business and professional (13)				P	P				
Mortuary or funeral home (13)				P					
Studio, television or radio (13)				P	P				
Tattoo Parlor				С					
Tool rental (domestic equipment) (13)				P	P				
Travel bureau or consultant (13)				P	P				
Veterinary clinic (13)	P			P	P				
Recreational and entertainment uses							11		
Amusement, commercial (2)(13)				P					
Boat service (13)			1	С					
Boat storage and service (13)					P				
Campground				C	C	С			

Use Type	<b>Zoning District</b>									
	A	R	M	В	I	P	MD			
		1	H	1						
Day camp for children	C	C	C		P	P	С			
Dude ranch	C									
Equestrian center	C					P				
Park	P	P	P	P	P	P	P			
Rodeo grounds	C					C				
Recreational vehicle storage and service business					P					
Recreational Vehicle/Travel Trailer Park (11)(13)				С	С	С				
Stable, commercial boarding or rental	С									
Theater (13)				P	С					
Automobile and related service uses										
Auto leasing or rental					P					
Auto parts and accessory sales (indoors) (13)				P	P					
Auto repair garage (13) (15)				P	P					
Car wash (13)				С	P					
Gasoline service station (15)				P	P					
New or used car or truck sales (13)				С	P					
Snowmobile, motorcycle or ATV, sales or service (13)				С	P					
Wrecking or auto salvage yard					С					
Heavy commercial uses (13)						•				
Building material sales and yard				С	P					
Cabinet and upholstery shop				С	P					
Cleaning or laundry plant				С	P					
Contractor storage or equipment yard										
Electrical substation, high voltage	С	С	С		С					
Heating or air conditioning shop				С	P					
Print shop or Copy Shop				P	P					
Machinery, heavy - sales and rental				С	P		1			
Maintenance and repair service for buildings				С	P		1			
Mini-warehouse-Less than 2500 square feet					P		1			
Mini-Warehouse – greater than 2500 square feet					С		1			
Plumbing shop				С	P		1			

Use Type	Zo	ning	g Dis	tric	t		
	A	R 1	M H	B 1	I	P	MD
Production and Creation Establishments				С	P		
Repair services, limited (13)				P	P		
Storage warehouse (mini-storage units) (16)		C			P		
Trailer or mobile home sales or rental					P		
Welding or machine shop					P		
Industrial uses							
Asphalt/concrete batching plant, permanent							
Food processing							
Manufacturing, general							
Manufacturing, hazardous/objectionable					С		
Manufacturing, light (13) and (20)				C	P		
Repair services, heavy equipment					P		
Farm-related uses							
Animal pound or kennel (public or private) (13)	С			С	P		
Animal production and husbandry (4)	P						
Farm buildings (under 25' in height)							
Farm buildings (over 25' in height and under 35')							
Feed store (13)				P			
Earth, clay, rock or soil storage	P						

- (d) Special use conditions. The following conditions apply to the listed uses when referenced in the use regulations of a particular zoning district, Sec. 3.07, A, Agricultural District through Sec. 3.13, PUD, Planned Unit Development Overlay District, whether listed as the use is permitted or conditional.
  - (1) Accessory use or structure. Accessory uses or structures may be permitted subject to the following conditions:
    - a. Such accessory uses shall be limited to those customarily associated with and appropriate, incidental and subordinate to the principal use,
    - b. Such accessory uses shall be located on the same lot or tract as the associated principal use.

c. Such accessory uses shall be controlled in the same manner as the associated principal use, except as otherwise expressly provided in this chapter.

In no event shall an accessory use be construed to authorize a use not otherwise permitted in the zone district in which the principal use is located, and in no event shall an accessory use or structure be established prior to the principal use or structure to which it is accessory.

- (2) Amusement, commercial. An indoor commercial amusement enterprise shall be wholly enclosed in a building that is treated acoustically so that noise generated by the enterprise is not perceptible at the bounding property line.
- (3) Asphalt or concrete batching plant (temporary). A temporary asphalt or concrete batching plant permit may be approved by the Planning and Zoning Commission subject to the following conditions.
  - a. The batching plant site shall comply with all applicable provisions of Town, State and Federal laws.
  - b. The batch plant shall not be located within six-hundred feet (600') of a residence.
  - c. Hours of operation will be limited to Monday through Friday, 7:00 a.m. to 7:00 p.m.
  - d. The batch plant permit be valid for up to six (6) months.
  - e. No portion of the batch plant or its operation shall be located on a public street.
  - f. The batch plant shall only furnish concrete, asphalt, or both, to the specific project for which the temporary Zoning Development Permit is issued. The placement of a temporary batching plant for a private project is restricted to the site of the project.
  - g. The temporary plant shall be operated in a manner that eliminates unnecessary dust, noise and odor (as illustrated by, but not limited to, covering trucks, hoppers, chutes, loading and unloading devices and mixing operations, and maintaining driveways and parking areas free of dust).
  - h. The site must be clear of all equipment, material and debris upon completion of the project.
  - i. All public improvements that are damaged during the operation of the temporary batching plant must be repaired or replaced within thirty (30) days of completion of the project.
  - j. At termination and/or removal of plant permit, permittee shall have the person responsible walk the site with the Building Official or his/her designee to verify the site meets Town approval.

- (4) Animal Production and Husbandry. The production and care for typical farm animal such as poultry, cows and sheep on a small scale, as in a 4-H project, shall be permitted only on a lot or tract of one acre or more. In the event that the Planning and Zoning Commission receives written complaint from an adjacent property owner that animals or animal husbandry practices on a particular lot or tract are causing a public nuisance or preventing an adjacent property owner from enjoying the use of their property, the animal production and husbandry shall be subject to approval of a conditional use permit and the Commission shall schedule a hearing for the purpose of deciding whether such use may continue. The approval of a Conditional use permit for animal production and husbandry in any zone district shall be subject to the following standards:
  - a. Sec. 6.10, Conditional uses, and
  - b. Sec. 5.07, Performance Criteria.

Animal production and husbandry shall not include the commercial feeding of offal and garbage to swine and other animals or any type of husbandry specifically prohibited by Ordinance or law.

- (5) Home occupation. A home occupation shall be permitted as an accessory use to a dwelling unit subject to compliance with the following conditions.
  - a. A home occupation shall be permitted only when it is an accessory use to a dwelling unit and does not change the essential residential character of the use.
  - b. A home occupation shall not involve any structural alteration in the main building or any of its rooms which changes the residential character.
  - c. A home occupation shall not involve the installation of machinery or additional equipment other than that customary to normal household operations.
  - d. Not more than one person other than a member of the family of the owner or the user of the principal dwelling shall be employed or work in or at such home occupation.
  - e. A person who engages in a home occupation may have one sign no greater than two (2) square feet in size on the building in which the home occupation is located.
  - f. A home occupation shall not operate during the hours other than 8:00 a.m. to 7:00 p.m.
  - g. A home occupation shall not involve more than four (4) patrons on the premises at one time.
  - h. A home occupation shall not generate loud and raucous noise, or utilize mechanical, electrical or other equipment or items that produce noise, electrical or magnetic interference, vibration, heat, glare, smoke, dust, odor or other nuisance causing

annoyance outside the residential building or accessory structure by the sense of normal human beings that substantially and unreasonably interferes with the use and enjoyment of adjacent property or interferes with public peace and comfort.

- i. A home occupation does not store outside of the dwelling unit any equipment or materials used in the home occupation.
- j. The home occupation may be located in an accessory building on the property.
- k. A home occupation shall be totally contained within a structure(s) and shall be secondary to primary use of residential. The floor area devoted to a home occupation shall not exceed fifty (50) percent of the floor area of the primary structure or 1000 square feet, whichever is less.
- l. No retail sales may occur on the premises unless by appointment. The home occupation shall be a service, contractor or similar businesses whose customers are by appointment only.
- m. There shall be no storage on site of dangerous, toxic, combustible or hazardous materials.
- n. At least one additional off-street parking space shall be provided for the non-resident employee.
- o. The use shall not cause off-site impacts on the surrounding neighborhood due to traffic and deliveries.
- p. The operator of a home occupation may apply to the Town Board for special or temporary outside or other types of activities that may cause off-site noise impacts no more than two (2) times per calendar year.
- (6) Mobile home park or subdivision. Mobile home parks and mobile home subdivisions shall be subject to the following requirements. Mobile Home Parks shall be reviewed according the procedures and standards included in Sec. 6.10, Conditional Uses and according to the standards listed in Sec. 3.05 (d) (19). Mobile Home Subdivisions shall be reviewed according to the procedures in Sec. 6.04, 6.05, 6.05A, 6.05B, 6.06, 6.07, 6.08 and 6.09 and according to the standards included in Sec. 3.05 (d) (19).
  - a. Existing mobile home parks or mobile home subdivisions shall not be enlarged, expanded or additional mobile homes permitted unless the same shall be brought into compliance with the standards of this subsection and the area regulations of section 3.09, MH, Mobile Home District regulations.
  - b. Individual mobile homes in mobile home parks or mobile home subdivisions shall not be replaced unless the individual mobile home site is in compliance with the

standards of this subsection and the area regulations of section 3.09, MH, Mobile Home District regulations.

- c. Mobile home parks or mobile home subdivisions shall have an internal driveway not less than twenty feet (20') wide and the internal driveway shall be paved with two inches (2") of asphalt.
- d. Each mobile home site in mobile home parks and mobile home subdivisions shall be clearly designated and arranged so that all mobile homes have access to the internal driveway.
- e. There shall be a minimum of twenty feet (20)' between mobile homes.
- f. Mobile home parks and mobile home subdivisions shall be improved with adequate and sufficient night lighting to enable persons to walk in such areas at night without difficulty.
- g. Mobile home parks and mobile home subdivisions shall be properly graded and well drained, so as to prevent the accumulation of surface water.
- h. Each trailer site shall be improved with a four inch (4") concrete or gravel pad. No trailer with a footprint larger than the pad shall be placed on a site.
- i. All refuse shall be stored in fly-tight, water-tight, rodent-proof, dog-proof containers, which shall be located no more than one hundred feet (150') from each mobile home site. Containers shall be provided in sufficient number and capacity to properly store all refuse generated in the mobile home park or mobile home subdivision.
- j. Exposed ground surfaces in all parts of a mobile home park or mobile home subdivision shall be paved or covered with stone screening or other solid material, or protected with a vegetative growth that is capable of preventing soil erosion and of eliminating objectionable dust.
- k. A detached, accessory storage building or buildings shall be provided within a mobile home park or mobile home subdivision containing a minimum storage capacity of ten (10) cubic yards per mobile home site.
- l. Mobile home dwelling units shall be "skirted" within sixty (60) days of its placement in a Mobile Home Park or Subdivision by the affixing thereto of a solid, nonporous screening or skirt between the underside of the dwelling unit at its outer edge around ground level.
- m. Exterior boundaries of mobile home parks and mobile home subdivisions shall be developed with a fence, or other acceptable border to create an attractive border. The land between the fence/border and the public street improvements shall be landscaped with street trees and other landscaping materials and shall be maintained by the owner of

the mobile home park.

- n. Mobile home parks and mobile home subdivisions shall connect to the sanitary sewer system of the Norwood Sanitation District. All utility lines shall be installed underground.
- o. Each mobile home park or mobile home subdivision shall include a recreational area and facilities for the use and enjoyment of the residents encompassing an area of ten percent (10%) or two (2) acres, whichever is less, of the total area of the mobile home park or mobile home subdivision. Lawns, parking, driveways, accessway, streets, etc shall not be considered part of the required recreational area and facilities.
- p. Notwithstanding any other provision of this chapter, mobile home sites in mobile home parks and mobile home lots within mobile home subdivisions shall be developed according to the area regulations for mobile home dwellings set forth in Sec. 3.09, MH, Mobile Home District regulations, except as such standards may be modified by more restrictive conditions imposed by the Board of Trustees in a Planned Unit Development Overlay District.
- q. All parks and individual mobile home sites shall be provided with safe, convenient, all season pedestrian access of adequate width for the intended use.
- r. The minimum size of a mobile home park or subdivision shall be five (5) acres.
- (7) Sales trailer, temporary. A temporary use permit may be issued by the Planning and Zoning Commission for a mobile office trailer used while model homes are being constructed, subject to the following conditions:
  - a. A temporary permit allowing sales from a mobile office trailer may be issued for a period not to exceed one hundred twenty (120) days from approval of a Zoning Development Permit, with no extensions, while permanent model homes are being constructed.
  - b. The trailer shall not be used for living or sleeping purposes.
  - c. Skirting shall be installed around the trailer to conceal the undercarriage.
  - d. The trailer shall be located in a platted subdivision on a lot that is owned by the applicant/builder and shall not be located within a required front yard setback.
  - e. A building permit must be issued for the model home at the time of the temporary permit for the trailer.
- (8) Stable, private. A private stable shall be located on a lot or tract containing a minimum of two (2) acres of land and shall be set back from all adjacent property lines at least fifty

feet.

- (9) Multi-family dwelling. Multi-family dwellings are permitted in the B-1, Business District only in conjunction with, attached to, and under the same roof as a permitted non-residential use(s).
- (10) Utility structures and facilities (above ground). Above ground utility structures and facilities, shall be comply with the requirements of Sec. 5.05, Landscaping and Screening.
- (11) Recreational Vehicle/Travel Trailer Park and Campground. A recreational vehicle/travel trailer park, campground and incidental facilities shall comply with the standards in this section.
  - a. Such areas may be occupied only by persons using travel trailers, truck campers and tents for overnight and short duration camping (30 consecutive days maximum or a total of 30 days within any 120 day period);
  - b. Each space for travel trailers, truck campers and tents shall be at least 1500 sq. ft. in area:
  - c. Each space shall be at least 30 ft. in width;
  - d. Each park or camping area shall connect be served by the Norwood Water Commission and the Norwood Sanitation District:
  - e. No dependent travel trailer, truck camper or tent shall be located more than 200 ft. from a water and sewage service building; and
  - f. Provisions shall be made for adequate all-weather parking and walkways to each space.
  - g. No park or campground shall be used for long term housing (stays of longer than 30 consecutive days or a total of 30 days within any 120-day period).
  - h. Parks and campgrounds shall be landscaped and screened from adjacent roadways and residential land uses.
  - i. The minimum setback from a residential use may be increased and the Town may apply the screening and landscaping standards included in Sec. 5.05 as necessary as part of a conditional use review outlined in Sec. 6.10.
  - j. Recreational vehicle/travel trailer parks shall not exceed 2 acres.
  - k. Convenience establishments may be permitted as an accessory use subject to the following restrictions:
    - (i) such establishment shall be restricted in their uses to occupants of the

park.

- (ii) such establishment shall present no visible evidence from any street outside the park of their commercial character which would attract customers other than occupants of the park.
- (iii) the structure of such facility shall not be located closer than 100 feet to any public street and shall be accessible only from a street within the park or campground.
- (iv) should the recreational vehicle park be discontinued, the convenience establishment shall also be discontinued.
- (12) Double-wide manufactured/ mobile home. Notwithstanding other provisions to the contrary, a HUD Standard double-wide manufactured/ mobile home shall be permitted subject to the following conditions:
  - a. The roof must be pitched, have overhangs and be constructed of materials that are the same as, or resemble standard house roofs. Bowed roofs, and/or roofs painted with aluminum appearing tar are not acceptable.
  - b. The siding must be brick, rock, vinyl, stucco, wood or cosmetically similar to standard house siding. Mobile home type, "slick siding" is not acceptable.
  - c. There must be off-street parking provided.
  - d. If there is no garage, an enclosed outside storage area must be provided.
  - e. There shall be a permanent foundation meeting the UBC Guidelines for Manufactured Housing Installation and there shall be a continuous perimeter that produces the appearance of a site built home.
  - f. All axles, wheels and hitches must be removed.
  - g. The residence shall be a minimum of 21 feet wide with an average roof pitch of at least 4:12, a minimum eave overhang of 12 inches, a minimum floor area of 800 square feet and shall be designed for a 40 pound snowload.
  - h. The double-wide manufactured/mobile home shall be subject to all of the setback and minimum lot area requirements of the underlying zone district.
  - i. The ownership of the home and the lot or parcel shall be the same.
  - j. How the structure appears is very important. How the structure was put together is not as important as how the finished product looks. If the structure "looks and feels" like a house, even though it was delivered to the site in a similar fashion to the way mobile homes are, and it meets the above criteria, the structure would be permitted as if it were a single family house and as a use permitted by right.

- (13) Commercial Development in B-1 District. Permits for Commercial Development may be issued only where alley service access is dedicated to adequately serve the proposed Commercial Development across the rear of the subject property. Such access shall be consistent with the standards of Section 4.03 (k) and (p) and the Town of Norwood Major Streets and Future Land Use Plan. Alley easements shall be required unless waived in accordance with Section 4.03(o).
- (14) Alley House, Accessory. An accessory alley house may be permitted as an accessory use to a conforming single family dwelling including a double wide manufactured/mobile home, subject to the following conditions:
  - a. The alley house may contain up to 500 square feet or 50% of the finished and heated square footage of an otherwise conforming single family dwelling or double-wide manufactured/mobile home (excluding garage space) whichever is less;
  - b. One (1) additional off-street parking space with access to and from an alley shall be provided for the alley house, in addition to off-street parking otherwise required in Sec. 5.02 (d);
  - c. The alley house may be attached or detached to an otherwise conforming single family dwelling unit or a double wide manufactured/mobile home, a garage or other accessory building, subject to compliance with applicable setbacks and maximum lot coverage requirements;
  - d. All applicable water and sewer fees shall be paid for the accessory alley house as required for any residence or dwelling unit; and
  - e. The alley house shall not be sold separately.
- (15) Auto Repair Garage. Auto repair garages may be permitted subject to the following conditions:
  - a. All parts and equipment shall be shored indoors at all times or screened from offsite view.
  - b. There shall be no off-site parking of vehicles awaiting repairs.
- (16) Storage warehouses (mini-storage units) Storage units may be permitted as a conditional use subject to the following conditions:
  - a. Daily hours of operation 8 A.M. to 8 P.M.
  - b. 6 foot perimeter fencing that will be conductive to the neighborhood.
  - c. Unlit signage.
  - d. No exterior storage allowed at this site, including but not limited to boats, motor

homes, trailers, or automobiles.

- e. Covered trash dumpsters adequate to serve the facility shall be maintained on site at all times.
- f. No commercial or retail business shall be permitted on site. Units will be used for the purpose of storage only.
- g. Exterior lighting shall illuminate downward.
- h. A landscape plan that allows the property to blend with the surrounding area.
- i. Location and surrounding uses must be considered.
- (17) Cottage Housing Development. A cottage housing development may be permitted, subject to the following conditions:
  - a. The minimum lot area requirement for a cottage housing development shall be thirty thousand (30,000) square feet and the maximum shall be 45,000 square feet.
  - b. The maximum lot coverage for a cottage dwelling unit shall be 850 square feet.
  - c. The maximum floor area for a cottage housing dwelling unit shall be one thousand five hundred (1,500) square feet or 1.5 times the lot coverage, whichever is less. Interior stairways shall be allocated between floors served. The floor area of attached garages shall be included in total floor area.
  - d. The following areas shall not be included in floor area calculations:
    - (i) Unheated storage space which is more than 34 below grade;
    - (ii) Architectural projections such as bay windows, fireplaces or utility enclosures not greater than eighteen (18) inches in depth or six (6) feet in width;
    - (iii) Attached porches with roofs.
  - e. Cottage Housing Developments shall include a minimum of four (4) units and a maximum of nine (9) cottage dwelling units, all included in one development application site plan.
  - f. Cottage Housing Development shall not exceed one unit per 2,500 square feet lot area, and shall not exceed nine units.
  - g. Cottage dwelling units are encouraged to have a covered porch.
  - h. A minimum of two hundred and fifty (250) square feet of common open space shall be included for each cottage dwelling unit. At least fifty (50) percent of the each cottage dwelling unit shall abut common open space. All cottage dwelling units shall be within sixty (60) feet of common open space and the common open space shall have

cottage dwelling unit abutting at least fifty percent of its boundaries.

- i. Parking shall be provided for each cottage dwelling unit as follows:
  - (i) Units less than 750 square feet: one space
  - (ii) Units 750 square feet or greater: two spaces
- j. The rules and regulations of the Norwood Water Commission, Norwood Sanitation District and the Norwood Fire Protection District shall apply to all cottage housing developments.
- k. The floor area shall not be increased following construction.
- l. Approval of a condominium subdivision by the Town of Norwood Board of Trustees. The condominium shall include covenants to guide design and use and a phasing plan for development.
- m. The town grid pattern shall be respected in site plans.
- n. A phasing plan shall be submitted for the project.
- (18) Community Center. A Community Center may be permitted as a conditional use subject to the following conditions:
  - a. The use provides adequate off-street parking. At a minimum, one space shall be provided for every five hundred square feet of usable floor area and one space per four seats of auditorium area.
  - b. The use will be compatible with the surrounding uses allowed in the zone district.
  - c. Lighting will be screened.
  - d. Noise will not interfere with the surrounding uses allowed in the zone district.
  - e. Adequate provision has been made in the design and/or the location of the building for loading and unloading of people, equipment and supplies.
  - f. Hours of operation do not exceed 8 a.m. to 9 p.m, or as conditioned by the Town in its review to minimize impacts on surrounding uses.
  - g. The access and streets in the vicinity of the site are adequate for the use.
  - h. Except at private, invitation-only functions, no alcohol may be served except pursuant to a valid special events liquor license.
- (19) Medium density within the R-1 Zone District, manufactured housing and/or mobile home parks and subdivisions, and trailer parks may be permitted as a Special Use subject to meeting the following standards:

- a. The use is compatible with the surrounding neighborhood.
- b. Roads accessing the development are adequate for traffic generated by the use or the applicant's plan for mitigating the impacts is acceptable to the Town.
- c. Impacts associated with the use such as but not limited to traffic, parking and noise are adequately mitigated through measures included in the applicant's plan.
- d. The development provides a variety of model or building types that are compatible with the surrounding neighborhood.
- e. The plans are designed to break up the scale of the project and to provide visual diversity.
- f. The project is located adjacent to or close enough to the Town's commercial core, schools and other services so as to minimize vehicle use.
- g. The use can be designed and constructed to achieve compatibility with the surrounding environment and in harmony with the site's environmental characteristics.
- h. Utilities in the vicinity of the property are adequate to serve the development or the applicant can bring the utilities to a level of service to serve the development without negatively affecting existing utility users.
- (20) Restriction on light manufacturing: To ensure the vitality of the B-1, Business District, no light manufacturing shall be allowed on the first or ground floor of a building or structure within the B-1 zone district unless it has associated retail sales of products.

#### (21) Formula Business Establishment.

- a. Any approval of a Conditional Use Permit for a Formula Business Establishment shall be, at a minimum, subject to:
  - i) Section 6.10, Conditional Use Permits; and
- ii) A determination of whether the business can be operated and managed in a manner compatible with Norwood's Master Plan, including without limitation Norwood's small-town scale and rural character by assessing how well it will:
  - provide for town residents in supplying jobs and services
  - provide for local management authority
  - support the community; i.e. ensure it supports economic cooperation with existing small, locally owned businesses
  - put long-term community interest over short-term economic gain
  - tie its future economic success to that of the community
  - work with the town to resolve conflicts

- provide living wages and benefits for its employees
- use best management practices to meet applicable environmental standards
- iii) Consideration of architectural design, building materials, color scheme, signage, landscaping and other site development aspects to fit the Town's character and be compatible with the Master Plan.
- iv) Consideration of potential requirements for bonding and/or decommissioning any building or other improvement as part of the approval process.
- v) Compliance with all other Land Use Ordinance provisions and other applicable laws.

# § 3.06 Area Regulations

(a) Schedule of residential area regulations. The following Residential Area Regulations Schedule summarizes the regulations of this Chapter with regard to minimum lot size, minimum yard setbacks, maximum lot coverage, minimum floor area per dwelling unit and maximum building height of residential uses in the various zoning districts. The standards shown in the following schedule may be modified by additional provisions contained in this section or in the individual district regulations. In the event of any conflict between the text of this Chapter and the Schedule of Residential Area Regulations, the text shall control.

## RESIDENTIAL AREA REGULATIONS

Development Standard	Zoning District										
	A	R-1	MH	B-1	I	PUD	MD				
	Single	-Famil	y		•	•					
Min. Lot Area/unit (sq. ft.)	2 Ac.	7,500	3,000	7,500		4,000	3,500				
Min. Lot Width (ft.)	200	48	30	48		40	40				
Min. Front Yard Setback (ft.)	40	15	25	15		15	15				
Min. Side Yard Setback (ft.)	25	10	10	10			5-10				
Min. Rear Yard Setback (ft.)	25		10	0		10	0				
Max. Lot Cover (%)	25%	40%	25%	25%		40%	45%				
Min. Floor Area (sq. ft.)	800	800	800	800		800	800				
Min. Common Open Space (%)						25%					
Max. Height (ft.)	25	25	25	25		30	30				
Cottag	e Housi	ng Dev	elopm	ent							
Min. Lot Area/unit (sq. ft.)		2,500				2,500	2,500				
Min. Lot Width (ft.)		35				40	40				
Min. Front Yard Setback (ft.)		15				15	15				
Min. Side Yard Setback (ft.)		5				5	5-10				
Min. Rear Yard Setback (ft.)		5				5	5-10				
Max. Lot Cover (%)		40%				40%	45%				
Min. Floor Area (sq. ft.)		500				500	500				
Min. Common Open Space %						25%	25%				
Max. Height (ft.)		18				18	18				
	Du	plex									

# RESIDENTIAL AREA REGULATIONS

Development Standard	Zoning District										
	A	R-1	MH	B-1	I	PUD	MD				
Min. Lot Area/unit (sq. ft.)	Ī	5,000		5,000		3,000	3,500				
Min. Lot Width (ft.)		60		48		50	60				
Min. Front Yard Setback (ft.)		15		15		25	15				
Min. Side Yard Setback (ft.)		10		10		6	5-10				
Min. Rear Yard Setback (ft.)				0		10	0				
Max. Lot Cover (%)	<b>-</b>	40%		25%		50%	40%				
Min. Floor Area (sq. ft.)	<b>-</b>	500		500		500	500				
Min. Common Open Space						25%					
Max. Height (ft.)		25		25		35	30				
	Mult	ti-Famil	. <b>y</b>	L	<u> 1                                   </u>						
Min. Lot Area/unit (sq. ft.)	T	2,800		5,000	<u> </u>	2,800	2,800				
Min. Lot Width (ft.)		60		60		50	60				
Min. Front Yard Setback (ft.)		15		15		25	15				
Min. Side Yard Setback (ft.)		10		10		10%	5-10				
Min. Rear Yard Setback (ft.)		10		10		10	10				
Max. Lot Cover (%)		40%		100%		40%	40%				
Min. Floor Area (sq. ft.)		350		350			350				
Min. Common Open Space (%)						25%					
Max. Height (ft.)		25		25		35	30				
	Mob	ile Hom	ie	L	<u> </u>						
Min. Lot Area/unit (sq ft)			3,000			3,000					
Min. Lot Width (ft.)			38			30					
Min. Front Yard Setback (ft.)			25			25					
Min. Side Yard Setback (ft.)			10			10					
Min. Rear Yard Setback (ft.)			10			10					
Max. Lot Cover (%)			30%			30%					
Min. Common Open Space (%)						25%					
Max. Height (ft.)			25			25					

# **Alley House**

### RESIDENTIAL AREA REGULATIONS

Development Standard	Development Standard Zoning District										
	A	R-1	MH	B-1	I	PUD	MD				
(on same lot as a single family		·	•	•	•	•					
dwelling or double wide unit)											
Min. Lot Area/unit (sq ft)		7500									
Min. Lot Width (ft.)											
Min. Front Yard Setback (ft.)											
Min. Side Yard Setback (ft.)		10									
Min. Rear Yard Setback (ft.)		0 - 10									
Max. Lot Cover (%)		50 up									
		to									
		5000									
		s.f.									
Min. Common Open Space (%)											
Max. Height (ft.)		18									

**Schedule of nonresidential area regulations**. The following Nonresidential Area Regulations Schedule summarizes the regulations of this Chapter with regard to minimum yard setbacks, maximum lot coverage, maximum floor area ratio, maximum floor area ratio and maximum building height of nonresidential uses in the various zoning districts. In the event of any conflict between the text of this Chapter and the Schedule of Nonresidential Area Regulations, the text shall control.

### NONRESIDENTIAL AREA REGULATIONS

Development Standard	Zoning Districts								
	A	R-1	B-1	Ι	PUD	MD			
Min. Lot Area (sq.ft.)	2 ac.	7,500	5,000	6000	4000				
Min. Lot Width (ft.)	200	48		50	40				
Min. Front Yard Setback (ft.)	40	15		15	15				
Min. Side Yard Setback (ft.)	25	10		10					
Min. Rear Yard Setback (ft.)	25	10	From alley center line; 0 ft. if alley is dedicated plus 5 for ingress/ egress	10	10				
Max. Lot Cover (%)	25%	25%	100%	50%	50%				
Min. Common Open Space (%)					25%				
Max. Height (ft.)	25	25	35	30	35				

- (c) Minimum lot area per dwelling unit. Residential uses shall comply with the minimum lot area per dwelling unit standards contained in the district regulations and summarized in the Residential Area Regulations Schedule, as may be modified by additional provisions in the district regulations, in this subsection or elsewhere in this Chapter.
- (d) Minimum lot width. Lots used for residential uses shall comply with the minimum lot width standards contained in the district regulations and summarized in the Residential Area Regulations Schedule, as may be modified by additional provisions in the district regulations, in this subsection or elsewhere in this Chapter.
  - (1) Measurement. Lot width shall be the length of the minimum required front yard setback line between the two side lot lines. If the front yard setback line is an arc or a curve, the lot width shall be the length of said arc or curve.
  - (2) Cul-de-sacs. Notwithstanding any other provisions of this Title, lots fronting on a cul-de-sac shall have a minimum front street line of fifty feet (50') and a minimum lot width of seventy-five feet (75'), provided that this provision shall not apply in the PUD, Planned Unit Development Overlay District.
- (e) Minimum floor area per dwelling unit. Residential uses shall comply with the minimum floor area per dwelling unit standards contained in the district regulations and summarized in the Residential Area Regulations Schedule, as may be modified by additional provisions in the district

regulations, in this subsection or elsewhere in this Title.

Measurement. The floor area of a dwelling, for the purpose of these minimum floor area per dwelling unit requirements, refers to the total of the horizontal area of each floor, measured from the outside face of the building walls and excluding garages, carports, cellars and accessory buildings.

- (f) Minimum front yard setback. The location of buildings shall comply with the minimum front yard setback standards contained in the district regulations and summarized in the Residential and Nonresidential Area Regulations Schedules, as may be modified by additional provisions in the district regulations, in this subsection or elsewhere in this Title.
  - (1) Corner lots. For lots with frontage on two (2) intersecting streets, the front yard setback shall be the yard adjacent to the shorter of the two street lines, unless otherwise designated on the plat approved by the Planning and Zoning Commission or by Resolution approving the Planned Unit Development.
  - (2) Double frontage lots. Where lots have double frontage, running through from one street to another, a required front yard setback shall be provided on both streets unless a building line for accessory buildings has been established along one frontage on the plat or by Resolution, in which event only one (1) required front yard setback need be observed.
  - (3) Detached accessory buildings shall not be located within ten feet (10') of any other building.
- (g) Minimum side yard setback. The location of buildings shall comply with the minimum side yard setback standards contained in the district regulations and summarized in the Residential and Nonresidential Area Regulations Schedules, as may be modified by additional provisions in the district regulations, in this subsection or elsewhere in this Title.
  - (1) Corner lots. For lots with frontage on two (2) intersecting streets, the side yard setback shall be the yard adjacent to the longer of the two street lines, unless otherwise designated on the plat approved by the Planning and Zoning Commission or by Resolution, subject to the following provisions.
    - a. On a corner lot used for a single-family or duplex dwelling, both street exposures shall be treated as front yards on all lots platted.
    - b. On a corner lot used for a single-family or duplex dwelling that was an official Lot of Record, the minimum side yard adjacent to a side street shall comply with the minimum required side yard setback for the respective district.
  - (2) Accessory buildings. Detached accessory buildings shall have a side yard setback of not less than three feet (3') measured from any side lot line, alley line or easement line unless otherwise required by the International Building Codes. Detached accessory buildings shall not be located within ten feet (10') of any other building.

(h) Minimum rear yard setback. The location of buildings shall comply with the minimum rear yard setback standards contained in the district regulations and summarized in the Residential and Nonresidential Area Regulations Schedules, as may be modified by additional provisions in the district regulations, in this subsection or elsewhere in this Title.

Accessory buildings. Detached accessory buildings may be located within a required rear yard setback, provided that, if no alley exists, the rear yard setback shall not be less than ten feet (10') as measured from the rear lot line. Detached accessory buildings shall not be located closer than ten feet (10') to any other building.

- (i) **Permits**. The size of buildings shall comply with the maximum lot coverage standards contained in the district regulations and summarized in the Residential and Nonresidential Area Regulations Schedules, as may be modified by additional provisions in the district regulations, in this subsection or elsewhere in this Title.
- (j) Maximum height. The height of buildings and structures shall comply with the maximum height standards contained in the district regulations and summarized in the Residential and Nonresidential Area Regulations Schedules, as may be modified by additional provisions in the district regulations, in this subsection or elsewhere in this Title.
  - (1) Measurement. Height refers to the maximum possible upward distance to the top of a building, measured adjacent to a building at a right angle to the horizon line from each and every point on the finished grade
  - (2) Permitted exceptions.
    - a. The ridge of a gable, hip, gambrel or similar pitched roof may extend up to five feet above the specified maximum height limit;
    - b. Antennas, chimneys, flues, vents or similar structures may extend up to ten feet above the specified maximum height limit, and radio transmission antennas may extend above the specified maximum limit with the approval of the Planning and Zoning Commission;
    - c. Water towers and mechanical equipment may extend up to five feet above the specified maximum height limit;
    - d. Church spires, bell towers and like architectural features, as well as flag poles, may extend over the specified maximum height limit;
    - e. Schools, churches and public administration buildings in residential zone districts may exceed the established height limitation by fifty percent, provided they comply with all other requirements for the zone district and that the total floor area of the structure does not exceed the total area of the lot on which the building is located.

## § 3.07 A, Agricultural District

- (a) **Purposes**. The A, Agricultural District is designed primarily for agricultural uses and for single-family development on lots of two acres or greater if required by septic system requirements. It is also intended for application as Temporary Zoning following annexation but prior to final subdivision approval, as defined in subsection 3.04. In addition to the use and area regulations of this section, development in the A, Agricultural District shall be in compliance with all other applicable provisions of this Title.
- **(b) Permitted uses.** The following uses shall be permitted in the A, Agricultural District, subject to compliance with any applicable conditions and all other provisions of this Title.

Accessory use or structure

Animal production and husbandry, subject to Sec. 3.05(d)(4)

Bed and breakfast

Church or place of worship

Day care center

Day care home

Double-wide manufactured/ mobile home, subject to Sec. 3.05(d)(12)

Essential services

Earth, clay, rock or soil storage

Farm buildings (under 25' in height)

Fire station

Fruit or vegetable stand

Gas line and regulating station

Group home

Home occupation, subject to Sec. 3.05(d)(5)

Park

Veterinary clinic

School, elementary or secondary

Sewage pump station

Single-family dwelling

Stable, private, subject to Sec. 3.05(d)(8)

Water treatment, storage or pump station

(c) Conditional uses. The following uses shall be permitted in the A, Agricultural District only upon approval of a Conditional use permit by the Board of Trustees in accordance with the procedures and standards of Sec. 6.10, Conditional use permits.

Animal pound or kennel (public or private)

Boarding or rooming house

Caretaker or guard residence

Day camp for children

Dude ranch

Electrical substation, high voltage

**Equestrian Center** 

Events center or fairgrounds

Farm buildings (over 25' in height and under 35')

Fraternal club or lodge

Greenhouse or plant nursery

Heliport

Nursing home

Radio, TV or microwave tower

Rodeo grounds

Sewage treatment plant

Stable, commercial boarding or rental

(d) Temporary uses. The following uses shall be permitted in the A, Agricultural District only upon approval of a temporary use permit in accordance with the procedures and standards of Sec. 6.17, Temporary use permits.

Building material yard/construction office, temporary

Field office, temporary

Sales trailer, temporary, subject to Sec. 3.05(d)(7)

- (e) Minimum lot area. The minimum lot area for all uses in the A, Agricultural District shall be two (2) acres or 87,120 square feet, subject to Sec. 3.06(c).
- (f) Minimum lot width. The minimum lot width for all uses in the A, Agricultural District shall be two hundred feet (200'), subject to Sec. 3.06(d).
- (g) Minimum front yard setback. The minimum front Yard Setback for all uses in the A, Agricultural District shall be forty feet (40'), subject to Sec. 3.06(f).
- (h) Minimum side yard setback. The minimum side yard setback in the A, Agricultural District shall be twenty-five feet (25') for all uses, subject to Sec. 3.06(g):
- (i) Minimum rear yard setback. The minimum rear yard setback in the A, Agricultural District shall be ten feet (10') for residential uses and fifteen feet (15') for nonresidential uses, subject to Sec. 3.06(h).
- (j) Maximum lot coverage. The maximum lot coverage in the A, Agricultural District shall be twenty-five percent (25%) for all uses, subject to Sec. 3.06(i).
- (k) Maximum height. The maximum height of buildings and structures in the A, Agricultural District shall be twenty-five feet (25') for all uses, subject to Sec. 3.06(k).
- (l) Minimum floor area per unit. The minimum floor area per dwelling unit in the A, Agricultural District shall be eight hundred (800) square feet, subject to Sec. 3.06(e).

### § 3.08 R-1, Residential District

- (a) Purpose. The R-1, Residential District is designed to accommodate a variety of housing types and to promote diversity. It is intended for application in areas designated "Res. P-1 or Res. P-2" on the Future Land Use Map of the 1984 Comprehensive Plan. In addition to the use and area regulations of this section, development in the R-1, Residential District shall be in compliance with all other applicable provisions of this Title.
- **(b) Permitted uses**. The following uses shall be permitted in the R-1, Residential District, subject to compliance with any applicable conditions and all other provisions of this chapter.

Accessory use or structure, subject to Sec. 3.05(d)(1)

Bed and breakfast

Church or place of worship

Day care home

Double-wide manufactured/ mobile home, subject to Sec. 3.05(d)(12)

Duplex dwelling

Essential services

Fire station

Group home

Home occupation, subject to Sec. 3.05(d)(5)

Park

School, elementary or secondary

Single-family dwelling

Water treatment, storage or pump station

(c) Conditional uses. The following uses shall be permitted in the R-1, Residential District only upon approval of a Conditional use permit by the Board of Trustees in accordance with the procedures and standards of Sec. 6.10, Conditional use permits.

Alley house, accessory, subject to Sec. 3.05 (d) (14)

Antique shop

Boarding or rooming house

Community center, subject to Sec. 3.05 (d) (18)

Cottage Housing Development

Custom personal services

Day camp for children

Day care center

Electrical substation, high-voltage

Gas line and regulating station

Heliport

Hospital or clinic

Medium Density, subject to Sec. 3.05 (d) (19) and 3.06(a)

Multi-family dwelling

Nursing home

Stable, private, subject to Sec. 3.05(d)(8) Storage warehouse (mini-storage units), subject to Sec 3.05 (d)(16)

(d) Temporary uses. The following uses shall be permitted in the R-1, Residential District only upon approval of a temporary use permit in accordance with the procedures and standards of Sec. 6.17, Temporary use permits.

Asphalt/concrete batch plant, temporary, subject to Sec. 3.05(d)(3) Building material yard/construction office, temporary Field office, temporary Sales trailer, temporary, subject to Sec. 3.05(d)(7)

- (e) Minimum lot area per dwelling. The minimum lot area per dwelling unit in the R-1, Residential District shall be as follows, subject to Sec. 3.06(c):
  - (1) Seventy-five hundred (7,500) square feet per single-family dwelling unit or five thousand (5,000) square feet per single-family dwelling for lots created prior to adoption of Ordinance #256 on 12-03-86, and
  - (2) Five thousand (5,000) square feet per duplex dwelling unit, and
  - (3) Two Thousand eight hundred (2,800) square feet per multi-family dwelling unit.
  - (4) Two thousand five hundred (2,500) square feet per Cottage Housing Unit.
  - (5) Alley House: Seven thousand five hundred (7,500 square feet) for a combination single family dwelling unit and alley house, provided that an alley house is constructed after or concurrently with a single family dwelling unit on the same lot.
- **Minimum lot width.** The minimum lot width in the R-1, Residential District shall be as follows, subject to Sec. 3.06(d):
  - (1) forty-eight feet (48') for single-family dwellings, and
  - (2) sixty feet (60') for duplex or multi-family dwellings, and
  - (3) thirty-five feet (35') for cottage housing.
- (g) Minimum front yard setback. The minimum front yard setbacks for all uses in the R-1, Residential District shall be fifteen feet (15'), subject to Sec. 3.06(f).
- (h) Minimum side yard setback. The minimum side yard setbacks for all uses in the R-1, Residential District shall be ten feet (10') for all uses, subject to Sec. 3.06(g). For Cottage Housing Developments, the setback shall be an average of ten (10) feet but no less than five (5) feet, subject to Sec. 3.06(g).
- (i) Minimum rear yard setback. The minimum rear yard setbacks in the R-1 Residential District

shall be as follows, subject to Sec. 3.06(h):

- (1) none (0') for single family dwellings, duplex dwellings, alley houses, non-residential and mixed uses when the rear yard is adjacent to an alley, and
- (2) ten feet (10') for multi-family dwellings, Alley houses and for all uses when the rear yard is not adjacent to an alley.
- (3) For Cottage Housing Developments, the setback shall be an average of ten (10) feet but no less than five (5) feet.
- (j) Maximum lot coverage. The maximum lot coverage in the R-1, Residential District shall be forty (40%) percent for single family and Cottage Housing Development uses and twenty-five percent (25%) for all other uses, subject to Sec. 3.06(i). The maximum lot coverage shall be fifty percent (50%) or five thousand (5000) square feet, whichever is less, for a single family dwelling unit combined with an Alley house, provided that an alley house is constructed after or concurrently with a single family dwelling unit on the same lot.
- (k) Maximum height. The maximum height of buildings and structures in the R-1, Residential District shall be twenty-five feet (25') for the primary use and eighteen (18) feet for accessory uses, Cottage Dwelling Units and Alley houses and all uses, subject to Sec. 3.06(j).
- (I) Minimum floor area per unit. The minimum floor area in the R-1, Residential District shall be as follows, subject to Sec. 3.06(e):
  - (1) eight hundred (800) square feet for single family dwelling units,
  - (2) five hundred (500) square feet for duplex dwelling units, and Cottage Dwelling units.
  - (3) three-hundred fifty (350) square feet for multi-family dwelling units and Alley houses.

### § 3.09 MH, Mobile Home District

- (a) Purpose and intent. The MH, Mobile Home District is designed primarily to accommodate mobile home parks and mobile home subdivisions. It is intended for application in areas currently being used for mobile home parks and for selective application in the future to areas designated "Res. P-1 or Res. P-2" on the "Future Land Use" map of the Comprehensive Plan. In addition to the use and area regulations of this section, development in the MH, Mobile Home District shall be in compliance with all other applicable provisions of this Title.
- **(b) Permitted uses**. The following uses shall be permitted in the MH, Mobile Home District, subject to compliance with any applicable conditions and all other provisions of this Title.

Accessory use or structure, subject to Sec. 3.05(d)(1)

Church or place of worship

Day care home

Double-wide manufactured/ mobile home, subject to Sec. 3.05(d)(12)

Essential services

Fire station

Gas line and regulating station

Home occupation, subject to Sec. 3.05(d)(5)

Mobile home park, subject to Sec. 3.05(d)(6)

Mobile home subdivision, subject to Sec. 3.05(d)(6)

Park

Sewage pump station

Single-family dwelling

Water treatment, storage or pump station

(c) Conditional uses. The following uses shall be permitted in the MH, Mobile Home District only upon approval of a Conditional use permit by the Board of Trustees in accordance with the procedures and standards of Sec. 6.10, Conditional use permits.

Community center, subject to Sec. 3.05 (d) (18)

Day camp for children

Day care center

Electrical substation, high voltage

Gas line and regulating station

(d) Temporary uses. The following uses shall be permitted in the MH, Mobile Home District only upon approval of a temporary use permit in accordance with the procedures and standards of Sec. 6.17, Temporary use permits.

Asphalt/concrete batch plant, temporary, subject to Sec. 3.05(d)(3)

Building material yard/construction office, temporary

Field office, temporary

Sales trailer, temporary, subject to Sec. 3.05(d)(7)

- (e) Minimum lot area per dwelling. The minimum lot area per dwelling unit in the MH, Mobile Home District shall be as follows, subject to Sec. 3.06(c):
  - (1) Five thousand (5,000) square feet per single-family dwelling unit, and
  - (2) Three thousand (3,000) square feet per mobile home dwelling unit.
- (f) Minimum lot width. The minimum lot width for residential uses in the MH, Mobile Home District shall be as follows, subject to Sec. 3.06(d):
  - (1) Forty-eight feet (48') for single-family dwellings, and
  - (2) Thirty-five (35') for mobile home dwellings.
- (g) Minimum front yard setback. The minimum front yard setback for all uses in the MH, Mobile Home District shall be twenty-five feet (25'), subject to Sec. 3.06(f).
- (h) Minimum side yard setback. The minimum side yard setback for all uses in the MH, Mobile Home District shall be ten feet (10'), subject to Sec. 3.06(g).
- (i) Minimum rear yard setback. The minimum rear yard setback setbacks in the MH, Mobile Home District shall be as follows, subject to Sec. 3.06(h):
  - (1) Ten feet (10') for mobile home dwelling units
  - (2) None for single family dwelling units when the rear yard is adjacent to an alley.
  - (3) Ten feet (10') for single family dwelling units when the rear yard is not adjacent to an alley.
- **Maximum lot coverage**. The maximum lot coverage in the MH, Mobile Home District shall be as follows, subject to Sec. 3.06(i):
  - (1) fifty percent (50%) for mobile home dwellings, and
  - (2) twenty-five percent (25%) for all other uses
- (k) Maximum height. The maximum height of buildings and structures in the MH, Mobile Home District shall be twenty-five feet (25') for all uses, subject to Sec. 3.06(j).
- (l) Minimum floor area per unit. The minimum floor area per single family dwelling in the MH, Mobile Home District shall be eight hundred (800) square feet, subject to Sec. 3.06(e). There shall be no minimum floor area per mobile home dwelling unit.

## § 3.10 B-1, Business District

- (a) Purpose and intent. The B-1, Business District is designed primarily to accommodate a wide variety of residential, office, retail and repair/service uses as may be appropriate for a mixed-use community center. The B-1, Business District, is intended to accommodate the majority of retail uses within the Town, with residential uses located above and/or behind commercial uses on a lot. It is intended for application in areas designated "Commercial" on the future land use map of the Comprehensive Plan. In addition to the use and area regulations of this section, development in the B-1, Business District shall be in compliance with all other applicable provisions of this Title.
- **(b) Permitted uses.** The following uses shall be permitted in the B-1, Business District, subject to compliance with any applicable conditions and all other provisions of this Title. Further, notwithstanding any contrary term or provision herein, in the event that any use, regardless of classification, meets the definition of Formula Business Establishment it may only be permitted as a conditional use subject to special use conditions 13 and 21 of Section 3.05(d) in addition to any other special use condition applicable to that certain use.

Accessory use or structure, subject to Sec. 3.05(d)(1)

Amusement, commercial, subject to Sec. 3.05(d)(2)

Antique shop

Art gallery or handicraft sales

Art supply store

Auto parts and accessory sales (indoors)

Bakery or confectionery shop

Bank or savings and loan

Bar

Bed and Breakfast

Book or stationary store or newsstand

Bus station or terminal

Church or place of worship

Cleaning shop or laundry pick-up station

Convenience store

Custom personal services

Day care center

Day care home

Department or variety store

Drug store or pharmacy

Essential services

Fabric, drapery or needlework store

Fire station

Fraternal club or lodge

Furniture or appliance store

Gasoline service station

Greenhouse or plant nursery

Group home

Feed store

**Florist** 

Fruit or vegetable stand

Hardware store and hobby shop

Hotel or motel

Household appliance service and repair

Key shop

Laundry, self-service

Library

Liquor store

Mortuary or funeral home

Multi-family dwelling (see Note 1)

Museum

Office, business and professional

Optical sales

Paint and wallpaper store

Park

Pet store

Photography shop

Print shop or copy shop

Public building, shop or yard

Recreational vehicle/travel trailer park, subject to Sec. 3.05(d)(11)

Repair services (limited)

Restaurant, general

Retail, general (indoors)

Secondhand store or pawn shop

Sporting goods

Studio, television or radio

Supermarket

Theater

Tool rental (domestic equipment)

Toy store or hobby shop

Travel bureau or consultant

Veterinary clinic

**(c) Conditional uses.** The following uses shall be permitted in the B-1, Business District only upon approval of a Conditional use permit by the Board of Trustees in accordance with the procedures and standards of Sec. 6.10, Conditional use permits. Further, notwithstanding any contrary term or provision herein, in the event that any use, regardless of classification, meets the definition of Formula Business Establishment it may only be permitted as a conditional use subject to special use conditions 13 and 21 of Section 3.05(d) in addition to any other special use condition applicable to that certain use.

Animal pound or kennel (public or private)

Boat service

Boarding or rooming house

Campground

Car wash

Community center, subject to Sec. 3.05 (d) (18)

Duplex dwelling (See note 1)

Heavy commercial uses as identified in §3.05(c), subject to §3.05(d) (13)

Home occupation, subject to §3.05(d)(5)

Hospital or clinic

Manufacturing, Light, subject to §3.05(d)(13) and (20)

New or used car or truck sales

Nursing home

Parking lot, trucks or trailers (transport)

Recreational vehicle/travel trailer park

Sewage pump station

Single family dwelling unit (See note 1)

Snowmobile, motorcycle, scooter or ATV sales and service

Hospital

New or used car or truck sales

Tattoo parlor

Water treatment, storage or pump station

# **NOTE 1:** Restriction on Residential Dwellings.

No residential use as defined in Section (b) or (c) above, as Uses Permitted by Right (b) or Uses permitted as Conditional Uses (c) shall be permitted within the first or ground floor of a building or structure (that floor which is closest to the elevation of the adjacent street), unless such use is setback an average of thirty-five (35') from the front exterior wall of the building or structure as measured perpendicularly from the front exterior wall. In no case shall such use be closer than twenty feet (20') from the front exterior wall. For the purposes of this section, the "front exterior wall" of building or structure shall mean that wall which establishes the front yard, or in the absence of a front yard, the front lotline. Also, for the purpose of this section, the "front lotline" shall mean the frontage along the street with the shortest linear frontage for lots which have more than one street frontage. Provided, however, that single family and duplex units located in areas zoned B-1 to the east of Market Street and to the west of Spruce Street need not comply with the setback requirements provided for herein.

**(d) Temporary uses**. The following uses shall be permitted in the B-1, Business Center District only upon approval of a temporary use permit in accordance with the procedures and standards of Sec. 6.17, Temporary use permits.

Field office, temporary Building material yard/construction office, temporary Sales trailer, temporary, subject to Sec. 3.05(d)(7)

(e) Minimum lot area. The minimum lot area in the B-1, Business District shall be five thousand (5,000) square feet; seven thousand-five hundred (7,500) square feet per single family dwelling units and five thousand (5,000) square feet per duplex and multi-family dwelling unit.

- (f) Minimum lot width. There shall be no minimum lot width for non-residential and mixed use projects in the B-1 Zone District. The minimum lot width for residential uses in the B-1, Business District shall be as follows, subject to Sec. 3.06(d):
  - (1) Forty-eight feet (48') for single-family dwellings, and
  - (2) Sixty feet (60') for all other residential uses.
- (g) Minimum floor area per dwelling unit. The minimum floor area per dwelling unit in the B-1, Business District shall be as follows, subject to Sec. 3.06(e):
  - (1) Eight hundred (800) square feet per single-family dwelling unit,
  - (2) Five hundred (500) square feet per duplex dwelling unit, and
- (h) Minimum front yard setback. The minimum front yard setback for residential uses in the B-1, Business District shall be as follow fifteen feet (15') and no requirement for nonresidential and mixed use structures, subject to Sec. 3.06(f).
- (i) Minimum side yard setback. The minimum side yard setback in the B-1, Business District shall as follows, subject to Sec. 3.06(g):
  - (1) None for non-residential and mixed use structures with masonry exterior walls, subject to compliance with the International Building Codes, and
  - (2) Ten feet (10') for all residential uses.
- (j) Minimum rear yard setback. The minimum rear yard setback in the B-1, Business District shall be none (0') for single family dwellings and duplex structures, and ten feet (10') from centerline of alley for non-residential, commercial uses, mixed and multi-family dwellings, for property adjacent to an alley for which an easement or dedication of property has not been granted to the Town subject to Sec. 3.06 (h). All structures shall be setback a minimum of five feet (5') for ingress and egress.
- (k) Maximum lot coverage. The maximum lot coverage in the B-1, Business District shall be twenty-five percent (25%) for single family and duplex residential uses up to one hundred percent (100%) for all non-residential, multi-family, and mixed use structures, subject to Sec. 3.06 (i).
- (l) Maximum height. The maximum height of buildings and structures in the B-1, Business District shall be twenty-five feet (25') for single family and duplex residential uses and thirty-five (35') for all other uses, subject to Sec. 3.06 (j).
- (m) Minimum floor area per unit. The minimum floor area per single family dwelling in the B-1, Business District shall be eight hundred (800) square feet, five hundred (500) per duplex dwelling unit, subject to Sec. 3.06(e).

## § 3.11 I, Industrial District

- (a) **Purpose and intent**. The I, Industrial District is designed primarily to accommodate a wide variety of repair/service and light industrial uses. It is intended for application in areas designated "Light Industrial" on the "Future Land Use" map of the Comprehensive Plan. In addition to the use and area regulations of this section, development in the I, Industrial District shall be in compliance with Sec. 5.05, Landscaping and Screening, and all other applicable provisions of this Title.
- **(b) Permitted uses**. The following uses shall be permitted in the I, Industrial District, subject to compliance with any applicable conditions and all other provisions of this Title.

Accessory use or structure, subject to Sec. 3.05(d)(1)

Animal pound or kennel (public or private)

Auto parts and accessory sales (indoors)

Auto leasing or rental

Auto repair garage

Auto parts and sales (outdoor display)

Bakery or confectionery shop

Bank or savings and loan

Building materials sales and yard

Bus or intercept lot

Cabinet and upholstery shop

Campground

Car wash

Caretaker or guard residence

Church or place of worship

Cleaning or laundry plant

Convenience store

Custom personal services

Day care center

Day care home

Day camp for children

Educational facilities

Fraternal club or lodge

Fruit or vegetable stand

Furniture or appliance store

Gasoline service station

Greenhouse or plant nursery

Hardware store and hobby shop

Heating or air conditioning shop

Household appliance service and repair

Key shop

Laundry, self service

Lithographic or print or copy shop

Maintenance and repair service for buildings

Mini-warehouse- equal to or less than 2500 square feet

Motorcycle or scooter sales and service

Museum

New or used car or truck sales

Office, business and professional

Park

Pet store

Plumbing shop

Public building, shop or yard

Radio, TV or microwave tower (10)

Recreational vehicle storage and service

Recreational vehicle/travel trailer park with less than twenty (20) spaces, for visitor stays of less than 30 consecutive days

Repair services, limited

Retail general (outdoors)

Single family dwelling unit

Snowmobile, motorcycle or scooter sales and service

Storage or hauling company

Storage warehouse

Studio, television or radio

Tool rental (domestic equipment)

Trailer or mobile home sales or rental

Travel bureau or consultants

Veterinary clinic

Welding and machine shop

**(c) Conditional uses.** The following uses shall be permitted in the I, Industrial District only upon approval of a conditional use permit by the Board of Trustees in accordance with the procedures and standards of Sec. 6.10, Conditional use permits.

Community center, subject to Sec. 3.05 (d) (18)

Mini-warehouse greater than 2500 square feet

Motel

Nursing home

Private utility shop, yard or building

Private franchise utility (not listed)

Recreational vehicle or travel trailer park – twenty (20) spaces or greater, for visitor stays of less than 30 consecutive days

School, commercial trade

Supermarket

Theater

(d) Temporary uses. The following uses shall be permitted in the I, Industrial District only upon approval of a temporary use permit in accordance with the procedures and standards of Sec. 6.17, Temporary use permits.

Asphalt/concrete batch plant, temporary, subject to Sec. 3.05(d)(3) Building material yard/construction office Field office, temporary Sales trailer, temporary, subject to Sec. 3.05(d)(7)

- (e) Minimum lot area. The minimum lot area requirement shall be six thousand square feet (6,000 s.f.) in the I, Industrial District.
- (f) Minimum lot width. The minimum lot width requirement shall be fifty feet (50') in the I, Industrial District.
- **(g) Minimum floor area per unit**. There shall be no minimum floor area per dwelling unit requirement in the I, Industrial District.
- (h) Minimum front yard setback. The minimum front yard setback requirement shall be fifteen feet (15') in the I, Industrial District, subject to Sec. 3.06(f).
- (i) Minimum side yard setback. The minimum side yard setback requirement shall be ten feet (10') for all uses in the I, Industrial District, subject to Sec. 3.06(g).
- (j) Minimum rear yard setback. The minimum rear yard setback in the I, Industrial District shall be ten feet (10') for all uses, subject to Sec. 3.06(h).
- (k) Maximum lot coverage. The maximum lot coverage in the I, Industrial District shall be fifty percent (50%) for all uses, subject to Sec. 3.06(i).
- (l) **Maximum height**. The maximum height of buildings and structures in the I, Industrial District shall be thirty feet (30') for all uses, subject to Sec. 3.06(j).

### § 3.12 P, Public District

- (a) **Purpose and intent**. The P, Public District is designed primarily to accommodate the development of governmental and quasi-governmental facilities for the cultural, education, civic, recreation and other governmental purposes. It is intended for application to areas that are currently publicly owned. In addition to the use and area regulations of this section, development in the P, Public District shall be in compliance with all other applicable provisions of this Title.
- **(b) Permitted uses.** The following uses shall be permitted in the P, Public District, subject to compliance with any applicable conditions and all other provisions of this Title.

Accessory use or structure, subject to Sec. 3.05(d)(1)

Caretaker or guard residence

Church or place of worship

Day camp for children

Day care center

Day care home

Fire station

**Essential services** 

Events center or Fairgrounds

Equestrian center

Fraternal club or lodge

Heliport

Library

Museum

Park

Public building, shop or yard

Radio, TV or microwave tower

Sewage pump station

School, elementary or secondary

Studio, radio or television

Water treatment, storage or pump station

(c) Conditional uses. The following uses shall be permitted in the P, Public District only upon approval of a conditional use permit by the Board of Trustees in accordance with the procedures and standards of Sec. 6.10, Conditional use permits.

Campground

Community center, subject to Sec. 3.05 (d) (18)

Correctional facility or jail, public

Gas line and regulating station

Hospital or clinic

Recreational Vehicle/Travel Trailer Park

Rodeo Grounds

(d) **Temporary uses**. The following uses shall be permitted in the P, Public District only upon

approval of a temporary use permit in accordance with the procedures and standards of Sec. 6.17, Temporary use permits.

Asphalt/concrete batch plant, temporary, subject to Sec. 3.05(d)(3) Building material yard/construction office Field office, temporary

- (e) Minimum lot area. There shall be no minimum lot area requirement in the P, Public District.
- (f) Minimum lot width. There shall be no minimum lot width requirement in the P, Public District.
- **Minimum front yard setback**. The minimum front yard setback requirement in the P, Public District shall be 10 feet (10').
- (h) Minimum side yard setback. The minimum side yard setback requirement in the P, Public District, shall be 10 feet (10').
- (i) Minimum rear yard setback. The minimum rear yard setback in the P, Public District shall be ten feet (10') for all uses, however, no rear yard setback shall be required for non-residential buildings if an alley is located adjacent the rear lot line.
- (j) Maximum lot coverage. The maximum lot coverage in the P, Public District shall be sixty percent (60%) for all uses, subject to Sec. 3.06(i).
- (k) Maximum height. The maximum height of buildings and structures in the P, Public District shall be thirty-five feet (35') for all uses, subject to Sec. 3.06(j).

## § 3.13 PUD, Planned Unit Development Overlay District

- (a) Purpose and intent. The Planned Unit Development Overlay District is a rezoning process that, upon approval results in a detailed plan that is the basis for development of a particular property. An application for a PUD plan approval may be made for land located in any zoning district. The PUD Planned Unit Development Overlay District is designed primarily to provide flexibility to accommodate mixed-use, nontraditional and innovative developments and to encourage more efficient utilization of sites and the preservation of usable open space. The PUD shall be consistent with the requirements of the zone district in which the parcel is located unless a variation or variations of the dimensional limitations have been granted pursuant to this section. In addition to the use and area regulations of this section, development in the PUD Planned Unit Development Overlay District shall comply with all other provisions of this Land Use Code.
- (b) Types of planned unit developments. The Board of Trustees of the Town of Norwood, Colorado, after public hearing and proper notice and after recommendation from the Planning and Zoning Commission, may authorize the creation of the Planned Unit Development Overlay Districts on parcels of land containing more than 20,000 square feet. A Planned Unit Development Overlay designation may be applied to land intended for residential, commercial or other development purposes.
- (c) Site plan requirement. In establishing a Planned Unit Development Overlay District in accordance with this section, the Planning and Zoning Commission shall require a comprehensive site plan of the development. Such site plan shall be approved as part of the resolution approving a Planned Unit Development Overlay and recorded as a burden to run with the land. Such required plan and resolution shall set forth the requirements for ingress and egress to the property, public streets or drives, with adequate right-of-way, sidewalks, utilities, drainage, parking space, height of building, maximum lot coverage, yards and open spaces, screening walls or fences, landscaping and other development and protective requirements including maintenance considered necessary to create a reasonable transition to and protection of the adjacent property.
- (d) Procedures for approval and conditions. Every Planned Unit Development Overlay District approved under the provisions of this chapter shall follow the procedure of Sec. 6.02 for and be considered an amendment to the zoning map. In approving the Planned Unit Development Overlay District, the Board of Trustees may impose conditions relative to the standard of development and such conditions shall be complied with before a Zoning Development Permit is issued for the use of the land or any structure that is part of the Planned Unit Development Overlay District and such conditions shall not be construed as conditions precedent to the approval of the zoning amendment, but shall be construed as conditions precedent to the granting of a Zoning Development Permit. All Planned Unit Development Overlay Districts approved in accordance with the provisions of this chapter shall be referenced on the Official Zoning Map and a list of such Planned Unit Development Overlay Districts, together with the category of uses permitted therein, shall be maintained in the office of the Town Clerk.
- (e) Status of prior and future approvals.

- (1) In the Planned Unit Development Overlay Districts previously approved, where the Board of Trustees approved conditional minimum development requirements for single-family dwellings that are less stringent than the minimum development requirements contained in this chapter for single-family dwellings in Planned Unit Development Overlay Districts, the less stringent minimum development requirements set forth in such Planned Unit Development Overlay Districts are hereby ratified.
- (2) In all Planned Unit Development Overlay Districts previously approved that are silent or contain no conditional minimum development requirements for single-family dwellings, the minimum development requirements contained in this chapter for single-family dwellings in Planned Unit Development Overlay Districts shall apply.
- **(f) Permitted, conditional and temporary uses**. The land uses permitted shall be those of the underlying zone district.
- **(g) Maximum density**. The maximum density shall be no greater than that permitted in the underlying zone district prior to PUD approval. Furthermore, densities may be reduced if:
  - (1) There is not sufficient water pressure and other utilities to service the proposed development;
  - (2) There are not adequate roads to ensure fire protection, snow removal and road maintenance to the proposed development;
  - (3) The land is not suitable for the proposed development because of ground reasons of soil or geologic conditions;
  - (4) The proposed development will have a deleterious effect on air quality in the surrounding area and the Town; or
  - (5) The design and location of any proposed structure, road, or driveway in the proposed development is not compatible with surrounding land uses, would adversely affect the character of the Town or cause harmful disturbance to critical natural features of the site.
- (h) Minimum lot area per dwelling. The minimum lot area for non-residential uses shall be four thousand feet (4000') in the PUD, Planned Unit Development Overlay District shall be as follows, subject to Sec. 3.06(c):
  - (1) Four thousand (4,000) square feet per single-family dwelling unit,
  - (2) Three thousand (3,000) square feet per duplex dwelling unit,
  - (3) Two thousand eight hundred (2,800) square feet per multi-family dwelling unit.
  - (4) Three thousand (3,000) square feet per mobile home dwelling unit.

- (5) Two thousand five hundred (2,500) square feet per Cottage Dwelling Unit, provided no less than four units nor more than nine units may be nine cottage units may be located in a given application.
- (i) Minimum lot width. The minimum lot width for non-residential uses shall be forty feet (40') in the PUD, Planned Unit Development Overlay District and for residential uses the minimum lot width shall be as follows, subject to Sec. 3.06(c):
  - (1) Forty feet (40') for single-family dwellings, and
  - (2) fifty feet (50') for duplex and multi-family dwellings, and
- (j) Minimum front yard setback. The minimum front yard setback requirement shall be fifteen feet (15') in the PUD, Planned Unit Development Overlay District, subject to Sec. 3.06(f). Where the front yard is not specified for an approved PUD, the minimum front yard setback shall be established according to the standards of the underlying zone district.
- (k) Minimum side yard setback. The minimum side yard setback requirements in a PUD, Planned Unit Development Overlay shall be as established by the approved PUD. Where the side yard is not specified for an approved PUD, the minimum side yard setback shall be established according to the standards of the underlying zone district.
- (l) Minimum rear yard setback. The minimum rear yard setback in a PUD, Planned Unit Development Overlay District shall be established by the approved PUD. Where the rear yard is not specified for an approved PUD, the minimum side yard setback shall be established according to the standards of the underlying zone district. The minimum rear yard setback for multi-family dwellings in Planned Unit Development Overlay Districts shall be as follows:
  - (1) Ten feet (10') from the foundation line to the rear property line. For cottage housing developments, five (5) feet from the rear or side property line.
  - (2) On lots backing up to any other single-family residential district, and not separated by a public street, a minimum rear yard setback of twenty feet (20') shall be provided.
  - (3) A minimum distance of ten feet (10') from foundation line to foundation line shall be provided between all buildings.
- (m) Maximum lot coverage. The maximum lot coverage in the PUD, Planned Unit Development Overlay District shall be fifty percent (50%) for all uses except for cottage housing developments, which shall be forty (40) percent, subject to Sec. 3.06(i).
- (n) Maximum height. The maximum height of buildings and structures in the PUD, Planned Unit Development Overlay District shall be three (3) stories or thirty-five feet (35') for all uses, subject to Sec. 3.06(j), provided that the maximum height for single-family dwellings shall be twenty-five feet (25'). Cottage dwelling units shall not exceed fifteen (15) feet.

- (o) Minimum floor area per unit. The minimum floor area per single family dwelling in the PUD, Planned Unit Development Overlay District shall be eight hundred (800) square feet, five hundred (500) per duplex dwelling unit and Cottage dwelling unit, subject to Sec. 3.06(e).
- (p) Minimum common open space. The minimum common open space in the PUD, Planned Unit Development Overlay District shall be twenty-five percent (25%).
- (q) Site plan notations. For any PUD in which zone district variations or density transfer for one portion of the site to another are approve, a note shall be placed on the site plan sufficient to place purchasers of all or part of the tract on notice of how exactly the zoning requirements and densities allowed on the property may differ from underlying zoning requirements.
- (r) PUD agreement. For any PUD in which variations are granted, public improvements required, or for PUD's involving numerous or complex condition of approval the applicant shall submit a PUD agreement describing the variances, public improvements required and conditions of approval which agreement shall provide guarantees for the improvements as required of subdividers in Section 6.07. The PUD agreement shall run with and be a burden on the land to which it applies.
- **Zone district map.** An approved PUD plan shall be illustrated and designated as overly on the official zone district map. Permanent underlying zoning designation and regulations shall remain unchanged and in the event a PUD is not completed or is terminated, the underlying zone district requirements shall apply.

## § 3.14 Medium Density Residential District

- (a) Purpose. The MD, Medium Density Residential District is designed to accommodate a variety of housing types and to promote diversity. It is intended for application in areas in Norwood where services, utilities and roads can accommodate greater density as referenced in the Future Land Use Plan. In addition to the use and area regulations of this section, development in the MD, Medium Density Residential District shall be in compliance with all other applicable provisions of this Title.
- **(b) Permitted uses**. The following uses shall be permitted in the MD, Medium Density Residential District, subject to compliance with any applicable conditions and all other provisions of this Chapter.

Accessory use or structure, subject to Sec. 3.05(d) (1)

Day care home

Double-wide manufactured/ mobile home, subject to Sec. 3.05(d) (12)

Duplex dwelling

Group home

Home occupation, subject to Sec. 3.05(d) (5)

Park

School, elementary or secondary

Single-family dwelling

(c) Conditional uses. The following uses shall be permitted in the MD, Medium Density Residential District only upon approval of a Conditional use permit by the Board of Trustees in accordance with the procedures and standards of Sec. 6.10, Conditional use permits.

Boarding or rooming house

**Community Center** 

Cottage Housing Development

Custom personal services

Day camp for children

Day care center

Multi-family dwelling

Nursing home

(d) Temporary uses. The following uses shall be permitted in the MD, Medium Density Residential District, only upon approval of a temporary use permit in accordance with the procedures and standards of Sec. 6.17, Temporary use permits.

Asphalt/concrete batch plant, temporary, subject to Sec. 3.05(d) (3)

Building material yard/construction office, temporary

Field office, temporary

Sales trailer, temporary, subject to Sec. 3.05(d) (7)

- (e) Minimum lot area per dwelling. The minimum lot area per dwelling unit in the MD, Medium Density Residential District shall be as follows, subject to Sec. 3.06(c):
  - (1) Thirty-five hundred (3,500) square feet per single-family unit, and
  - (2) Thirty-five hundred (3,500) square feet per duplex dwelling unit, and
  - (3) Two Thousand five hundred (2,500) square feet multi-family dwelling unit.
  - (4) Two thousand five hundred (2,500) square feet per Cottage Housing Unit.
- (f) Minimum lot width. The minimum lot width in the MD, Medium Density Residential District shall be as follows, subject to Sec. 3.06(d):
  - (1) forty feet (40') for single-family dwellings, and
  - (2) sixty feet (60') for duplex or multi-family dwellings.
- **(g) Minimum front yard setback**. The minimum front yard setbacks for all uses in the MD, Medium Density Residential District shall be fifteen feet (15'), subject to Sec. 3.06(f).
- (h) Minimum side yard setback. The minimum side yard setbacks for all uses in the MD, Medium Density Residential District shall be an average of ten (10) feet but no less than five (5) feet, subject to Sec. 3.06(g).
- (i) Minimum rear yard setback. The minimum rear yard setbacks in the MD, Medium Density Residential District shall be as follows, subject to Sec. 3.06(h):
  - (1) none (0') for single family dwellings, duplex dwellings, non-residential and mixed uses when the rear yard is adjacent to an alley, and
  - (2) ten feet (10') for multi-family dwellings and for all uses when the rear yard is not adjacent to an alley.
  - (3) For Cottage Housing Developments, the setback shall be an average of ten (10) feet but no less than five (5) feet.
- (j) Maximum lot coverage. The maximum lot coverage in the MD, Medium Density Residential District shall be forty-five (45%) percent for single family and Cottage Housing Development uses and forty percent (40%) for all other uses, subject to Sec. 3.06(i).
- (k) Maximum height. The maximum height of buildings and structures in the MD, Medium Density Residential District shall be thirty feet (30') for all uses, subject to Sec. 3.06(j). Cottage Dwelling Units shall not exceed eighteen (18) feet.
- (l) Minimum floor area per unit. The minimum floor area in the MD, Medium Density Residential District shall be as follows, subject to Sec. 3.06(e):
  - (1) eight hundred (800) square feet for single family dwelling units,

- (2) five hundred (500) square feet for duplex dwelling units, and Cottage Dwelling units.
- (3) three-hundred fifty (350) square feet for multi-family dwelling units.

## **Chapter 4.00 SUBDIVISION STANDARDS**

# § 4.01 Scope and Applicability

- (a) Applicability. All plats and subdivision of land within the corporate limits of the Town of Norwood, and all land outside the corporate limits of the Town of Norwood that the Board of Trustees may be petitioned to include within the corporate limits of the Town, by an extension of said corporate limits, shall conform to the following rules and regulations.
- (b) Creation of building site. No permit for the construction of a building or building upon any tract or lot shall be issued until a building site, building tract or building lot has been created by compliance with one of the following conditions:
  - (1) The lot or tract is part of a plat of record, approved by the Board of Trustees of the Town of Norwood and filed in the plat records of San Miguel County, Colorado; or
  - (2) The parcel, tract or lot faces upon a dedicated street and was separately owned prior to the effective date of the original subdivision regulations of the Town or Norwood or prior to annexation to the Town or Norwood, whichever is applicable. Any further subdivision of said parcel tract or lot shall be in conformance with the Town subdivision regulations then in effect. Any development of nonconforming lots must satisfy the requirements of Section 1.06 (b).

# (c) Platting property not permanently zoned.

- (1) The Planning and Zoning Commission of the Town of Norwood shall not approve any plat of any subdivision within the Town limits of the Town of Norwood until the area covered by the proposed plat shall have been permanently zoned by the Board of Trustees of the Town of Norwood, Colorado.
- (2) The Planning and Zoning Commission of the Town of Norwood shall not approve any plat of any subdivision within any area where a petition or Ordinance for annexation or a recommendation for annexation to the Town of Norwood is pending before the Board of Trustees.
- (3) In the event the Planning and Zoning Commission holds a hearing on proposed annexation, it may, at its discretion, at the same time hold a hearing upon the permanent zoning that is to be given to the area or tract to be annexed, and make a recommendation on both matters to the Board of Trustees so that the Board of Trustees can, if it desires, act on the matter of permanent zoning and annexation at the same time.

## § 4.02 Building Lots

(a) Lot configuration. The lot size, width, depth, shape and orientation, and the minimum building setback lines shall be appropriate for the location of the subdivision and for the type of development and use contemplated and shall not be less than those specified as minimum standards by the zoning district.

The depth and width of properties reserved or laid out for commercial and industrial purposes shall be adequate to provide for the off-street service and parking facilities required by the type of use and development contemplated.

- **Side lot lines**. Side lot lines shall be substantially at right angles to street lines unless otherwise approved by the Planning and Zoning Commission.
- (c) Street frontage required. Each lot or building tract shall front upon a public street.
- (d) **Double frontage lots**. Double frontage lots shall be avoided, except where essential to provide separation of residential development from traffic or to overcome specific disadvantages of topography and orientation.
- (e) Large lots. Where the area is divided into larger lots than for normal town building sites and, in the opinion of the Planning and Zoning Commission, any or all of the tracts are susceptible of being resubdivided, the original subdivision shall be such that the alignment of future street dedications may conform to the general street layout in the surrounding area and so that the larger tracts may be later subdivided in conformance with the requirements of this chapter and the minimum standards specified by the zoning district.

## § 4.03 Streets and Alleys

- (a) Applicability. The following design standards shall be required for all subdivisions.
- (b) Street surfaces. On and after the passage of these regulations, all developers shall be required to construct asphalt streets with curbs and gutters and a road base with gravel alleys in accordance with the Town of Norwood Road Design Standards and Regulations.
- (c) Street layout. Unless otherwise approved by the Planning and Zoning Commission, provisions shall be made for the extension of streets in accordance with the Major Streets and Future Land Use Plan for street extension and shall bear a logical relationship to the topography and to the location of existing or planned streets on adjacent properties. Adequate local streets shall be provided to accommodate the subdivision and provide access to lots. Where the layout of streets is not shown in an approved streets plan, the arrangement of streets in a subdivision shall either:
  - (1) Provide for the continuation or appropriate projection to existing principal streets in surrounding areas; or
  - (2) Conform to a plan for a neighborhood or planned unit development approved or adopted by the Planning and Zoning Commission to meet a particular situation where topographical or other conditions make continuance or conformance to existing streets impracticable or where neighborhood design makes a varied plan appropriate.
- (d) Street connections. The system of streets designated for the subdivision, except in unusual cases, must connect with streets already dedicated in adjacent subdivision; and where no adjacent connections are platted, must in general be the reasonable projection of streets in the nearest

subdivided tracts, and must be continued to the boundaries of the tract subdivided, so that other subdivisions may connect therewith. Reserve strips of land controlling access to or egress from other property or to or from any street or alley or having the effect of restricting or damaging the adjoining property for subdivision purposes or that will not be taxable or accessible for special improvements shall not be permitted in any subdivision unless such reserve strips are conveyed to the Town in fee simple.

- **(e) Half streets**. Half streets shall be prohibited unless they are for the purpose of increasing the width of an inadequate existing right-of-way.
- (f) Street intersections. More than two (2) streets intersecting at a point shall be avoided, except where it is impractical to secure a proper street system otherwise and all intersections shall be as near ninety (90) degrees as possible and in no case shall the intersection angle be less than sixty degrees (60°).
- **Street jogs**. Non-intersecting streets with centerline offset of less than one hundred twenty-five feet (125') shall not be approved.
- (h) **Block lengths**. Block lengths shall be consistent with the Town of Norwood Master Plan and Major Streets Plan/Future Land Use Plan. Block lengths shall generally not be less than three hundred feet (300') or more than four hundred fifty feet (450').
- (i) Cul-de-sacs and hammerhead turnarounds. Streets with sul-de-sacs or hammerhead turnarounds shall not exceed four hundred (400') in length. Cul-de sacs shall have a turnaround diameter of one hundred feet (100'). Hammerheads shall extend a minimum 24' beyond each side of the improved roadway. Hammerheads are preferred over cul-de-sacs.
- **Dead-end streets**. Dead-end streets, except for cul-de-sacs, shall be prohibited unless they are designed to connect with future streets on adjacent lands and that have not been platted. In cases where these type of dead-end streets are allowed, a temporary turnaround of one hundred feet (100') shall be constructed.
- (k) Street design standards. Street and alley widths, curves, grades design speed and centerline radius shall meet following standards.

STREET DESIGN STANDARDS

Design Feature	Alley	Local	Local Collector		
Number of lanes	1	2	2		
Lane width	15'	10'	11'		
Pavement or chip/seal width (ft.)		28 ft.	28 ft.		

Right-of-way width	20	50 ft.	60 ft.		
Shoulder width (ft.)		2'	2'		
Crown (%)	2%	2% 2% 2%			
Borrow ditch		5ft.	5ft.		
Design speed (mph)		20-30	20-30		
Maximum grade	5%	7%	6%		
Min. centerline radius	50 ft.	150 ft.	250 ft.		

# (l) Designation of local and local collector streets.

- (1) Streets designated a local collector streets include San Miguel Street, Market Street, 44 Z (Vet Road), and 42Z (Aspen Street).
- (2) Streets designated as local streets include all other streets.
- (m) Street grade and curves. Streets may have a maximum grade of seven percent (7%). Centerline grade changes with an algebraic difference of more than two percent (2%) shall be connected with vertical curves of sufficient length to provide a minimum two-hundred (200') feet of sight distance. No vertical curve shall be less than two hundred feet (200') in length.
- (n) Street curve radii. Streets shall have a minimum radii at the centerline of one hundred fifty feet (150'), unless in special circumstances the Board of Trustees approves a local residential street with a smaller minimum radius in which case the developer shall pay any additional cost of installing the water and sewer lines caused by the smaller radius.
- (o) Alley right of way dedications required. Alleys right of way dedications shall be provided in all residential areas unless otherwise approved by the Planning and Zoning Commission and in commercial and industrial districts, except that the Planning and Zoning Commission may waive the requirement where other definite and assured provision is made for service access such as off-street loading, unloading, and parking consistent with and adequate for the uses proposed.
- (p) Minimum alley width. The minimum right-of-way width of an alley shall be twenty feet (20'). The minimum gravel surface width of alleys shall be fifteen feet (15').
- (q) Construction and dedication of internal streets. Streets shall be constructed by the developer and dedicated to the Town, along with all necessary right-of-way, with no pro-rata share from the Town.
- (r) Turn by-passes and turn lanes. Right-turn by-passes or left-turn lanes may be required at the intersection of collector streets if traffic conditions indicated they are needed. Sufficient right-of-way shall be dedicated to accommodate such lanes when they are required.

- (s) Street names. Street names shall be as established by the Norwood Major Streets and Future Land Use Plan, subject to approval of the Planning and Zoning Commission. When streets are in alignment with existing streets, any new streets shall be named according to the streets with which they correspond. Streets which do not fit into an established street-naming pattern shall be named in a manner which will not duplicate or be confused with existing streets within the Town or its environs. Street numbers shall be assigned by the Building Official in accordance with the Town numbering system.
- (t) Street posts and markers. The developer shall pay the cost of purchasing and installing street posts and markers at each street intersection, which posts and markers shall be the same type as used throughout the Town. The cost of such street posts and markers shall be paid to the Town Clerk upon final approval of construction plans for the subdivision. No subdivision construction, including but not limited to, street grading, street paving, storm sewer installation, curb and gutter work, sanitary sewer and water main installation, can begin until the cost of purchasing and installing such street posts and markers is paid to the Town Clerk.
- (u) **Traffic signs**. The developer shall pay the costs of purchasing and installing traffic signs at each street intersection, which traffic signs and posts shall be the same type as used throughout the Town.
- (v) Street Lighting. The developer shall pay the costs of purchasing and installing all street lighting equipment. The developer shall also pay the cost of all street lighting service for a period of two (2) years or until such time as seventy percent (70%) of the buildings for which building permits have been issued are completed, whichever is sooner. Prior to the issuance of any building permits or Certificates of Occupancies in the subdivision, the developer shall enter into a contract with the Town, in a form approved by the Town, setting forth the specific street lighting requirements for such subdivision. The type of equipment, method of installation and location of the wiring and light poles shall meet the minimum standards and requirements of the electric company from which electricity is to be purchased.
- (w) Street improvements. All final plats shall be subject to the following requirements regarding street improvements, provided that compliance with this subsection shall not be required for plat amendments approved and issued pursuant to Sec. 6.07(b), Plat amendments.
  - (1) No permit will be issued on property abutting any street in a subdivision prior to the approval of street grades and street improvements by the Town Engineer. Construction of the street improvements as required by the provisions of this chapter will not be necessary where the Town Engineer has determined that such street improvements are not possible or practical at the time the street improvements are required to be constructed. In the event the Town Engineer makes such a determination, a cost estimate for the construction for the required street improvements shall be prepared by the Town Engineer. The developer shall enter into an agreement with the Town, in a form approved by the Town, for the deposit of funds in accordance with the agreement and shall have no further liability for the construction of the required street improvements. The terms and conditions under which construction shall be accomplished and a disposition of the escrow account shall be provided for in the agreement. In lieu of depositing such funds in

- escrow, an Irrevocable Letter of Credit made out to the Town in an amount approved by the Town and in a form approved by the Town Attorney will be accepted.
- (2) No permit will be issued on property in a subdivision abutting a State Highway 145 until compliance with this paragraph has been met.
  - a. The Town Planner shall make a determination whether the State of Colorado will require participation in the cost of the improvements, and shall prepare a cost estimate for the required participation of the improvement of the state roadway.
  - b. In the event participation in the cost of improvements is required, the developer shall enter into an agreement with the Town, in a form approved by the Town, to deposit funds equal to the required cost of participation, including the cost for curbs, gutters, parallel storm sewer system, right-of-way, and utility adjustments.
  - c. The developer shall then deposit funds in accordance with the agreement and shall have no further liability for the construction of the required street improvements. The terms and conditions under which construction shall be accomplished and a disposition of the escrow account shall be provided for in the agreement.
- (3) Guarantee for construction or maintenance of streets. Approval of the plat shall not impose any duty upon the Town or County concerning the maintenance of improvements of any such dedicated parts until the proper authorities of the Town or County have made acceptance of the same by entry, use or improvement.

### § 4.04 Easements

- (a) Utility easements required. Utility easements shall be provided in all residential areas unless otherwise approved by the Planning and Zoning Commission and in commercial and industrial districts, except that the Planning and Zoning Commission may waive the requirement where other definite and assured provision is made for service access such as off-street loading, unloading, and parking consistent with and adequate for the uses proposed.
- **(b) Minimum width**. The minimum right-of-way width of a utility easement shall be ten feet (10') respectively.
- (c) Utility easements. Utility easements of ten (10)' feet in width on each side of all rear lot lines and five (5') feet in width on each side of side lot lines shall be provided where necessary. Where the rear or side lot lines abut property outside of the subdivision on which there are no rear or side lot line easements at least five (5') feet in width, the easements on the rear and side lot lines in the subdivision shall be ten (10') feet in width.
- (d) Potable water and sewer easements. Water and sewer easements shall be a minimum of twenty (20') feet in width.
- (e) "T" intersections and cul-de-sacs. Easements twenty (20') feet in width shall be provided in "T" intersections and cul-de-sacs for the continuation of utilities or drainage improvements, if

necessary.

- (f) Fire lanes and emergency access easements. Fire lanes and emergency access easements twenty (20') feet in width shall be provided where required by the Fire Marshal.
- (g) **Drainage easements**. When a proposed subdivision is traversed by an irrigation ditch or channel, natural creek or stream or a proposed drainage easement, there shall be provided a an easement sufficient for drainage and to allow for maintenance of the ditch.
- (h) Adjoining areas. When easements in areas adjoining proposed subdivisions are necessary to provide adequate drainage thereof or to serve such subdivisions with utilities, the subdivider shall obtain such easements.
- (i) Trail easements. Whenever a subdivision embraces any part of a bikeway, bridle path, cross country ski trail or hiking trail designated on the Major Streets and Future Land Use Plan, an easement shall be provided to accommodate the plan within the subdivision. A developer seeking final plat, rezoning or any other site specific development approval shall dedicated trail easements extending 25 feet either side of the proposed trail centerline for construction purposes and ten feet either side of the proposed trail centerline for public use purposes.

#### § 4.05 Park Land Dedication

- (a) **Purpose**. The requirements for open space, park, trails and recreational areas contained in this section are intended to ensure that in the Town of Norwood there will be sufficient land dedicated or otherwise set aside to meet the demand and need of the future residents of the development for open space, and parks, containing passive or active recreational areas that are reasonably attributable to such development.
- **(b) Applicability**. Every subdivision shall include a dedication of land to the Town, or other entity, as determined by the Board of Trustees, to be used for parks and recreation, trails, open space or cash-in-lieu of such dedication in an amount established by this section as a condition of final plat approval and prior to the recording of a final plat.
- (c) **Dedication requirement**. The obligation of the subdivider shall be to dedicate to the Town eight percent (8%) of the gross land area in fee simple, or other equivalent consideration, unless such a land dedication was required for the subject land at the time of annexation.

### (d) Specifications.

- (1) Land dedicated or otherwise set aside for open space, trails, and park and recreational areas shall be of such size, dimensions, topography, and general character as is reasonably required for the type of use necessary to meet the demand and need of future residents, e.g., open space buffer, public trails, active recreation for team or individual sports, playground, tot lot, picnic area, etc.
- (2) Unique natural areas or flood prone areas that provide an opportunity for linkage parks

- may be included in areas dedicated or otherwise set aside or reserved for open space.
- (3) No land dedicated or otherwise reserved in compliance with this section, except for areas dedicated for public trail purposes, shall have dimensions smaller than one hundred feet (100') in width and one hundred fifty feet (150') in depth, without the specific approval of the Town Board.
- (4) The Town, at its sole discretion, may elect to use the dedicated land for any municipal, school or other public function deemed necessary. Such use shall be compatible with surrounding use.
- (5) All lands to be dedicated must have access via a minimum 50-foot right-of-way and also must accommodate connection of all utilities necessary to operate the dedicated land as a public park or recreation area.
- (6) Signage designating this area as a trail or as park land shall be supplied by developer and/or owner. The selection and type of signage shall be approved by Board of Trustees.
- (7) Whenever a development proposal includes any part of a pedestrian, bicycle, equestrian or skiing trail designated on the Norwood Major Streets and Future Land Use Plan, a right-of-way shall be dedicated in compliance with the plan, and that portion of the land which is dedicated for public use purposes may be deducted from the otherwise required park land dedication.
- **(e) Platting requirements.** Any land dedicated for open space, trails, or park and recreational areas shall be shown on the face of a plat submitted for approval by the Planning and Zoning Commission and Board of Trustees.
  - (1) Pins to be installed. Each corner of the park land to be donated shall be marked with permanent monument consisting of three-fourths inch (3/4") iron pins set in concrete. These shall be located and identified on a recordable land survey completed by a land surveyor registered in the State of Colorado and provided to the Town by Owner and/or developer.
  - (2) Plat to be recorded. Upon approval by the Board of Trustees, said plat shall be filed of record in office of the San Miguel County Clerk and Recorder.
- (f) Payment of Cash in Lieu of Dedication. Payment of cash in-lieu of dedication of land for trails and park and recreational purposes shall be made prior to the recording of a final plat and shall be subject to the following provisions.
  - (1) Applicability. In any case in which the land required to be dedicated or otherwise reserved by this section would be less than one (1) acre, and in all cases in which the Board of Trustees may find that the park and recreational needs of a proposed development would be better served by a park in a different location or the expansion or improvement of an existing park or recreational area, the Town Board shall require a

- developer or subdivider to the Town of Norwood cash in lieu or to dedicate or convey other equivalent consideration in lieu of applicable cash dedication.
- (2) Schedule for Cash in Lieu. The amount of cash payment required shall be based on the number of acres that otherwise would be required to be dedicated. The Board of Trustees, following recommendation by the Planning and Zoning Commission shall, by resolution, set the per-acre fee for dedicated land based upon the current fair market value for raw lands adjacent to the Town of Norwood, which shall be annually updated. The current schedule for cash in lieu is \$25,000 per acre or \$0.574 per square foot.
- (3) Accounting, expenditure and refunds. All such payments of cash in-lieu fees shall be accounted for and spent according to the following requirements.
  - a. Cash in-lieu payments shall be segregated in a separate fund to be spent on a first in, first out basis and used only for the acquisition and improvements of open space, trails, and park and recreational areas within the Town of Norwood that will meet the needs of the residents of the development or subdivision in respect of which such payment was made;
  - b. Cash in-lieu payments shall be expended on the acquisition or improvement of open space, trails, or park land within reasonable proximity to the proposed development or subdivision from said development or subdivision; and,
  - c. If cash in-lieu payments are not expended or unconditionally committed to be expended within ten (10) years of receipt, the developer or subdivider shall be entitled to a refund of the amount paid upon written request by the developer or subdivider filed with the Town Clerk within one (1) year after the right to such refund arises.

### § 4.06 Drainage

- (a) Adequate drainage required. A developer shall provide, at his expense, drainage structures that will become integral parts of the existing street or roadway drainage system, and the dimensions of all drainage structures must be approved by the Town Public Works Director prior to installation. Drainage structures and ditches shall be of a size and nature sufficient to carry the calculated storm water from streets, roadways and open drainage areas as based on standard engineering principles.
- (b) Minimum standards. All provisions for drainage and flood control shall be established at a minimum to handle the anticipated 100 year frequency storms for maximum period of intensity over the entire drainage basin which the subdivision serves, and they shall be made in accordance with the approved improvement plan. The 100-year floodplain referred to herein shall mean that floodplain calculated on the basis of a fully-developed watershed, regardless of any regulated floodplain designations.
- **Erosion**. Where free fall of water occurs, satisfactory means shall be provided to prevent erosion of soil. Culverts shall have concrete head walls and wing walls where conditions require.

- (d) Catch basins. Standard drop inlet catch basins shall be constructed.
- **(e) Engineered design**. The Town may require that improvements be designed by a Colorado registered engineer.
- **Water and sewer system protection**. Water supply systems and sanitary sewage systems shall be designed to minimize or eliminate infiltration of floodwaters.

# § 4.07 Water Supply

All municipal water service shall be subject to the Rules and Regulations of the Norwood Water Commission. All potable water lines, fire hydrants and appurtenances shall be designed and constructed to meet the requirements of the Norwood Water Commission's Standards and Specifications, Exhibit "B" to the Rules and Regulations, and the Norwood-Redvale Fire District. Fire hydrants shall be provided to serve new subdivisions sufficient to maintain a Class 7 fire rating by the Insurance Service Office ("ISO").

### § 4.08 Sanitation

Sanitary sewer service and facilities shall be provided by and meet all requirements of the Norwood Sanitation District.

### § 4.09 Underground Utilities

All utilities shall be placed underground, except transformers, switching boxes, terminal boxes.

### § 4.10 Impact Fees (Reserved)

### § 4.11 Prominent Site Features

Prominent site features, such as landmarks, historic character and structures on the site, and significant topographic features shall be identified and taken into consideration in site planning and development. The Town may require such features to be preserved or otherwise protected.

# **Chapter 5.00 SITE DEVELOPMENT STANDARDS**

### § 5.01 Scope and Applicability

No building permit or certificate of occupancy shall be issued for any development until all of the applicable standards of this Sec. 5.00, Site development standards, are met.

## § 5.02 Parking, loading and access

- (a) **Purpose**. The purpose of this section is to require off-street parking and loading facilities in proportion to the parking demand for each use in order to ensure functionally adequate, aesthetically pleasing and secure off-street parking. The regulations and design standards of this section are intended to accomplish the following.
  - (1) To ensure the usefulness of parking facilities.
  - (2) To ensure sufficient parking spaces on-site in order to prevent excessive parking on public streets and in residential neighborhoods.
  - (3) To ensure that access to parking does not impair the function of adjacent roadways or endanger the public safety.

# (b) Applicability.

- (1) New and complying development. New development occurring after the effective date of this section, and development existing on the effective date of this section and complying with the number of off-street parking spaces required by this Chapter shall be subject to the following provisions.
  - a. Every use of a building or land hereafter established shall provide the minimum off-street parking spaces as required by this section.
  - b. The number of parking spaces may be reduced when the land use or floor area of a building is changed or reduced to a use or floor area for which fewer parking spaces are required.
  - c. When a building is expanded or a land use is changed so as to increase the number of parking spaces required, the number of such spaces shall be increased.
- (2) Existing noncomplying development. Developments with legally noncomplying parking areas shall be subject to the following provisions.
  - a. Existing parking spaces shall not be reduced below the minimum required by this section.

- b. Building permits and certificates of occupancy may be issued for a change of use or remodeling or structural alterations in developments containing legally noncomplying parking areas, without requiring compliance with this section, provided that such redevelopment does not result in an increase in the number of required parking spaces.
- c. Any building expansion or change of use that results in an increase over the number of parking spaces that would be required under this section for the lot shall provide additional parking spaces only for that increment of the expansion, as if it were a separate development. Only the expanded portion of the parking area shall be required to comply with the provisions of this section.
- (c) Computing parking. The minimum number of parking spaces required for a specific development proposal shall be based on the requirements listed in Sec. 5.02(d), Off-street parking requirements, and the following provisions.
  - (1) Unlisted uses. Where questions arise concerning the minimum off-street parking and requirements for any use not specifically listed, the requirements may be interpreted as those of a similar use.
  - (2) Multiple uses. In computing the parking requirements for any building or development, the total parking requirements shall be the sum of the specific parking space requirements for each class of use included in the building or development.
  - (3) Fractions. When measurements determining the number of required parking spaces result in fractions, any fraction less than one-half (½) shall be disregarded and any fraction of one-half (½) or more shall be rounded upward to the next highest full number.
- (d) Off-Street parking requirements. Off-street parking spaces shall be provided in accordance with the following minimum requirements.

### SCHEDULE OF OFF-STREET PARKING REQUIREMENTS

District	Churches and Auditoriums	and	Residential Uses Single Family	Residential Uses Multi Family	Non-Residential Uses
A, Agricultural	1 space per 4 seats	None	2 spaces per dwelling unit		3 spaces per 1000 sq. ft. of net leasable space
R-1, Residential	1 space per 4 seats	None	_	1 space per dwelling unit up to 750 square feet 2 spaces per dwelling unit 751 square feet or larger	3 spaces per 1000 sq. ft. of net leasable space

District	Churches and Auditoriums	Hotel and Motel Units	Residential Uses Single Family	Residential Uses Multi Family	Non-Residential Uses
R-1, Residential Cottage Housing Developmen t			1 space per unit for units >750 square feet 2 space per unit for units 750 square feet or larger		
MH, Mobile Home	1 space per 4 seats	None	2 spaces per dwelling unit		None
B-1, Business	1 space per 4 seats	1 space per rental unit or manager's unit	2 spaces per Dwelling Unit	1 space per dwelling unit up to 750 sf 2 spaces per dwelling unit 751 to 2000 sf 3 spaces per dwelling unit 2001+sf	1 spaces per 450 sq. ft. of net leasable space 1 space per 150 sf of space accessible by public and 1 space per 450 sf of space not accessible to public for bars and restaurants.  1 space per 1,500 sf of floor area for buildings in the Commercial Core regardless of type of non-residential use
I, Industrial	1 space per 4 seats	None	2 spaces per dwelling unit	1 space per dwelling unit up to 750 sf 2 spaces per dwelling unit 751-2000 sf 3 spaces per dwelling unit 2001+sf	3 spaces per 1000 sq. ft. of net leasable space 1 space per 150 sf of space accessible by public and 1 space per 450 sf of space not accessible to public for bars and restaurants
P, Public	1 space per 4 seats	None	2 spaces per unit	1 space per dwelling unit up to 750 sf	None

District	and Auditoriums	and	Residential Uses Single Family	Residential Uses Multi Family	Non-Residential Uses
				2 spaces per dwelling unit 751-2000 sf	
				3 spaces per dwelling unit 2001+sf	
Planned Unit seats the		the underlyin g zone	Those of the underlying zone district	Those of the underlying zone district	Those of the underlying zone district

- (e) Off-site parking. Required off-street parking spaces shall be on the same lot as the principal use, provided that all or a portion of the required parking spaces may be located on a separate lot from the lot on which the principal use is located, based on the following provisions.
  - (1) Contiguous. The off-site parking shall be located on a lot that is immediately adjacent to and contiguous with the lot containing the principal use.
  - (2) Common ownership. Any off-site parking area shall be under the same ownership as the principal use to which it is accessory, and all necessary legal instruments shall be executed to ensure that the required number of spaces shall remain available throughout the life of the principal use.
- (f) Parking space dimensions and design. Each off-street parking space shall consist of an open area measuring eight feet (8') wide by eighteen feet (18') long and seven feet (7') high and shall have vehicular access to a public street or alley. Off-street parking shall be properly drained and must be paved with all-weather surfacing or be covered with gravel and maintained in a usable condition at all times.
- (g) Restricted use of parking areas. No junked vehicle shall be parked or stored in public view on any street or private property in any residential district. Junk vehicles shall be defined as those that either (1) lack a current license or inspection sticker or (2) are wrecked and/or dismantled.
- (h) **Design of loading areas**. All loading spaces in the I, Industrial District shall comply with the following design requirements.
  - (1) Location. No loading space shall be located within the right-of-way of a public street. Any loading dock or door shall be set back far enough from the right-of-way so that no

portion of the right-of-way is occupied by trucks or other vehicles while loading or unloading. The location of the loading area shall not interfere with the free circulation of vehicles in the off-street parking area. No loading space shall be located so as to block access by emergency vehicles.

- (2) Size of berths. The minimum required dimensions of loading spaces, open or enclosed, shall be twelve feet (12') in width by fifty-five feet (55') in length, with a minimum vertical clearance of fifteen feet (15'). Where tractor-trailer units will be using the facility, the minimum length shall be sixty-five feet (65').
- (3) Paving standards. The surface of all open off-street loading spaces shall conform to the requirements for off-street parking areas.
- (4) Use of loading areas. Required off-street loading spaces and associated aisles and maneuvering areas shall be used for vehicle loading only. No sales, storage, display of merchandise (including automobiles), repair work or dismantling shall be permitted in such areas.

# (i) Driveways and access.

- (1) Access required. All required off-street parking shall be provided with driveway access to a public street or alley.
- (2) Minimum driveway width. Driveways shall be a minimum of ten feet (10') in width when serving one (1) dwelling unit, or fourteen (14) feet wide when serving more than one residence or another use such as a boarding house.
- (3) Corner visibility. No walls, fences or other obstruction to view shall be placed on any corner lot that would obstruct vision to automobile drivers.

### § 5.03 Sidewalks and Trails

(a) Applicability. This section is intended to ensure pedestrian access is available to serve uses that need and benefit from such access. Builders shall be required to construct sidewalks and trails in accordance with the Major Streets and Future Land Use Plan, on all streets adjacent to their building sites in all zoning districts. No Certificate of Occupancy shall be issued until the requirements of this section are met.

If a builder or property owner has reason to believe a proposed use does not need or benefit from pedestrian access, a written request for interpretation per Section 2.03 of this Chapter may be submitted to the Town Planner describing the use and explaining why pedestrian access should not be required as a condition of the building permit for the proposed use.

**(b) Sidewalk construction standards**. Sidewalks shall be a minimum of five feet (5') in width in the B-1, Business District. Sidewalks shall be a minimum of four feet (4') in width in the all other districts. Only in the case of sidewalks being repaired may the permitted width match the existing sidewalk; however, a minimum width of four (4) feet is encouraged. The construction specification

of sidewalks will conform to Town specifications for streets on file with the Town Engineer and Town Clerk.

(c) Trails construction standards. Trails in the Town of Norwood may be designed specifically as for horseback riding, hiking, mountain biking, nordic skiing or any combination thereof. As a result, the Board of Trustees may require any specific standards (including less restrictive standards) for a particular proposed trail, or section thereof, based on the design standards set forth in the following figure and in the "Trail" definitions in subsection 2.02. However, in most cases and unless otherwise specified by the Board of Trustees, all trails shall be designed based as multi-use urban trails.

TRAIL DESIGN STANDARDS											
		X-Slope Range	Tread Width	Clearing		Surface Materials			Cross Slope	Max Profile	
				h.	v.	natural	gravel	asphalt	concrete		
Access	Disabled access	0-10%	7'	+4'	10'	No	No	Pref.	Pref.	<2%	5% avg, 8.33% max
Hiking	Walking	0-10%	10' min	+4'	10'	No	No	Pref.	Pref.	4%	5% avg
	Standard	10-70%	24-36"	+4'	10'	Pref.	Pref.	No	No	4%	12%
Mountain Bike	Rural double Track	10-70%	48-96"	+6'	10'	Pref.	Pref.	No	No	4%	12%
Equestrian	Rural double track	10-70%	48-96"	+6'	10'	Pref.	Pref.	No	No	4%	12%
Nordic	Single track	0-70%	12"	5'	+6'	Pref.	Pref.	OK	OK	NA	10% desired
	Double track	0-70%	12"/12"	10'	+6'	Pref.	Pref.	Ok	OK	NA	10% desired
	Skate track	0-70%	8'	10'	+6'	Pref.	Pref.	OK	OK	NA	10% desired
Multi-Use	Urban	0-10%	10'	+4'	10'	Pref.	Pref.	No	No.	2%	5% avg

### **§ 5.04** Fences

- (a) Solid Fence is described as open less than 50% of the fabric of the fence.
- **Fence Face** is the finished side having no structural supports. Where applicable, the Fence Face shall face abutting property or rights of way.
- **Fence Height**. In any district where a fence or screening separation is erected, the following standards for height and design shall be observed.
  - (1) Fences shall be erected in accordance with the requirements of Sec. 5.02(i) (3), Corner visibility.
  - (2) Front yards. A Solid Fence extending across the front or along the sides of front yards must not exceed 48 inches in height. An extension that is "transparent" in character is allowed; however, the total height of the fence shall not exceed 72 inches.
  - (3) Rear yards. No fence extending across the rear or along the sides of rear yards may exceed 72 inches in height.
  - (4) Pergolas, arbors or similar open structures at pedestrian entry ways may exceed the height limitation.
  - (5) Height shall be the average distance between the top element in the fence and the adjacent grade over a straight section of fence with no corners exclusive of pergolas, arbors or similar open structures.
  - (6) Fences shall be constructed of durable materials designed to withstand weathering with a life expectancy of ten years. Fences shall be affixed to the ground for stability using normal and customary construction methods; i.e., footers, posts, and evenly spaced horizontal and vertical bracing. Appropriate materials are wood, metal, iron, twisted/decorative wire and faux wood. Slatted "snow" style and mesh "construction" style materials are inappropriate. Harmonizing designs with consistent repeated patterns are encouraged.
  - (7) Fences in easements shall not impede the flow of water or the underlying uses authorized for the easement.
  - (8) Fences including footings, shall be located entirely upon the private property on which the fence is constructed.
  - (9) Fences required for screening or protective measures by other provisions of this Code shall be considered pursuant to the conditions of the section of the Code that requires the fence.
  - (10) Fences must be maintained so as not to endanger life or property. Any fence which,

through lack of repair, type of construction, or otherwise, imperils life or property, shall be deemed a nuisance and subject to the abatement procedures of the Norwood Municipal Code.

- **(d) Boundaries**. Any fence placement that may violate a property line becomes a civil issue and the Town shall not arbitrate any dispute.
- (e) Any **landscaping** along any property line may augment privacy with no regulation.
- (f) Fences containing barbed wire, concertina or similar wire and electrification are prohibited.

## § 5.05 Landscaping and Screening

- (a) Purpose. This section is designed to provide standards for the installation and maintenance of landscaping, walls and screening devices so as to promote the general welfare of the community. This is accomplished by encouraging the creation of an attractive appearance along streets and by screening from view those uses that may be unattractive to the public eye. Landscaping materials, including ground covers, shrubs, and trees further facilitate the control of erosion and the reduction of glare and dust, as well as the visual softening of building masses. Walls and screening devices allow for the separation of incongruous uses and for the buffering of intensive activities. Landscaping, walls and screening devices together, help to effectuate privacy and logical development.
- **(b) Applicability**. This section shall apply to all development in B-1, Business District, Planned Unit Development Overlay District and I, Industrial District.

## (c) Landscaping Standards.

- (1) Landscape plan. Landscape plan. Any proposed development application shall include a landscape plan when landscaping is required. It shall indicate the location of existing and proposed improvements, parking areas, street improvements, locations and types of landscaped areas, walls, and descriptions of the vegetation to be utilized and the method of irrigation.
- (2) Street yard landscaping. All undeveloped areas of the street yard(s), not used for parking, maneuvering areas, ingress or egress, fire lanes, pedestrian areas, shall be landscaped, including development in the Commercial Core with street yard(s).
- (3) Native plant materials. Native, trees, grass, shrubs or other plants appropriate for the climate and low water use plant materials shall be utilized in order to minimize the consumption of water.
- (4) Irrigation. All required landscaped areas shall be required to include an irrigation system to insure the health and growth of the landscape. When available, irrigation systems shall utilize untreated, irrigation water instead of treated water. Drip irrigation or other water efficient systems are encouraged.
- (5) Location of utilities. Proposed utilities shall be located, when possible, so that their installation will not adversely affect vegetation to be retained on a site.

- (6) Installation. Landscaping, watering devices, walls and screening structures shall be installed in accordance with the approved landscape or screening plan prior to issuance of a final Certificate of Occupancy for the building or use. The Building Official may grant a temporary Certificate of Occupancy during the winter months when installation is impracticable or not feasible.
- (7) Removal of topsoil prohibited. No person, firm or corporation shall strip, excavate or otherwise remove top soil except in connection with the construction or alteration of a building on such premises and excavation or grading incidental thereto.
  - (8) Maintenance requirements.
    - a. Landscaped areas shall be reasonably maintained by the owner or the lessee of the property, including pruning, trimming, watering, weed control or treatment, and other requirements necessary to create an attractive appearance for the development.
    - b. Any plant materials not surviving shall be replaced within thirty (30) days of its demise or in the next appropriate season.
    - c. Lack of maintenance of required landscaping material shall constitute a violation of this chapter.

# (d) Screening.

- (1) Any proposed development application shall include a screening plan when screening is required. It shall indicate the location of existing and proposed improvements required to be screened and the screening method to be utilized.
- (2) Screening refers to a wall, fence, hedge, informal planting, or berm, provided for the purpose of buffering a building or activity from neighboring residential areas or from the street. When required, screening may be provided by one or more of the following means:
  - a. A solid masonry wall, fence, or equivalent meeting the standards of sec.5.04.
  - b. A plant screen or evergreen hedge.
  - c. An earth berm may be used in combination with any of the above types of screening. The faces of a berm's slope shall be planted with ground cover, shrubs, and trees. Berms are encouraged for screening between commercial uses and residential uses.
  - d. When vegetation is used with the screening method, it shall conform with the landscaping standards 5.05(c)(3) and (4).

## (3) B-1, Business District screening standards.

a. Height of screening devices. The height of screening devices shall be measured from the highest finished adjacent grade of the element to be screened. Unless otherwise specified, screening required by this Section shall be arranged within the boundaries of

the lot as to substantially hide from adjoining properties the building, facility or activity required to be screened.

- b. Heights of plant screens or hedges specified herein indicate the height which may be expected within three (3) years of planting. The height at the time of planting shall be such that in accordance with good landscape practice the fully required height may be achieved within a three (3) year period.
- c. No screening shall be placed in such a manner that it obstructs the vision of right-of-way users to safely navigate sidewalks, streets or alleys.
- d. **Parking areas**. Parking areas shall be screened from street view to a minimum height of three feet (3') above the highest finished grade of the parking area. The required screening shall provide breaks as necessary for vehicular and pedestrian access Such screening may be accomplished by the use of plants, earth berms, or natural topography.
- e. **Outdoor storage areas**. All outdoor storage areas shall be screened from street view.
- f. Exterior refuse and recycling collection areas. Refuse collection areas and recycling collection areas must be orientated away from and screened from street views. When the collection area is fully enclosed, the enclosure shall be appropriately sized to screen refuse receptacles and shall be constructed of solid, durable and attractive walls/fences, with solid doors and shall be visually consistent with project architecture. Enclosures shall be compliant with applicable fire codes. Refuse and recycling receptacles for pedestrian use are exempt.

# g. Mechanical equipment.

- i. Mechanical equipment located on the ground, must be screened. Screening shall be tall enough to conceal the equipment.
- ii. Mechanical equipment placed on roofs must be screened by a parapet around the equipment that is as tall as the tallest part of the equipment and shall not exceed the maximum height allowed in the applicable zone district. Screening shall be compliant with applicable building and fire codes and height requirements.

## (4) I, Industrial District screening standards.

- a. Height of screening devices. The height of screening devices shall be measured from the highest finished adjacent grade of the element to be screened. Screening shall be high enough to shield the view of storage areas, parking and outdoor lighting from uses off-site.
- b. Parking areas. Parking areas shall be screened from street view to a minimum height of three feet (3') above the highest finished grade of the parking area. Such screening may be accomplished by the use of plants, earth berms, or natural

topography.

- c. Outdoor storage areas. All outdoor storage areas for materials, refuse, mechanical equipment (to include ground based satellite dishes), vehicles, or other similar items shall be screened from street view.
- d. Roof mounted equipment. Roof mounted mechanical equipment to include satellite dishes and antennas shall be screened by parapet walls or other screening devices to be no lower in height than six feet (6') below the height of the mechanical equipment on side, front, or rear walls, whichever are adjacent to public streets or residential districts.
- e. Outdoor lighting. All outdoor lighting shall comply with the Section 5.08, "Exterior Lighting Standards".
- f. Landscaping. When vegetation is used in conjunction with screening, it shall comply with Section 5.05(c)(3) and (4), above.
- g. Location Adjacent to Residential Areas. Where industrial uses abut residential uses, screening shall block views of storage areas and high activity and traffic portions of the industrial use to the extent possible. Screening shall also be used to shield adjacent uses from noise and light emanating from the industrial uses. Berms and landscaping shall be included in the screening materials to create a soft transition to adjacent residential areas.
- h. Berms. Berms are encouraged to provide adequate screening between industrial and other uses to reduce noise and light impacts on adjacent property.

## § 5.06 Signs

- (a) Purpose and intent. The purpose of regulating signs is to promote safety of persons and property so as: 1) not to confuse or distract motorists or impair drivers' ability to see pedestrians, obstacles or other vehicles; 2) to promote efficient communication of messages; 3) to promote the public welfare by reducing visual clutter; 4) to assist in wayfinding; and 5) to provide fair and consistent enforcement.
- **(b) Applicability**. The following regulations shall govern the placement and construction of all outdoor advertising display within the Town of Norwood.
- (c) Temporary yard signs, pennants, banners, and posters.
  - (1) Maximum size: 6 sq. ft. except banners and/or pennants for civic events.
  - (2) Maximum Number: one for each street frontage upon which the property faces, and so placed that only one sign faces each street.
  - (3) Illumination: none.

- (4) Time limit: two months maximum. Once the purpose for which the sign has been erected has been fulfilled, the permit shall be deemed void.
- (5) Pennants and banners that are to hang over Colorado State Highway 145 must also be approved by the Colorado Department of Transportation.

### (d) Site Signs.

- (1) Number: one for each street frontage upon which the property faces, and so placed that only one sign faces each street.
- (2) Size:
  - a. Residential use and home occupations 2 sq. ft. per sign.
  - b. Freestanding and projecting signs: 20 sq. ft. per sign.
  - c. Wall signs surface of a building: 32 sq. ft. per sign.
  - d. Signs without specific borders, frames or edges which may be located on awnings, porches, windows or other surfaces shall include the area within the perimeter surrounding lettering, colors, designs and symbols distinctive from the surface on which it is located.
  - e. Maximum Aggregate Size: In multiple tenant buildings, multiple signs up to an aggregate of 32 sq. ft. may be permitted on lots with up to 75 linear feet frontage. On lots with more than 75 linear feet frontage, up to 6 sq. ft. for each additional 25 linear feet up to a maximum of 24 additional sq. ft. may be allowed.
- **Structural Characteristics**. The structural characteristics of signs shall comply with the standards of this section.
  - (1) Free-standing signs shall not exceed twenty feet(20') in height. Each free-standing sign may have two faces, each with the maximum area 20 sq. ft. per sign, provided the two faces are the same size and join back-to-back without any overlap.
  - (2) Projecting Signs shall not be higher than the eave line or parapet wall of the principal building and shall be a minimum of 8 feet above grade when located adjacent to or projecting over a pedestrian way; shall not extend more than 4 feet from the building wall except where such a sign is an integral part of an approved canopy or awning. Each projecting sign may have two faces, each with the maximum area 20 sq. ft. per sign, provided the two faces are the same size and join back-to-back without any overlap.
  - (3) Wall Signs shall not be higher than the eave line or parapet wall of the principal building and no sign part, including cutout letters, shall project more than 12 inches from the building wall.

- (4) Signs shall not advertise business, activity, product or service not conducted on the premises upon which sign is located, have a moving part or are portable or wheeled; or contain or consist of ribbon streamers, strings of light bulbs, spinners, or other similarly moving devices.
- (5) There shall be no more than one sign for each street frontage upon which the property faces, and so placed that only one sign faces each street.

## (f) Illumination.

- (1) The light from any illuminated sign shall be so shaded, shielded or directed that the light intensity, brightness and other qualities of the light will meet the requirements of Sec. 5.08.
- (2) No sign shall have blinking, flashing or fluttering lights or other illuminating devices which have a changing light intensity, brightness or color. Beacon lights are not permitted. This does not apply to normal and customary Christmas lighting.
- (3) No colored lights shall be used at any location or in any manner so as to be confused with or construed as traffic control devices.
- (4) Neither the direct nor reflected light from primary light sources shall create a traffic hazard to passing motorists.
- (g) Signs shall be maintained in **good repair**.
- (h) **Permits**. It shall be unlawful to erect, construct, reconstruct, alter, paint, or repaint, or change the use of any sign without first obtaining a sign permit; however, a sign permit shall not be required to repaint a sign exactly as it was permitted for the purpose of maintenance. For all sign permits, a fee shall be charged to cover the costs of administration.

The following are exempt from the requirements to obtain a permit:

- a. signs erected by proper authority of the Town shall be exempt from these regulations;
- b. signs carried by people;
- c. flags or crests except when displayed in connection with commercial promotion;
- d. signs which advertise a private garage or yard sale on the lot on which the sign is located; provided such signs are displayed no more than twice per year per dwelling unit for a period not to exceed three (3) days;
- e. Merchandise, pictures or models of products or services which are incorporated as an integral part of a window display.

### § 5.07 Performance Standards

- (a) Applicability. All uses in any district of the Town of Norwood shall conform in operation, location and construction to the subjective performance standards herein specified so that the public health, safety and welfare will be protected.
- **(b) General**. The location, size, design and operating characteristics of all uses shall minimize adverse effects, including visual impacts, on surrounding properties.

### (c) Unreasonable Noise.

- (1) It is unlawful to any person to make, continue, or cause or permit to be made or continued any unreasonable noise upon any premises within the Town of Norwood under such person's control or operation.
- (2) For purposes of this section "unreasonable noise" means any sound of such level or duration as to be or tend to be injurious to human health or welfare, or which would interfere with the enjoyment of life or property throughout the Town or any part thereof.
- (3) The following noises shall be prima facie a violation of this section:
  - a. Use of loudspeakers, amplifiers or other devices for amplification of sound which is plainly audible a distance of fifty feet from the building, structure or vehicle from which it is emanating between the hours of:
    - (i) For Sunday through Thursday: 9:00 p.m. and 7:00 a.m. the following morning;
    - (ii) For Friday and Saturday: 11:00 p.m. and 7:00 a.m. the following morning;
  - b. Noise resulting from construction related activities during the time of 7:00 p.m. and 7:00 a.m. the following morning.
- (4) The following are factors which may be considered in determining whether a noise is in violation of this section:
  - (i) The volume of the noise.
  - (ii) The intensity of the noise.
  - (iii) The duration of the noise.
  - (iv) Time of day or night of the noise.
  - (v) The volume and intensity of the background noise, if any.
  - (vi) The proximity of the noise to residential development.
  - (vii) The zoning, density, and types of development of the areas from which the noise emanates and to which it travels.
  - (viii) Whether the noise is recurrent, intermittent or constant.

- (ix) And all other factors tending to show the magnitude and/or disruptive effect of the noise.
- (5) This section shall not apply to traditional agricultural operations being conducted within the A, Agricultural Zone District.
- (6) This section shall not apply to construction projects conducted between the hours of 7:00 a.m. and 7:00 p.m., Monday through Saturday, and in compliance with any time limitations imposed by any construction permit issued by the Town of Norwood, or, if no time limitation is imposed, for a reasonable period of time for completion of the project.
- (7) Noise caused in the performance of rescue or emergency work for the immediate safety, health or welfare of the community or any individual shall be exempt from the provision of this chapter.
- (8) Noise caused in the maintenance, construction, repair or improvement of any public road or public facility by public employees, by any person acting pursuant to a public works contract, or any person acting under the direction or control of any employee of any public agency, shall be exempt from the provision so this section.
- (9) This section is not applicable to the use of property owned or used by the State of Colorado, the Town of Norwood, San Miguel County, or any political subdivision of the State of Colorado, or any of their lessees, licensee, or permittees for the purpose of promoting, producing, or holding cultural, entertainment, athletic or patriotic events, including, but not limited to concerts, music festivals and fireworks displays.
- (10) Any person may petition the Board of Trustees, thirty (30) days in advance, for a temporary hardship permit setting for the basis of the undue hardship in writing, the anticipated duration of the condition creating hardship and any other relevant matters. A temporary hardship permit may be granted if it is found that the activity, operation or noise source will be of a temporary duration, and that compliance with the requirements of this section cannot reasonably be obtained. No temporary permit may be issued for a period exceeding forty five (45) days and the Board of Trustees may set any conditions, limitations or requirements necessary to minimize adverse effect on the area impacted by the noise. In making a determination as to whether or not to issue a permit or the conditions, limitation or requirements to be imposed thereon, the Board may consider, among other things, the following circumstances:
  - a. Whether additional time is necessary for the permittee to alter or modify his activity or operation to comply with this Section.
  - b. Whether the activity, operation or noise source will be of a temporary duration, and cannot be done in a manner that would comply with this section.
  - c. Whether any other reasonable alternative is available to the permittee.

- d. Any interim noise control measures to be taken to minimize the noise.
- e. And the schedule of noise control measures that will be taken to bring the source into compliance within a reasonable time.
- (11) Any violation of this section shall be deemed a violation of this Land Use Code and shall be subject to all remedies for a violation hereof. Each day of a violation shall constitute a separate offense.
- (d) Smoke and Particulate Matter. No operation or use in any district shall at any time create smoke and particulate matter that, when considered at the bounding property line of the source of operation creates a nuisance or distracts from the use and enjoyment of adjacent property.
- (e) Odorous matter. No use shall be located or operated in any district that involves the emission of odorous matter from a source of operation where the odorous matter exceeds the odor threshold at the bounding property line or any point beyond the tract on which such use or operation is located. The odor threshold shall be the concentration of odorous matter in the atmosphere necessary to be perceptible to the olfactory nerve of a normal person.

# (f) Fire and explosive hazard material.

- (1) Explosives. No use involving the manufacture or storage of compounds or products that decompose by detonation shall be permitted in any district, except that chlorates, nitrates, phosphorus and similar substances and compounds in small quantities for use by industry, school laboratories, druggists or wholesalers may be permitted when approved by the Fire Marshall as not presenting a fire or explosion hazard.
- (2) Flammables. The storage and use of all flammable liquids and materials such as pyroxylin plastics, nitrocellulose film, solvents and petroleum products shall be permitted only when such storage or use conforms to the standards and regulations of the Town of Norwood.
- (g) Toxic and noxious matter. No operation or use in any district shall emit a concentration across the bounding property line of the tract on which such operation or use is located of toxic or noxious matter that will exceed the threshold limits set forth by the Colorado Department of Health.
- (h) **Vibration.** No operation or use in any district shall at any time create earthborne vibration that, when considered at the bounding property line of the source of operation creates a nuisance or distracts from the use and enjoyment of adjacent property.
- (i) Open storage. No open storage of materials or commodities shall be permitted in any district except as an accessory use to a main use located in a building in an "I" or "PUD" Industrial District. No open storage operation shall be located in front of a main building. No wrecking, junk, or salvage yard shall be permitted as a storage use in any district.

## § 5.08 Exterior Lighting Standards

- (a) **Purpose and Intent.** The purpose of the regulation is to regulate outdoor night-time lighting fixtures to preserve, protect and enhance the Town's dark sky while conserving energy, permitting reasonable and safe use of outdoor night-time lighting, minimizing glare and obtrusive light, and helping to protect the natural environment from the damaging effects of night lighting.
- **(b) Applicability.** All outdoor lighting Luminaires (fixtures) shall be installed in conformance with this section to maintain the Town's designation as an International Dark-Sky Association (IDA) Dark Sky Community. These regulations are based on the requirements of the International Dark-Sky Association Dark Sky Community guidelines as described in its Model Lighting Ordinance for Lighting Zone 1<sup>1</sup>.

### (c) Residential.

(1) Luminaires and lighting installations shall be Fully Shielded and/or IDA approved luminaires with no uplight. IDA approved luminaires are readily available. Look for the IDA symbol:



- (2) Individual Luminaires shall not exceed 850 Lumens (equivalent to 60w incandescent).
- (3) Maximum allowed Lumens per Site: 3,400 with limitation of no more than 2,400 Lumens for Luminaires that are not on automatic light control devices.
- (4) Maximum allowed Correlated Color Temperature: 3,000 Kelvin (K).
- (5) Directional flood lighting is prohibited.
- (6) Lighting installations on structures shall not exceed the height of the eave or fourteen (14) feet whichever is less.
- (7) Residential mounting pole height restrictions can be considered but mounting heights shall not exceed fourteen (14) feet.
- (8) Residents are encouraged to use timers to control Luminaires.
- (9) Residents are encouraged to use motion sensors for Luminaires after midnight.

<sup>&</sup>lt;sup>1</sup> Table D, Performance Method Allowed Total Site Lumens, Table E, Performance Method Additional Initial Luminaire Lumen Allowances and Table G, Residential Lighting Limits.

#### (d) Non-residential.

- (1) A photometric analysis must be submitted with exterior lighting plans unless the Town Planner deems it unnecessary.
- (2) All Luminaires and lighting installations shall be Fully Shielded or have a BUG rating not to exceed:
  - a. B1 rating for backlighting.
  - b. U0 rating for uplight.
  - c. G1 rating for glare.
- (3) Maximum allowed Lumens per square foot: 1.25.
- (4) Maximum allowed Lumens per Site: 10,000.
- (5) Maximum allowed Correlated Color Temperature: 3,000 K.
- (6) Directional flood lighting is prohibited.
- (7) Luminaires must be extinguished at close of business unless required for safety of employees during employee working hours. If employee lighting is required, it must be approved in conjunction with the Exterior Lighting Plan. After close of business, automatic lighting control devices, such as motion sensors, must be used for canopies, pedestrian pathways and entrances/exits where required for public safety.
- (8) Light for illuminating signs may not be used after midnight. Vertical illuminance from Luminaires shall not exceed ten (10) foot-candles on surface of sign.<sup>2</sup>
- (9) Building entrances or exits shall not exceed 1,000 lumens within 20 feet of door.
- (10) Canopies. Luminaires must be located on underside of canopy and shall not exceed 1.5 to 3 foot-candles per sq. ft. of unit area within the drip line of the canopy.
- (11) Outdoor Dining, per unit area for the total hardscape of outdoor dining maximum of 3 Lumens/sq. ft.
- (e) Parking Lots (see definition below) for General Commercial, Public and mixed uses.
  - (1) In addition to the requirements for Non-residential exterior lighting, parking lot light installations must also meet these standards:

[106]

<sup>&</sup>lt;sup>2</sup> Additional restrictions in Land Use Code sec. 5.06 may apply to exterior lighting.

Mounting Height	14 feet	
Average light levels at ground	d Maximum 1.5 foot-candles	
Recommended Average light level at ground	No more than 0.3 foot-candles	
Maximum light level at ground	5 foot-candles	
Uniformity ratio maximum: minimum	20:1	
Uniformity ratio average: minimum	8:1	

- (2) After close of business, event or midnight whichever is later, automatic light control devices that dim light levels by a minimum of fifty percent (50%) are required but extinguishing the luminaires is encouraged.
- (f) High Intensity or Special Purpose, including, but not limited to, Conditional Uses, vehicle service station fueling areas, ATMs, night depositories, and stadiums shall be reviewed on a case by case basis for approval by the Planning and Zoning Commission. To obtain such a permit, applicant shall demonstrate that it has made every effort to mitigate the effects of light on the environment and surrounding properties. A statement of the mitigation measures must be submitted with the Exterior Lighting Plan and shall include calculations showing variances from these standards. The Planning and Zoning Commission may approve the lighting installation if it finds that it will not create unwarranted glare, sky glow or light trespass and will not cause an undue threat to the community International Dark Sky designation.
- **(g)** The following are exceptions from the requirements of this section:
  - (1) Outdoor lighting fixtures existing or legally installed prior to [insert date of adoption of ordinance]; compliance with the provisions of Chapter 1, Section 1.06 is required and when compliance is required under said section lighting installations are subject to all the provisions of this section.
  - (2) Holiday lights are allowed to be displayed for up to forty-five (45) days in any one year. Use of automatic light control devices are encouraged to extinguish light after midnight.
  - (3) Temporary lighting that cannot meet these standards shall not be used more than thirty (30) days in any one year and shall require approval as a temporary Conditional Use.
  - (4) Emergency lighting used by police, firefighters or medical personnel and that is in operation as long as the emergency exists.
  - (5) Existing outdoor recreational facilities, whether public or private, shall not be illuminated after 11:00 p.m. except to conclude any recreational or sporting event or other activity conducted, which is in progress prior to 11:00 p.m. at a ballpark, arena or similar facility.

## (h) Additional Requirements:

- (1) Architectural Lighting is prohibited.
- (2) Landscape Lighting shall be included in the total allowed lumens per site.
- (3) Light Trespass is prohibited.
- (4) Use of mercury vapor lighting fixtures is prohibited.

#### (i) Definitions:

**Architectural Lighting.** Lighting designed to reveal architectural shape and/or form and for which lighting for any other purpose is incidental.

**BUG.** A luminaire classification system that classifies backlight (B), uplight (U) and glare (G).

**Correlated Color Temperature.** A unit of measure for light color based on the Kelvin temperature scale (K). Color temperatures over 4000 K are called cool colors (bluish white), while lower color temperatures (2700K–3000 K) are called warm colors (yellowish white through red).

**Foot-candle.** The unit of measure expressing the quantity of light received on a surface. One foot-candle is the illuminance produced by a candle on a surface one foot square from a distance of one foot.

**Fully Shielded.** A Luminaire that is constructed and installed in such a manner that all light emitted by the Luminaire, either directly from the lamp or a diffusing element, or indirectly by reflection or refraction from any part of the Luminaire, is projected below the horizontal plane through the Luminaire's lowest light-emitting part.

**Glare.** Lighting entering the eye directly from Luminaires or indirectly from reflective surfaces that causes visual discomfort or reduced visibility.

**Landscape lighting.** Luminaries mounted in or at grade (but not more than 3 feet above grade) and used solely for landscape rather than any area lighting.

**Light Trespass.** Light that falls beyond the property it is intended to illuminate.

**Lumen**. A unit used to measure the amount of light that is produced by a lamp at the lamp, not by the energy input, which is indicated by the wattage.

**Luminaire** (**light fixture**). The complete lighting unit consisting of one or more electric lamps, the lamp holder, any reflector or lens, ballast (if any), together with the components designed to distribute light, to position and protect lamps, and to connect the lamps to power supply and any other accessories.

**Parking Lot.** A parking area with more than twenty (20) spaces for vehicle parking.

#### Site:

- (1) *Residential*: an undivided or combination of undivided lots under one ownership occupied by a residential structure or a residential duplex structure regardless of the square footage.
- (2) *Non-residential*: an undivided or combination of undivided lots under one ownership occupied by a non-residential structure regardless of the square footage.

**Photometric Analysis:** details of a proposed exterior lighting plan used to verify compliance with these standards of light output over a given area, such as parking lots and commercial areas & walkways. (Using photometric software, a property layout can be entered, designed, and adjusted.)

**Obtrusive light.** Spill light that causes glare, annoyance, discomfort, or loss of visual ability.

**Spill light.** Light from a lighting installation that falls outside of the boundaries of the property on which it is located. Usually results in obtrusive light.

## **Chapter 6.00 ADMINISTRATION AND PROCEDURES**

## § 6.01 Planning and Zoning Commission

(a) Planning and Zoning Commission created. There is hereby created an advisory board known as the Planning and Zoning Commission.

# (b) Organization.

- (1) The commission shall consist of five at-large regular members and may include two (2) alternates. All members shall be qualified electors of the town appointed by the mayor and ratified by a majority vote of the board of trustees.
- In addition to the seven (7) members, there shall be at least one ex-officio member from the Board of Trustees which shall be the Mayor or his/her designee. There shall also be one (1) ex-officio member who shall be the Town Fire Marshal. The Town Board may appoint two (2) non-resident ex-officio members of the Commission.
- (3) Alternate members may not vote, except in the absence of a regular member; and the exofficio members may not ever vote. All regular members, alternate members and ex-officio members may participate in all discussions.
- (4) Each non-resident ex-officio member must be a resident of Wright's Mesa, but not the Town of Norwood.
- (5) Each regular and alternate member of the Commission shall be a resident citizen and qualified elector of the Town of Norwood, Colorado, at the time of his or her appointment. A member who ceases to reside in the Town during his or her term of office shall immediately forfeit his or her office.
- (6) The term of office of regular, alternate and ex-officio members of the Commission shall be for two (2) years; except that the term of office of the ex-officio member from the Board of Trustees, the Mayor and the ex-officio member who is the Fire Marshal shall coincide with their other respective and relevant positions. The three (3) regular members shall be appointed to two (2) year terms ending on Feb 1 in even years and three (3) members shall be appointed for two (2) year terms beginning on Feb 1 in odd years.
- (7) All members of the Commission shall serve without compensation.
- (8) Other qualifications for regular membership and for non-resident ex-officio membership shall be a genuine interest in the planning and zoning of the Town, ability to work with the public and time to devote to this service.
- (9) Regular members, alternates, and non-resident ex-officio members of the Commission shall be appointed by the Mayor and all appointments shall be subject to ratification by the Board of Trustees in accordance with established procedures.

- (c) Officers and procedures. At the first scheduled Commission meeting in September of each year, or as soon thereafter as practicable, the first item of business shall be the selection of the Commission's chairperson and vice-chairperson from among the membership of the Commission.
  - (1) The chairperson shall preside over meetings. In the event a question over procedures arise, Robert's Rules of Order shall prevail.
  - (2) The vice-chairperson shall assist the chairperson in directing the affairs of the Planning and Zoning Commission. In the absence of the chairperson, the vice-chairperson shall assume the duties of the chairperson.
  - (3) The recording of minutes of all Commission meetings shall be the responsibility of the Town staff.
  - (4) A majority of the Commission shall constitute a quorum to do business and the affirmative vote of the majority of those attending shall be necessary to pass any motion.
  - (5) The Commission shall establish rules and procedures that govern its operation.
- (d) **Powers and duties.** The Planning and Zoning Commission shall have the following powers and duties:
  - (1) To recommend the boundaries of the various zoning districts and appropriate regulations to be enforced therein under this Title or the laws of the State of Colorado to the Board of Trustees and to recommend approval or denial of zoning changes and regulations under this chapter.
  - (2) To hear, recommend or determine any matter relating to zoning, planning or subdivision control as they may be specified or required under this Title or applicable laws of the State of Colorado.
  - (3) To make and adopt a comprehensive plan for the physical development of the Town, including any areas outside its boundaries, subject to the approval of the Board of Trustees, which in the Commission's judgment bear relation to the planning of the Town of Norwood (C.R.S. 31-23-306).
  - (4) To exercise the duties and powers as may be now or hereafter conferred by this Title or applicable laws of the State of Colorado.
  - (5) To hear and decide special exceptions to the terms of this Title, pursuant to Sec. 6.16, Special exceptions.
- (e) Allocation of Powers and Duties. The Board of Trustees, by enactment of this Title, hereby assumes and shall have the authority to exercise all powers granted to and duties placed upon the Planning and Zoning Commission by Colorado law and reserves the authority to delegate by Ordinance to the Planning and Zoning Commission any power granted to or duty placed upon the Board of Trustees by part 2, Article 23, Title 31 of the Colorado Revised Statutes.

# § 6.02 Zoning Map and Land Use Code Amendments

(a) Initiation of amendment. Any person having a proprietary interest in any property may apply to the Board of Trustees for a change or amendment to the provisions of this Title, or the Planning and Zoning Commission may on its own motion or on request from the Board of Trustees, institute study and proposal for changes and amendments in the public interest.

# (b) Application for zoning map amendment.

- (1) Any person having a proprietary interest in any property within the corporate limits of the Town of Norwood, Colorado, requesting a change or amendment to the zoning classification of such property shall file ten (10) copies of an application requesting such change or amendment with the Town Planner, which application shall clearly state the requested change or amendment and describe the property to be affected by such request by metes and bounds or by other legal description. The application shall include signature of property owner. If the applicant is not the owner of the property, s/he shall also submit a letter from the property owner designating the applicant as his/her agent with the authority to represent the application.
- (2) The application shall be accompanied by a title certificate from a licensed title company or attorney listing the name of the property owner(s) and all liens, easements and judgments of record that affect the title to the subject property.
- (3) Applications for map amendments shall include:
  - a. A legal description of land area to be rezoned along with a sketch to scale showing the boundaries of the area to be rezoned which indicates the current zoning for all areas adjacent to the area proposed to be rezoned;
  - b. A statement of the requested new zone district classification and justification for the rezoning;
  - c. Description and sketches of buildings or uses proposed in the area proposed to be rezoned, along with a description of land and building uses within one hundred feet (100') in all directions;
  - d. A time schedule for any contemplated new construction or uses; and
  - e. A description of the effect that the rezoning would have on uses of adjacent properties in the neighborhood of the area proposed to be rezoned and on the town generally. The determination of the "neighborhood of the area proposed to be rezoned" shall be made by the planning and zoning commission and/or the board of trustees.
- (4) Applications for code amendments shall include the precise wording of the proposal

- along with an explanation of why the amendment should be adopted.
- (5) A filing fee shall be submitted to cover the cost of review and processing with every petition in accordance with the fee schedule adopted by resolution of the board of trustees.
- (c) Review by Planning and Zoning Commission. Before taking action on any proposed amendment, supplement or change, the governing body shall submit the same to the Planning and Zoning Commission for its recommendation and report.
  - (1) Public hearing required. The Planning and Zoning Commission shall hold a public hearing on any application for amendment or change prior to making its recommendation to the Board of Trustees.
  - (2) Notification requirements for zoning map amendment. When any such amendment or change relates to a change in the zoning classification of property or a change to the boundary of a zoning district, the applicant shall mail the written notice of public hearing before the Planning and Zoning Commission on the proposed amendment or change, after obtaining a copy of the notice from the Town Staff, to all owners of real property lying within one hundred feet (100') of the property on which the change is requested. The notice shall be given not less than fifteen (15) days before the date set for hearing by depositing in the mail such notice properly addressed and postage paid to each such owner as the ownership appears on the last approved County assessor records. Notice shall be mailed to all owners of mineral estates of record in the records of the San Miguel county clerk and recorder. Applicants shall provide proof of mailing of notice to the Town prior to the public meeting or hearing that is the subject of the mailing of a notice.
  - (3) Notification requirements for text amendment. When any such amendment relates to a change of a regulation or to the text of this chapter not affecting specific property, the Town Clerk shall cause notice of the public hearing of the Planning and Zoning Commission to be given by one publication in a newspaper of general circulation in the Town of Norwood without the necessity of notifying property owners by mail. Notice shall be mailed to all owners of mineral estates of record in the records of the San Miguel county clerk and recorder. Such notice shall state the time and place of such hearing and the nature of the subject to be considered, which time shall not be earlier than fifteen (15) days from the date of publication.
- (d) Action by Board of Trustees. A public hearing shall be held by the governing body before adopting any proposed amendment, supplement or change.
  - (1) The Town Clerk shall cause notice of the public hearing of the Board of Trustees to be given by publication in a newspaper of general circulation in the Town of Norwood without the necessity of notifying property owners by mail. Such notice shall state the time and place of such hearing and the nature of the subject to be considered, which time shall not be earlier than fifteen (15) days from the date of publication.

- (2) Such amendments shall not become effective except by the favorable vote of a majority of the quorum of the Board of Trustees present and voting.
- (3) When any such amendment or change relates to a change in the zoning classification of property or a change to the boundary of a zoning district and a protest against such proposed amendment has been filed with the Town Clerk, duly signed and acknowledged by the owners of twenty percent (20%) or more, either of the area of land included in such a proposed change or twenty percent (20%) of the land adjacent thereto extending one hundred feet (100') therefrom, disregarding intervening public streets and alleys, such amendments shall not become effective except by a two-thirds (2/3) vote of the Board of Trustees.
- **(e) Standards for Approval.** The Planning and Zoning Commission and the Board of Trustees may approve an amendment to the zoning map or to the land use code when any one of the following criteria has been met:
  - (1) There has been a substantial change in conditions in the neighborhood or area proposed to be rezoned since the date of approval of the existing zoning map designation;
  - (2) There has been a substantial change in the circumstances or conditions of the town at large;
  - (3) There is demonstrated to be an error in the existing zoning or land use code text;
  - (4) The proposed amendment is in conformance with or would implement the Norwood Master Plan, Major Streets Plan/Future Land Use Plan, as amended;
  - (5) There exists a substantial and compelling public interest in such amendment; or
  - (6) The amendment protects the public health, welfare and safety of the town and its residents.

#### § 6.03 Master Plan Amendments.

(a) Initiation of Amendment. Any person having a proprietary interest in any property may petition the Board of Trustees for a change or amendment to the Town of Norwood Comprehensive Plan ("Master Plan"), or the Planning and Zoning Commission may on its own motion or on request from the Board of Trustees, institute study and proposal for changes and amendments in the public interest.

### (b) Application for Master Plan Amendment.

- (1) Any person having a proprietary interest in any property within the corporate limits of the Town of Norwood, Colorado, requesting a change or amendment to the master plan affecting such property, shall file ten (10) copies of an application for such change or amendment with the Town Planner, which petition shall clearly state the requested change or amendments with the Town Planner. The application shall include signature of property owner. If the applicant is not the owner of the property, s/he shall also submit a letter from the property owner designating the applicant as his/her agent with the authority to represent the application.
- (2) The application shall clearly include:
  - a. Requested change in designation for property;
  - b. Legal description of affected property;
  - c. Rationale for change and supporting information to justify change;
  - d. Map showing area included in request.
- (3) A filing fee shall be submitted to cover the cost of review and processing with every private sector application in accordance with the fee schedule adopted by resolution of the Board of Trustees.
- (c) Review by Planning and Zoning Commission. Before taking action on any proposed amendment, supplement or change, the governing body shall submit the same to the Planning and Zoning Commission for its recommendation and report.
  - (1) Public hearing required. Before adopting any plan, any part, amendment, extension or addition to the plan, the Planning and Zoning Commission shall hold a public hearing prior to making its recommendation to the Board of Trustees.
  - Notification requirements for master plan amendment. The Town Clerk shall cause notice of the public hearing to be given by one publication in a newspaper of general circulation in the Town of Norwood without the necessity of notifying property owners by mail. Such notice shall state the time and place of such hearing and the nature of the amendment to be considered, which time shall not be earlier than fifteen (15) from the date

- of publication. Notice shall also be mailed to all owners of mineral estates of record in the records of the San Miguel county clerk and recorder.
- (3) Requirements for approval. Such amendments shall not become effective except by resolution of the Commission carried by the favorable vote of a two-thirds of the entire membership of the Commission. The resolution shall refer expressly to the maps and descriptive matter intended by the commission to form the whole or part of the plan, and the action taken shall be recorded on the map and plan and descriptive matter by the identifying signature of the chairperson or secretary of the Commission. An attesting copy of the plan or part thereof shall be certified to each governmental body of the territory affected and, after the approval by each body, shall be filed with the county clerk and recorder of San Miguel County.
- (4) Board notification of disapproval. In the event of a disapproval by the Commission of a proposed master plan or an amendment to the plan, the commission shall communicate its reasons to the Board of Trustees.
- (5) Statutory approval requirement. Failure of the Commission to act within sixty days from and after the date of official submission to it shall be deemed approval.
- (d) Action by Board of Trustees. A public hearing shall be held by the governing body before adopting any proposed amendment, supplement or change.
  - (1) Notification requirements for master plan amendment. The Town Clerk shall cause notice of the public hearing to be given by one publication in a newspaper of general circulation in the Town of Norwood without the necessity of notifying property owners by mail. Notice shall be mailed to all owners of mineral estates of record in the records of the San Miguel county clerk and recorder. Such notice shall state the time and place of such hearing and the nature of the amendment to be considered, which time shall not be earlier than fifteen (15) days from the date of publication.
  - (2) Positive Commission recommendation. If the Commission has recommended approval, such plan or amendment shall not become effective except by the favorable vote of a majority of the quorum of the Board of Trustees present and voting.
  - (3) Negative Commission recommendation. If the Commission has recommended denial, such plan or amendment shall not become effective unless the Board of Trustees overrules such disapproval by a recorded vote of not less than two-thirds of its entire membership.
- (e) Standards for approval. The Planning and Zoning Commission and Board of Trustees shall evaluate the request and consider changes in circumstances since the adoption of the Master Plan.

## § 6.04 Preliminary Plat Procedures

- (a) Pre-application conference. Prior to the filing of a preliminary plat, the subdivider shall meet with the Town Planner or his or her designated agent to acquaint himself or herself with the requirements of the Town and the relationship of the proposed subdivision to the comprehensive plan. At such meeting, the application contents, referral agencies, review procedures, density standards, use and area standards, street requirements, utility service and the general character of the development may be discussed. At the pre-application conference, the subdivider may be represented by a land planner, engineer or surveyor. Applicants are encouraged to meet with the R-2 School District prior to submitting an application for subdivision.
- **Submittal requirements.** The subdivider or owner shall file fifteen (15) copies of an application requesting preliminary plat approval, a title certificate from a licensed title company or attorney listing the name of the property owner(s) and all liens, easements and judgements of record affecting the subject property, and of the preliminary plat. The application shall include signature of property owner. If the applicant is not the owner of the property, s/he shall also submit a letter from the property owner designating the applicant as his/her agent with the authority to represent the application. Such plat shall be accompanied by or show the following information:
  - (1) Boundary lines and bearings. Boundary lines, bearings, and distances sufficient to locate the exact area proposed for subdivision. At least one (1) subdivision corner shall be referenced to a survey (abstract) corner. The area, in acres, of the subdivision shall also be shown.
  - (2) Adjacent subdivisions. The name and location of a portion of adjoining subdivisions shall be drawn to the same scale and shown in dotted lines adjacent to the tract proposed for subdivisions in sufficient detail to show actually the existing streets and alleys and other features that may influence the layout and development of the proposed subdivisions. Where adjacent land is not subdivided, the name of the owner of the adjacent tract shall be shown.
  - (3) Intersecting streets. The angle of intersection of the centerline of all intersecting streets.
  - (4) Proposed streets, alleys and easements. The names, location and widths of all streets, alleys and easements proposed for the subdivision, and all known right-of-ways and/or easements within or affecting the area to be subdivided.
  - (5) Proposed blocks, lots and parks. The subdivision shall show all proposed streets and alleys, easements, blocks, lots, parks, etc., with principal dimensions.
  - (6) Contours. Topographic contours at five foot (5') intervals and all easements or right-of-way necessary for drainage within or without the boundaries of the addition.
  - (7) Subdivision title and planner. The title under which the proposed subdivision is to be recorded, the name of the owner and the name of the engineer or land planner who prepared the plat.

- (8) Dedicated parks, playgrounds and other public uses. Sites, if any, to be reserved or dedicated for parks, playgrounds or other public uses.
- (9) Scale, north point. Scale, north point, date and other pertinent data. The scale of the preliminary plat may be at one inch (1") equals twenty feet (20').
- (10) Name, address and telephone number. Property owner's name, address, and telephone number.
- (11) Proposed layout of utilities. A proposed preliminary layout of sanitary sewer and water lines to serve the subdivision.
- (12) Drainage report. A general drainage report or drainage statement shall accompany the preliminary plat. This study or report shall show the acreage draining into the subdivision, points of runoff through and away from the subdivision.
- (13) Protective covenants. Draft of any protective covenants where the subdivider proposes to regulate land use or development standards in the subdivision.
- (14) Proposed land uses. A designation of the proposed uses of land within the subdivision and any zoning amendments proposed to be requested.
- (15) Vicinity map. A vicinity map on a smaller scale showing the proposed subdivision and its relationship to the surrounding area and Town limits.
- (16) Application fee. A filing fee shall be submitted to cover the cost of review and processing with every preliminary plat in accordance with the fee schedule adopted by resolution of the Board of Trustees.

#### (c) Action of Commission.

- (1) The Town Planning and Zoning Commission consideration shall include paving and layout of streets, alleys and sidewalks; means of ingress and egress, provisions for drainage, parking spaces, protective screening and open spaces; areas designated for landscaping; and other aspect deemed by the Town Planning and Zoning Commission necessary to consider in the interest of promoting the public health, safety, order, convenience, prosperity and general welfare.
- Preliminary master plan. If the proposed subdivision is a portion of a tract that is later to be subdivided in its entirety, then a tentative master plat of the entire subdivision shall be submitted with the preliminary plat of the portion first to be subdivided. The master subdivision plan shall conform in all respects to the requirements of the preliminary plat; except, it may be on a scale of not more than one inch (1") to one hundred feet (100').
- (3) Subdivisions outside Town. If the information shown on a preliminary subdivision plat is of land located outside the corporate limits of the Town of Norwood and within Norwood's

Urban Growth Boundary, the procedure for approval, modification or disapproval, shall be the same as required for preliminary plats within the Town.

# (d) Application review procedures.

- (1) Date of filing. Ten (10) copies of the preliminary plat as described in subsection (b) above shall be submitted to the Town Planner thirty (30) days prior to the Planning and Zoning Commission meeting at which consideration is desired. The preliminary plat shall be considered officially filed after application review fees which are established by resolution of the Board of Trustees have been paid and after it is examined and found to be in compliance with the general provisions of these regulations by the Town Planner.
- (2) Distribution of preliminary plats. The following notice shall be stamped on the face of each preliminary plat: "Preliminary Plat for inspection purposes only, and in no way official or approved for record purposes." The Town Planner shall distribute the preliminary plats immediately upon receipt to the following:
  - a. The Town Planner/Consultant (one copy)
  - b. Town Engineer/consultant (one copy)
  - c. Town Public Works Director (one copy)
  - d. Town Attorney (one copy)
  - e. San Miguel Power Association or other power company (one copy)
  - f. Natural Gas company (one copy)
  - g. Telephone company (one copy)
  - h. R-2 School District (one copy)
  - i. Norwood Water Commission (one copy)
  - j. Norwood Sanitation District (one copy)
  - k. Norwood Town Clerk (one copy for public record)
  - 1. Norwood Fire Protection District (one copy)
  - m. US Postal Service (one copy)
  - n. Any other agencies as determined appropriate by the Town Planner Town Planner.
- (3) Distribution to County. Whenever a preliminary subdivision plat, involving land in the

Town's Urban Growth Boundary outside the corporate limits, is submitted to the Planning and Zoning Commission for approval, two (2) additional plat copies shall be provided and forwarded for comment and recommendation to the County Commissioners.

One (1) of the copies shall be specifically directed to the attention of the Commissioner in whose district the subdivision is located and the other to the County Planning Director. The transmittal to the County Commissioners shall specify the date at which the Planning and Zoning Commission will act on the preliminary plat and request a reply setting forth the improvement standards and other requirements essential to the acceptance of streets, roads and other public dedications proposed by the plat.

(4) Comments; written report. At least ten (10) days prior to the meeting of the Planning and Zoning Commission at which the plat is to be considered, each agency listed above shall submit their written recommendations concerning the plat in question to the Town Planner. The recommendations shall be given to the Planning and Zoning Commission with the plat for their consideration. A written report shall be prepared by the Town Planner and submitted to the Planning and Zoning Commission at the next regular meeting. Such report should include comments relative to the proposed subdivision's compliance to these regulations, the comprehensive plan or other master plans such as utility plans. The report may include comments from other municipal departments, county, or state agencies concerned with urban development.

# (e) Review by Commission.

- (1) The Planning and Zoning Commission shall hold a public hearing on a preliminary plat prior to making its recommendation to the Board of Trustees.
- (2) Notification requirements for preliminary plat. When a hearing is to be held on preliminary plat of property, the applicant shall mail the written notice of public hearing before the Planning and Zoning Commission on the proposed preliminary plat, after obtaining a copy of the notice from the Town Staff, to all owners of real property lying within one hundred feet (100') of the property on which the change is requested. Notice shall be mailed to all owners of mineral estates of record in the records of the San Miguel county clerk and recorder. The notice shall be given not less than fifteen (15) days before the date set for hearing by depositing in the mail such notice properly addressed and postage paid to each such owner as the ownership appears on the last approved County assessor records. Applicants shall provide proof of mailing of notice to the Town prior to the public meeting or hearing that is the subject of the mailing of a notice.
- (3) Items for consideration by Commission. The Planning and Zoning Commission shall, in its action on the preliminary plat, consider the physical arrangement of the subdivision, and determine the adequacy of street rights of way and alignment and the compliance of the streets with any Norwood Streets Plan, the existing street pattern in the area and with any other applicable provisions of the comprehensive plan. The Planning and Zoning Commission shall also ascertain that adequate easements for proposed or future utility

service and surface drainage are provided, and that the lot size and area are adequate to comply with the minimum requirements for the type of sanitary sewage disposal proposed. Where connection to an approved sanitary sewage collection and treatment system is not proposed, all lots and building sites shall contain a minimum area of one (1) acre or more depending upon studies of the particular soil condition and evaluation of the physiographic and geologic character of the area being subdivided.

- (4) Action within 30 days. Following review of the preliminary plat and other materials submitted for conformity thereof to these regulations, and negotiations with the subdivider on changes deemed advisable and the kind and extent of improvements to be made by the sub-divider, the Planning and Zoning Commission shall, within thirty (30) days, act thereon as submitted or modified, and if approved the Planning and Zoning Commission shall express its approval as conditional approval and state the conditions of such approval, if any, or if disapproved, shall express its disapproval and its reasons therefore.
- (5) Notation of action. The action of the Planning and Zoning Commission shall be noted on two (2) copies of the preliminary plat, referenced and attached to any conditions determined. One (1) copy shall be returned to the sub-divider and the other retained by the Planning and Zoning Commission. A notation of the action taken and requisite reasons therefore shall be entered in the records of the Planning and Zoning Commission.

# (f) Review by Board of Trustees.

- (1) Submittal to Board of Trustees. The Town Planner shall, at the next regularly scheduled Board of Trustees meeting following conditional approval or disapproval by the Planning and Zoning Commission, submit the preliminary plat with the conditions established by the Planning and Zoning Commission to the Board of Trustees for their consideration.
- (2) Action by Board of Trustees. The Board of Trustees shall approve or disapprove the preliminary plat as to street dedication and utility services either with or without special provisions.

## (g) Effect of preliminary plat approval.

- (1) Not approval of final plat. Conditional approval of a preliminary plat shall not constitute approval of the final plat. Rather, it shall be deemed an expression of approval to the layout submitted on the preliminary plat as a guide to the preparation of the final plat.
- (2) Lapse of approval. Preliminary approval of the subdivision shall be valid for a period of twelve (12) months from the date of approval and the general terms and conditions under which the preliminary approval was granted will not be changed. The Planning and Zoning Commission shall withdraw its preliminary approval of a subdivision unless the final plat is submitted within the twelve (12) month period or unless the twelve (12) month period is extended by the Planning and Zoning Commission at the request of the sub-divider.

#### § 6.05 Final Plat Procedures

- (a) Final plat submittal requirements. The owner of land on which preliminary plat approval has been obtained shall prepare and submit a final plat to the Planning and Zoning Commission. The owner shall provide a non-erasable Mylar copy of the original and six (6) copies drawn to a scale of one hundred feet (100') to one inch (1"). The drawing shall measure twenty-four inches by thirty-six inches (24" x 36"). When necessary the plat may be on several sheets accompanied by an index sheet showing the entire subdivision. For large subdivisions, the final plat may be submitted for approval progressively in sections satisfactory to the Planning and Zoning Commission. The final plat shall show or be accompanied by the following:
  - (1) Control points; acres. The primary control points, or descriptions and "ties" to such control points, to which all dimensions, angles, bearings, and similar data on the plat shall be referred shall be placed on the final plat. The area of the subdivision, in acres, shall be shown.
  - Boundary lines and bearings. Tract boundary line sufficient to locate the exact area proposed for subdivision, right of way lines of streets, easements and other rights of way, and property lines of residential lots and other sites; with accurate dimensions, bearings or deflection angles, and radii, arcs and central angles of all curves shall be placed on the final plat.
  - (3) Streets. Name and right of way width of each street or other right of way shall be placed on the final plat.
  - (4) Easements. Location and dimensions of all easements shall be placed on the final plat.
  - (5) Lot and block numbers. Number to identify each lot or site and each block, and the dimensions of lots and blocks, shall be placed on the final plat.
  - (6) Purpose of sites. The purpose for which sites, other than residential lots, are dedicated or reserved shall be indicated on the final plat.
  - (7) Building lines. Minimum building setback lines when required or approved by the Planning and Zoning Commission shall be placed on the final plat.
  - (8) Monuments. Location and description of monuments shall be placed on the final plat.
  - (9) Adjacent land. References to recorded subdivision plats or adjoining platted land by record name shall be placed on the final plat.
  - (10) Legal description. A legal description and surveyor's certificate, to, in the following form, shall be placed on the final plat:

Town Clerk

	KNOW ALL MEN BY THESE PRESENTS:  That I,, do hereby certify that I prepared this plat from an actual and accurate survey of the land and that the corner monuments shown thereon were properly placed under my supervision.			
	Signature			
(11)	Approval certification. Certification of approval by the Planning and Zoning Commission and Board of Trustees, in the following form, shall be placed on the final plat.			
Commission	APPROVED this day of of the Town of Norwood, Colorado.	, 20	_, by the Planning and Zoning	
	Chairman			
the Town of I	APPROVED this day of Norwood, Colorado.	, 19	_, by the Board of Trustees of	
	Mayor			

- (12) Title; scale. A title, scale, and north point shall be placed on the final plat.
- (13) Street intersections. The location of the point of intersection and points of tangency of street intersections, and the bearing and distance of each street right of way center line shall be placed on the final plat.
- (14) Plat identification. A positive reference and identification of the plat and date of plat shall be placed on the final plat.
- (15) Dedication certificate. The property owner's certificate and deed of dedication shall be placed on the final plat. The dedication deed or certificate of dedication shall be executed by all persons, firms or corporations owning an interest in the property subdivided and platted, and shall be acknowledged in the manner prescribed by the laws for the State of Colorado for conveyances of real property. In the case of surface lien holders, they may execute a subordination agreement subordinating their liens to all public streets, alleys, parks, school sites and any other public areas shown on the plat of such subdivision as being set aside for public uses and purposes. The dedication deed or certificate of dedication shall, in addition to the above requirements, contain the following:

- a. An accurate description of the tract of land subdivided.
- b. A statement and express representation that the parties joining in such dedication are the sole owners of such tract of land.
- c. An express dedication without reservation to the public for public use; the streets, alleys, rights of way, school site and any other public areas shown on the attached plat.
- d. A positive reference and identification of the plat of such subdivision date of plat and engineer.
- (16) Tax certificates. Tax certificates indicating that all taxes on the land being subdivided have been paid to the current year shall be submitted with the final plat.
- (17) Construction plans. Three (3) sets of plans for required improvements and a set of reproducible transparent sheets, 23" x 36" in size along with all data and calculations related to utilities, drainage or other construction in the subdivision shall be submitted with the final plat. Such plans shall also show all existing or proposed surface and subsurface improvements and obstruction.
- (18) Statutory notice of vested rights.

### (b) Application review procedures.

- (1) Date of filing. After approval of the preliminary plat by the Planning and Zoning Commission and Board of Trustees and within twelve (12) months of the approval date unless extended by action of the Planning and Zoning Commission, the sub-divider may submit for approval the final plat. The application, meeting all the requirements of subsection (a) above, shall be submitted to the Town Planner at least thirty (30) days prior to the meeting at which consideration is desired. The official filing date of the final plat shall be the date upon which the plat and construction drawings are found to be in full compliance with the provisions of the preliminary approval after examination by the Town Planner.
- (2) Conformance with preliminary plat. The final plat shall conform substantially to the preliminary plat as approved and, if desired by the sub-divider, it may constitute only that portion of the approved preliminary plat that he or she proposes to record and develop at the time, provided however, that such portion conforms to all requirements of these regulations.
- (3) Review of construction plans. After the presentation of the construction plans for a subdivision to the Town Planner, the Town Planner shall submit the construction plans to the Town Engineer/Consultant for review. The Town Engineer/Consultant shall review the plans and submit to the Planning and Zoning Commission his or her report at the final plat presentation. The developer shall pay the reasonable cost of review of the construction plans before the final plat is presented to the Board of Trustees.

## (c) Review by Commission.

- (1) Action by Commission. The final plat shall be presented to the Planning and Zoning Commission at its next regular meeting, along with any appropriate recommendation by the Town Planner. The Planning and Zoning Commission shall act on the final plat within thirty (30) days after the official filing date or within a reasonable time thereafter.
- (2) Review in stages. An owner or sub-divider, at his or her option, may obtain approval of a portion or a section of a subdivision provided he or she meets all the requirements of this chapter with reference to such portion or section in the same manner as is required for a complete subdivision. In the event a subdivision and the final plat thereof is approved by the Planning and Zoning Commission in sections, each final plat of each section is to carry the name of the entire subdivision, but is to bear a distinguishing letter, number or subtitle. Block numbers shall run consecutively throughout the entire subdivision, even though such subdivision may be finally approved in sections.
- (3) Approval by Commission. After the Planning and Zoning Commission has determined that the plat is in proper form, that the arrangement of the development proposed for the property being subdivided is consistent with zoning regulations and that the subdivision complies with the provisions of this chapter, it shall act to approve the plat, subject to action by the Board of Trustees.
- (4) Disapproval by Commission. Final plats that are disapproved by the Planning and Zoning Commission shall be returned to the sub-divider by the Town Planner with an attached statement of the reasons for such action.
- (5) Appeal. Any sub-divider aggrieved by any finding or action of the Planning and Zoning Commission that involves the jurisdiction of the Board of Trustees as to dedication or improvements, shall have the right of appeal to the Board of Trustees within thirty (30) days from the date of such finding or action but any appeal must be taken within thirty (30) days of such action and not thereafter.
- (d) Review by Board of Trustees. The Board of Trustees shall consider all proposals with respect to the dedication of right of way for public use, the construction of utilities, streets, drainage, and other improvements, and when satisfied with the proposals, shall authorize the establishment of agreements for same and ratify the approval action of the Planning and Zoning Commission relative to the final plat.

## (e) Action following approval.

- (1) Certification of approval. The Planning and Zoning Commission's approval and the Board of Trustees's ratification of the final plat shall authorize the Chairman of the Planning and Zoning Commission to execute the certificate of approval on the original copy of the final plat.
- (2) Recordation of plats. The final plat for any subdivision located within the corporate limits

- of the Town of Norwood shall then be caused to be filed of record by the sub-divider in the plat records of San Miguel County, but only after the Board of Trustees has officially acted upon the final plat with reference to improvements, dedications and utilities. Recording fees shall be paid by the developer.
- (3) Sign off on construction plans. Three sets of final construction plans shall be provided by the sub-divider and signed by the sub-divider and the Board of Trustees or the Boards' designee and retained as a record by the Town Clerk.

#### § 6.05A Small Scale Subdivision Procedures

- (a) Pre-application conference. Prior to the filing of a small scale subdivision plat, the subdivider shall meet with the Town Planner or his or her designated agent to acquaint himself or herself with the requirements of the Town and the relationship of the proposed subdivision to the comprehensive plan. At such meeting, the application contents, referral agencies, review procedures, density standards, use and area standards, street requirements, utility service and the general character of the development may be discussed. At the pre-application conference, the subdivider may be represented by a land planner, engineer or surveyor.
- **Submittal requirements.** The subdivider or owner shall file fifteen (15) copies of an application requesting small scale subdivision plat approval, a title certificate from a licensed title company or attorney listing the name of the property owner(s) and all liens, easements and judgments of record affecting the subject property, and of the small scale subdivision plat. The application shall include signature of property owner. If the applicant is not the owner of the property, s/he shall also submit a letter from the property owner designating the applicant as his/her agent with the authority to represent the application. Such plat shall be accompanied by or show the following information:
  - (1) Boundary lines, bearings, monuments and control points. Tract boundary line sufficient to locate the exact area proposed for subdivision, right-of-way lines of streets, easements and other rights-of-way, and property lines of residential lots and other sites; with accurate dimensions, bearings or deflection angles, and radii, arcs and central angles of all curves shall be placed on the final plat.

The primary control points, or descriptions and "ties" to such control points, to which all dimensions, angles, bearings, and similar data on the plat shall be referred shall be placed on the final plat. The area of the subdivision shall be shown.

Location and description of monuments shall be placed on the plat.

- (2) Adjacent subdivisions. The name and location of a portion of adjoining subdivisions shall be drawn to the same scale and shown in dotted lines adjacent to the tract proposed for subdivisions in sufficient detail to show actually the existing streets and alleys and other features that may influence the layout and development of the proposed subdivisions. Where adjacent land is not subdivided, the name of the owner of the adjacent tract shall be shown. References to recorded subdivision plats or adjoining platted land by record name shall be placed on the plat.
- (3) Intersecting streets. The location of the point of intersection and points of tangency of street intersections, and the bearing and distance of each street right-of-way center line shall be placed on the final plat.
- (4) Proposed streets, alleys and easements. The names, location, dimensions and widths of all streets, alleys and easements proposed for the subdivision, and all known right-of-ways and/or easements within or affecting the area to be subdivided.

- (5) Proposed blocks, lots and parks. The subdivision shall show all proposed streets and alleys, easements, blocks, lots, parks, etc., with principal dimensions. Each lot, block and site shall be numbered to identify.
- (6) Contours. Topographic contours at five foot (5') intervals and all easements or right-of-way necessary for drainage within or without the boundaries of the addition.
- (7) Subdivision title and planner. The title under which the proposed subdivision is to be recorded, the name of the owner and the name of the engineer or land planner who prepared the plat.
- (8) Dedicated parks, playgrounds and other public uses. Sites, if any, to be reserved or dedicated for parks, playgrounds or other public uses.
- (9) Scale, north point. Scale, north point, date and other pertinent data. The scale of the small scale subdivision plat may be at one inch (1") equals twenty feet (20').
- (10) Name, address and telephone number. Property owner's name, address, and telephone number.
- (11) Proposed layout of utilities. Layout of sanitary sewer and water lines to serve the subdivision.
- (12) Drainage report. A general drainage report or drainage statement shall accompany the small scale subdivision plat if required at the pre-application conference. This study or report shall show the acreage draining into the subdivision, points of runoff through and away from the subdivision.
- (13) Protective covenants. Draft of any protective covenants where the subdivider proposes to regulate land use or development standards in the subdivision, if identified at the preapplication meeting.
- (14) Purpose of sites. The purpose for which sites, other than residential lots, are dedicated or reserved shall be indicated on the plat.
- (15) Vicinity map. A vicinity map on a smaller scale showing the proposed subdivision and its relationship to the surrounding area and Town limits.
- (16) Application fee. A filing fee shall be submitted to cover the cost of review and processing with every small scale subdivision plat in accordance with the fee schedule adopted by resolution of the Board of Trustees.
- (17) Legal description. A legal description and surveyor's certificate, to, in the following form, shall be placed on the final plat:

#### KNOW ALL MEN BY THESE PRESENTS:

	I,, do hereby certify tha	
	arate survey of the land and that the corner monuter my supervision.	ments snown thereon were properly placed
unac	er my supervision.	
	Signature	
(18)	Approval certification. Certification of approvand Board of Trustees, in the following form, s	
	ROVED this day of, nmission of the Town of Norwood, Colorado.	19, by the Planning and Zoning
	Chairman	
	APPROVED this day of	, 19, by the Board of Trustees of
Town of I	Norwood, Colorado.	•
	Mayor	
	Town Clerk	

- (19) Title; scale. A title, scale, and north point shall be placed on the final plat.
- (20) Plat identification. A positive reference and identification of the plat and date of plat shall be placed on the final plat.
- Dedication certificate. The property owner's certificate or deed of dedication shall be placed on the final plat. The dedication deed or certificate of dedication shall be executed by all persons, firms or corporations owning an interest in the property subdivided and platted, and shall be acknowledged in the manner prescribed by the laws for the State of Colorado for conveyances of real property. The spouse of all married persons executing such dedication deed or certificate of dedication shall join therein unless satisfactory proof be provided showing that the property to be subdivided does not constitute any portion of such party's homestead, in which case the instrument of dedication shall state the fact that the property subdivided and platted does not constitute a part of such party's actual homestead. In the case of surface lien holders, they may execute a subordination agreement subordinating their liens to all public streets, alleys, parks, school sites and any other public areas shown on the plat of such subdivision as being set aside for public uses and purposes. The dedication deed or certificate of dedication shall, in addition to the above requirements, contain the following:

- a. An accurate description of the tract of land subdivided.
- b. A statement and express representation that the parties joining in such dedication are the sole owners of such tract of land.
- c. An express dedication without reservation to the public for public use; the streets, alleys, right-of-ways, school site and any other public areas shown on the attached plat.
- d. A positive reference and identification of the plat of such subdivision, date of plat and engineer.
- (22) Tax certificates. Tax certificates indicating that all taxes on the land being subdivided have been paid to the current year shall be submitted with the final plat.
- (23) Statutory notice of vested rights.
- (24) Construction plans, if any. Three (3) sets of plans for required improvements and a set of reproducible transparent sheets, 23" x 36" in size along with all data and calculations related to utilities, drainage or other construction in the subdivision shall be submitted with the final plat. Such plans shall also show all existing or proposed surface and subsurface improvements and obstruction.
- **(c)** Application review procedures.
  - (1) Date of filing. The application, meeting all the requirements of subsection (a) above, shall be submitted to the Town Planner at least thirty (30) days prior to the meeting at which consideration is desired. The small scale subdivision plat shall be considered officially filed after application review fees which are established by resolution of the Board of Trustees have been paid and after it is examined and found to be in compliance with the general provisions of these regulations by the Town Planner.
  - (2) Distribution of plans. The Town staff shall distribute the plans immediately upon receipt to the following:
    - a. Town Planner/Consultant (one copy)
    - b. Town Engineer/Consultant (one copy)
    - c. Town Public Works Director (one copy)
    - d. San Miguel Power Association or other electricity provider (one copy)
    - e. Natural gas provider (one copy)
    - f. Centurylink or other telephone provider (one copy)

- g. R-2 School District (one copy)
- h. Norwood Water Commission (one copy)
- i. Norwood Sanitation District (one copy)
- j. Norwood Fire District (one copy)
- k. Norwood Town Clerk (one copy for the public record)
- (3) Distribution to County. Whenever a small scale subdivision plat, involving land in the Town's Urban Growth Boundary outside the corporate limits, is submitted to the Planning and Zoning Commission for approval, two (2) additional plat copies shall be provided and forwarded for comment and recommendation to the County Commissioners. One (1) of the copies shall be specifically directed to the attention of the Commissioner in whose district the subdivision is located and the other to the County Planning Director. The transmittal to the County Commissioners shall specify the date at which the Planning and Zoning Commission will act on the small scale subdivision plat and request a reply setting forth the improvement standards and other requirements essential to the acceptance of streets, roads and other public dedications proposed by the plat.
- (4) Comments; written report. At least ten (10) days prior to the meeting of the Planning and Zoning Commission at which the plat is to be considered, each agency listed above shall submit their written recommendations concerning the plat in question to the Town Planner. The recommendations shall be given to the Planning and Zoning Commission with the plat for their consideration. A written report shall be prepared by the Town Planner and submitted to the Planning and Zoning Commission at the next regular meeting. Such report should include comments relative to the proposed subdivision's compliance to these regulations, the comprehensive plan or other master plans such as utility plans. The report may include comments from other municipal departments, county, or state agencies concerned with urban development.
- (5) Review of construction plans. After the presentation of the construction plans for a subdivision to the Town Planner, the Town Planner shall submit the construction plans to the Town Engineer/Consultant for review. The Town Engineer/Consultant shall review the plans and submit to the Planning and Zoning Commission his or her report at the final plat presentation. The developer shall pay the reasonable cost of review of the construction plans before the final plat is presented to the Board of Trustees.
- (d) Review by Commission.
  - (1) The Planning and Zoning Commission shall hold a public hearing on a small scale subdivision plat prior to making its recommendation to the Board of Trustees.

- (2) Notification requirements for small scale subdivision plat. When a hearing is to be held on a small scale subdivision plat of property, the applicant shall mail the written notice of public hearing before the Planning and Zoning Commission on the proposed small scale subdivision plat, after obtaining a copy of the notice from the Town Staff, to all owners of real property lying within one hundred feet (100') of the property on which the change is requested. Notice shall be mailed to all owners of mineral estates of record in the records of the San Miguel county clerk and recorder. The notice shall be given not less than fifteen (15) days before the date set for hearing by depositing in the mail such notice properly addressed and postage paid to each such owner as the ownership appears on the last approved County assessor records. Applicants shall provide proof of mailing of notice to the Town prior to the public meeting or hearing that is the subject of the mailing of a notice.
- **(3)** Items for consideration by Commission. The Planning and Zoning Commission shall, in its action on the plat, consider the physical arrangement of the subdivision, and determine the adequacy of street rights-of-way and alignment and the compliance of the streets with any Norwood Streets Plan, the existing street pattern in the area and with any other applicable provisions of the comprehensive plan. The Planning and Zoning Commission shall also ascertain that adequate easements for proposed or future utility service and surface drainage are provided, and that the lot size and area are adequate to comply with the minimum requirements for the type of sanitary sewage disposal proposed. Where connection to an approved sanitary sewage collection and treatment system is not proposed, all lots and building sites shall contain a minimum area of one (1) acre or more depending upon studies of the particular soil condition and evaluation of the physiographic and geologic character of the area being subdivided. After the Planning and Zoning Commission has determined that the plat is in proper form, that the arrangement of the development proposed for the property being subdivided is consistent with zoning regulations and that the subdivision complies with the provisions of this chapter, it shall act to approve the plat, subject to action by the Board of Trustees.
- (4) Action by Planning and Zoning Commission. Following review of the small scale subdivision plat and other materials submitted for conformity thereof to these regulations, and negotiations with the subdivider on changes deemed advisable and the kind and extent of improvements to be made by the subdivider, the Planning and Zoning Commission shall act thereon as submitted or modified, and if approved the Planning and Zoning Commission shall express its approval as conditional approval and state the conditions of such approval, if any, or if disapproved, shall express its disapproval and its reasons therefor.
- (5) Notation of action. The action of the Planning and Zoning Commission shall be noted on two (2) copies of the small scale subdivision plat, referenced and attached to any conditions determined. One (1) copy shall be returned to the subdivider and the other retained by the Planning and Zoning Commission. A notation of the action taken and requisite reasons therefor shall be entered in the records of the Planning and Zoning Commission.
- (6) Disapproval by Commission. Small scale subdivision plats that are disapproved by the Planning and Zoning Commission shall be returned to the subdivider by the Town Planner

- with an attached statement of the reasons for such action.
- (7) Appeal. Any subdivider aggrieved by any finding or action of the Planning and Zoning Commission that involves the jurisdiction of the Board of Trustees as to dedication or improvements, shall have the right of appeal to the Board of Trustees within thirty (30) days from the date of such finding or action but any appeal must be taken within thirty (30) days of such action and not thereafter.
- (e) Review by Board of Trustees. The Board of Trustees shall consider all proposals with respect to the dedication of right-of-way for public use, the construction of utilities, streets, drainage, and other improvements, and when satisfied with the proposals, shall authorize the establishment of agreements for same and ratify the approval action of the Planning and Zoning Commission relative to the small scale subdivision plat.

### (f) Action following approval.

- (1) Certification of approval. The Planning and Zoning Commission's approval and the Board of Trustees's ratification of the final plat shall authorize the Chairman of the Planning and Zoning Commission to execute the certificate of approval on the original copy of the final plat.
- (2) Recordation of plats. The final plat for any subdivision located within the corporate limits of the Town of Norwood shall then be caused to be filed of record by the subdivider in the plat records of San Miguel County, but only after the Board of Trustees has officially acted upon the final plat with reference to improvements, dedications and utilities. Recording fees shall be paid by the developer.
- (3) Sign off on construction plans, if any. Three sets of final construction plans shall be provided by the subdivider and signed by the subdivider and the Board of Trustees or the Boards' designee and retained as a record by the Town Clerk.

#### § 6.05B Condominium Subdivision and Conversion

- (a) **Purpose.** The purpose of this section is to provide review procedures to ensure that the creation or conversion of condominium subdivisions will comply with the provisions of the Norwood Land Use Code.
- **(b) Definition.** For purposes of this section a condominium is defined as a common interest community in which portions of the real estate are designated for separate ownership and the remainder of which is designated for common ownership solely by owners of the separate ownership portions.
- (c) **Procedures.** The procedures and standards for review and approval of a condominium subdivision shall be the same as that specified for all subdivisions, with applications meeting the definition of a small scale subdivision being processed pursuant to section 6.05A. However, all condominium conversions shall be reviewed as small scale subdivisions without regard to the number of units or size of the parcel proposed for conversion.
- **(d) Compliance.** Condominium subdivisions shall comply with all standards and review criteria generally applicable to subdivisions, as well as, to the extent applicable, the following:
  - (1) Provisions of the Condominium Ownership Act, Section 38-33-101, et seq. C.R.S.
  - (2) Provisions of the Colorado Common Interest Ownership Act. Section 38-33.5-101, et seq.
  - (3) Title 38, Articles 50 and 51, C.R.S.
  - (4) Norwood Land Use Code Section 1.00, General Provisions
  - (5) Norwood Land Use Code Section 3.00, Zoning Regulations
  - (6) Norwood Land Use Code Section 4.00, Subdivision Standards
  - (7) Norwood Land Use Code Section 5.00, Site Development Standards.
  - (8) Norwood Land Use Code Sections 6.04, 6.05, 6.05A, 6.06, 6.07 and 6.08, Subdivision Procedures.
- **(e) Preliminary Plat Submission Requirements.** In addition to the preliminary plat submission requirements contained in Section 6.04(b) or 6.05A (b), as applicable, the applicant shall submit:
  - (1) Fifteen (15) copies of a preliminary condominium plat/map meeting the requirements of a land survey as set forth in Section 38-51-106, C.R.S. and the requirements of a map/plat as set forth in Section 38-33.3-209, C.R.S., being a part of the Colorado Common Interest Ownership Act.
  - (2) A copy of the Condominium Declaration as required by Section 38-33.3-205, C.R.S.,

being a part of the Colorado Common Interest Ownership Act.

- (3) A narrative in sufficient detail to the satisfaction of the Town Planner demonstrating compliance of the condominium project with each of the following, as applicable:
  - a. Norwood Land Use Code Section 1.00, General Provisions
  - b Norwood Land Use code Section 3.00, Zoning District Regulations
  - c. Norwood land Use code Section 4.00, Subdivision Standards
  - d. Norwood Land Use Code Section 5.00, Site Development Standards.
  - e. Norwood Land Use Code Sections 6.06, 6.07 and 6.08, Subdivision Review Procedures
- (f) Final Plat Requirements. In addition to the plat statements specified in Section 6.05 (a) or 6.05A (b), as applicable, the following shall be required on the final condominium plat and shall be worded substantially as follows:

Heading: The heading of the final condominium plat shall include the complete name of the condominium subdivision, the lot, block, or land sections, township, range, principal meridian, "The Town of Norwood, San Miguel County, Colorado." Also, where applicable, the United States mineral claim name, number and mining district shall be shown.

Land Surveyor's Certificate: I( printed name of land surveyor) being a Registered Land Surveyor in the State of Colorado, do hereby certify that this plat and survey of (NAME OF CONDOMINIUM SUBDIVISION IN CAPITAL LETTERS) was made by me and under my supervision and that both are accurate to the best of my knowledge.

Dated this	day of	A.D. 20
(Signature)		
Colorado Registra	ation Number	(Seal)

County Treasurer's Certificate: I certify that according to the records in the San Miguel County Treasurer's office, there are no liens against the property included in the subdivision, or any part thereof, for unpaid state, county or municipal ad valorem taxes or special assessments certified to the County Treasurer for collection.

- (g) Condominium Conversion. In the case of an application to convert an existing structure or development to condominium units, the owner of such property shall demonstrate that the following requirements have been or will be met:
  - (1) The structure or development subject to the proposed condominium conversion meets current off street parking requirements for the underlying zone district found in Section 5.02. Each residential condominium unit shall be considered a separate dwelling unit for purposes of determining parking compliance.
  - (2) There is or will be a one (1) hour fire wall between all units.

- (3) Owners of the property proposed for condominium conversion have or will give the notice to residential tenants of the building required by, and otherwise comply with the requirements of, Section 38-33-112, C.R.S., being a part of the Condominium Ownership Act, as amended. Copies of the notification shall be filed with the Town Clerk as proof of compliance.
- **(h) Recording Prohibited.** No condominium subdivision shall be recorded until the Town determines that it has satisfied all Town requirements and has obtained all required Town approvals.

# § 6.06 Acceptance of Subdivision Improvements

- (a) Time frame for completion.
  - (1) Plan resubmittal. If construction has not commenced within one (1) year after approval of the plans, resubmittal of plans may be required by the Town Engineer for meeting current standards and engineering requirements. These plans will be reviewed and comments noted in fifteen (15) working days. A fee as provided for in the fee schedule adopted by resolution of the Town Board is required upon the resubmittal of plans for review. "Construction" shall mean the start or commencement of construction of Town maintained facilities.
  - (2) Expiration and extension of approval. If the public improvements for a subdivision have not been constructed and accepted by the Town, and the corresponding final plat for said subdivision filed in the plat records of the San Miguel County within eighteen (18) months from the date of final plat approval by the Town, said final plat shall be null and void and shall conclusively be deemed to be withdrawn, without further action by the Town; provided however, this provision shall not apply to final plats approved by the Town prior to the adoption of this Land Use Code (adoption date). If the public improvements for a subdivision that was approved prior to (adoption date) have not been constructed and accepted by the Town, and the corresponding final plat for said subdivision filed in the map and plat records of the San Miguel County by within 18 months of the adoption of the Land Use Code (insert appropriate date) said final plat shall be null and void and shall conclusively be deemed to be withdrawn, without further action by the Town. An approved, unexpired final subdivision plat may be extended once for a period not to exceed twelve (12) months, pursuant to the following provisions:
- a. The Board of Trustees may extend the approval of the final plat, for good cause shown by the applicant, if there has been no significant change in development conditions affecting the subdivision plan and the plat continues to comply with all applicable standards and ordinances.
- b. A request for an extension of time to complete final public improvements for a subdivision pursuant to these provisions shall be submitted to the Town Planner no later than the date the final subdivision plat expires. The request shall be in writing, and the application shall state the reason and justification for the requested extension.
- **(b) Improvements required.** The improvement of all streets, sidewalks, alleys, and drainage ways as herein required shall be in accordance with the standard specifications of the Town of Norwood for installation of such improvements. The Town shall withhold all improvements and services of whatsoever nature, from all additions that have not been approved in accordance with the regulations herein contained.
- (c) Inspection. All construction work, such as street grading, street paving, storm sewers, curb and/or gutter work, sanitary sewers or water mains performed by the owner, developer or contractor, shall be subject to inspection during construction by the proper authorities of the Town and shall be constructed

in accordance with the standard specifications approved by the Board of Trustees, and in accordance with the provisions of any other ordinance of the Town of Norwood applicable thereto.

- (d) As-built plans. Prior to the acceptance of a subdivision by the Town, the engineer for the developer shall submit to the Town staff a complete set of drawings of the paving, drainage, water, and sewer improvements showing all changes made in the plans during construction and containing on each sheet an "As-Built" stamp bearing the signature of the engineer and the date. In addition, one reproducible drawing of the utility plan sheets, containing the "As-Built" information, shall be submitted.
- (e) Maintenance bond. Prior to the acceptance of a subdivision by the Town, the subdivider shall furnish a good and sufficient maintenance bond in the amount of ten percent (10%) of the contract price with a reputable and solvent corporate surety in favor of the Town, to indemnify the Town against any repairs that may become necessary to any part of the construction work performed in connection with the subdivision arising from defective workmanship or materials used therein, for a full period of two (2) years from the date of final acceptance of the entire project.
- **Inspection fee.** Prior to the acceptance of a subdivision by the Town, the subdivider shall reimburse the Town for all design review and inspection costs incurred by the Town for design review and inspection of the water and sewer utilities, drainage facilities, and streets and other public improvements in each subdivision.

# § 6.07 Improvements Agreements and Performance Guarantees

(a) Improvements Agreements. Prior to the issuance of a building permit and the recording of a final plat, an applicant shall submit for approval to the Board of Trustees an improvements agreement for construction of any required public improvements designated on the final plat.

#### (b) Performance Guarantee

- (1) Prior to the issuance of any building permit, the Board of Trustees shall require an applicant to file a financial guarantee in order to insure compliance with any or all requirements of the Board stipulated in the improvements agreement and the final plat.
- (2) The financial guarantee, in the judgment of the Board of Trustees, shall be sufficient to make reasonable provision for completion of said improvements in accordance with design and time specifications.
- (3) Ordinarily, a letter of credit to the Board of Trustees from a commercial bank, savings and loan institution, insurance company or other qualified lending institution(s) licensed or authorized to do business in the State of Colorado, or a letter from the Federal Housing Administration or Veterans Administration in a form satisfactory to the Mayor shall be required.
- (4) Nothing in subsection 6.07 (b)(3) shall preclude the Board of Trustees from approving other forms of financial security.

#### (c) Release of Collateral

- (1) As public improvements are made, an applicant may apply to the Board for release of part or all of the collateral deposited with the Board.
- (2) Upon inspection and approval, the Board shall release collateral, provided that in the event a combination of forms of collateral has been accepted, the Board shall release collateral on a priority basis it deems appropriate.
- (3) If the Board determines that any of the required improvements are not constructed in substantial compliance with specifications, it shall furnish the applicant a list of specifications and shall be entitled to withhold collateral sufficient to insure substantial compliance.
- (4) If the Board determines that the applicant will not construct any or all of the improvements in accordance with all of the specifications, the Board may withdraw and employ from the deposit of collateral such funds as may be necessary to construct the improvements in accordance with the specifications.
- (d) Form of Agreement. All Improvement Agreements shall utilize the standard Town template or guide for the format and content of such Agreements.

# § 6.08 Mandatory Homeowners' Associations

- (a) Applicability. When a residential subdivision contains any physical facilities, structures, improvements, systems, areas or grounds held in common and necessary or desirable for the welfare of the area or subdivision, or that are of common use or benefit and that are not or cannot be satisfactorily maintained by the Town or another public agency, the Town may require the establishment and creation of a mandatory homeowners' association to assume and be responsible for the continuous and perpetual operation, maintenance and supervision of such facilities, structures, improvements, systems, areas or grounds.
- (b) Approval. If the establishment and creation of a mandatory homeowners' association is required by the Town, a copy of the agreements, covenants and restrictions establishing and creating the association must be approved by the Town Attorney and Board of Trustees prior to the approval of the final plat of the subdivision and must be filed of record with said final plat in the Map and Plat Records of San Miguel County, Colorado. Said final plat shall clearly identify all facilities, structures, improvements, systems, areas or grounds that are to be operated, maintained and/or supervised by said association.
- **(c) Responsibilities.** Such mandatory homeowners' associations shall be responsible for the continuous and perpetual operation, maintenance and/or supervision of landscape systems, features or elements located in parkways, common areas, between screening walls or living screens and adjacent curbs or street pavement edges, adjacent to drainage ways or drainage structures, or at subdivision entryways. Subdivision entryway treatments or features shall not be allowed unless a mandatory homeowners' association as required herein is established and created.

- **(d) Dedications to association.** All facilities, structures, improvements, systems, areas or grounds that are to be operated, maintained and/or supervised by a mandatory homeowners' association, other than those located in public easements or right-of-ways, shall be dedicated by easement or deeded in fee simple ownership interest to said association. Such easements or ownership shall be clearly identified on the final plat of the applicable subdivision.
- **(e) Contents of agreements.** At a minimum, the agreements, covenants and restrictions establishing and creating a mandatory homeowners' association required herein shall contain and/or provide for the following:
  - (1) Definitions of terms contained therein;
  - Provisions acceptable to the Town for the establishment and organization of the mandatory homeowners' association and the adoption of by-laws for said association, including provisions requiring that the owner(s) of any lot or lots within the applicable subdivision and any successive buyer(s) shall automatically and mandatorily become a member of the association:
  - (3) The initial term of the agreements, covenants and restrictions establishing and creating the association shall be for a twenty-five (25) year period and shall automatically renew for successive ten (10) year periods, and the association may not be dissolved without the prior written consent of the Town;
  - (4) Provisions acceptable to the Town to ensure the continuous and perpetual use, operation, maintenance, and/or supervision of all facilities, structures, improvements, systems, areas or grounds that are the responsibility of the association and to establish a reserve fund for such purposes;
  - Provisions prohibiting the amendment of any portion of the association's agreements, covenants or restrictions pertaining to the use, operation, maintenance and/or supervision of any facilities, structures, improvements, systems, areas or grounds that are the responsibility of the association without the prior written consent of the Town;
  - (6) The right and ability of the Town or its lawful agents, after due notice to the association, to remove any landscape systems, features or elements that cease to be maintained by the association; to perform the responsibilities of the association if the association fails to do so in compliance with any provisions of the agreements, covenants or restrictions of the association or of any applicable Town codes or regulations; to assess the association for all costs incurred by the Town in performing said responsibilities if the association fails to do so; and/or to avail itself of any other enforcement actions available to the Town pursuant to state law or Town codes or regulations; and
  - (7) Provisions indemnifying and holding the Town harmless from any and all costs, expenses, suits, demands, liabilities or damages, including attorney's fees and costs of suit, incurred or resulting from the Town's removal of any landscape systems, features or elements that cease to be maintained by the association or from the Town's performance of the

aforementioned operation, maintenance or supervision responsibilities of the association due to the association's failure to perform said responsibilities.

#### § 6.09 Replats and Plat Amendments

- (a) Replats. Replats shall be subject to the same requirements as small scale subdivisions regardless of the number of lots created, except as modified in this Section 6.09(a).
  - (1) Replats shall not be subject to the lot size limitations for a small scale subdivision.
  - (2) The submittal requirements are modified as follows:
    - a. Applicant shall submit five (5) copies of the application.
    - b. Topographic contours shall not be required.
    - c. Approvals and Certificates. The following statements shall be required on the plat if applicable and shall be worded substantially as follows:
      - (i) Owner Certificate. Know all persons by these presents: That (printed name of owner(s)), being the owner(s) of the land described as follows: (insert legal description of land being subdivided in the Town of Norwood, San Miguel County, Colorado, has platted same as shown on the plat, and by these presents do(es) hereby dedicate to the perpetual use of the Town of Norwood, San Miguel County, Colorado, the streets, alleys, roads and other public areas as shown hereon and hereby dedicate those portions of land labeled as utility easements for the installation and maintenance of public utilities as shown hereon.

In witness hered	of the said (pri	nted name of owner(s)) has	caused (his, her, their,
its) name to be	here unto subs	cribed thisday of	<b></b> ,
20		-	
BY:			
Notarial acknowled	dgment:		
State of		County of	
The foregoing ins	trument was ackı	nowledged before me this	
	day of	A.D. 20, by	My commission
expires on:	Witness	my hand and official seal.	
		(Seal)	
Notary Public			

(ii) *Mortgagee's Consent* (if applicable): The undersigned (insert name), as a beneficiary of a deed of trust (or identify other mortgage instrument or agreement creating security interest) which constitutes a lien upon the declarant's property, recorded at Reception No.\_, in the office of the San Miguel County Clerk and Recorder, hereby consents to the replatting of the land.

(print name of beneficiary) Notarial:

(iii) *Title Certificate*.\_\_\_\_\_, a Colorado licensed title company, does hereby certify that it has examined the title to the land shown upon this Plat and that

	title to such lands is in the name offree and clear of all liens, taxes, and encumbrances, except as follows:			
(List same or indicate N	NONE)			
Dated	d this	_day of	, A.D. 20	. (Signature)
(iv)	Registered plat and su under my s I further ce Town of N	Land Surveyor in arvey of (insert title supervision and the artify that all monu-	the State of Cole of plat in capital both are accurate ments and mark Code and Artic	e of Land Surveyor) being a lorado, do hereby certify that this al letters) was made by me and rate to the best of my knowledge. The series were set as required by the les 50 and 51 of Title 38, C.R.S.
	(Signature	e)	(Seal) Colorado	Registration No
	NOTES:			
(v)	San Migue included ir municipal	el County Treasure on the subdivision, of	r's office, there a or any part there	t according to the records in the are no liens against the property of, for unpaid State, county or ments certified to the County
	County T	reasurer	Date	
(vi)	corner of the Clerk and 1 20 Number	he cover sheet): Tl Recorder of San M	nis plat was acce liguel County, C Book Number Fime	e placed in the lower right-hand epted for filing in the office of the Colorado on this day of , A.D. , Page Number, Reception  Date
(vi	APPROV	Certificates. VED this_day of_ Commission of the		_, 20, by the Planning and ood, Colorado.
	Trustees			_, 20, by the Board of lo.
(vi	•	ry notice of vested	rights.	
				ng Commission for approval and

then to the Board of Trustees for approval. A public hearing shall be held by the Planning and Zoning Commission and the Board of Trustees on any application for a replat. The

Town Clerk shall cause notice of the public hearings of the Planning and Zoning Commission and the Board of Trustees to be given by one publication in a newspaper of general circulation in the Town of Norwood without the necessity of notifying property owners by mail. Such notice shall state the time and place of such hearing and the nature of the subject to be considered, which time shall not be earlier than fifteen (15) days from the date of publication.

- **(b) Plat amendments.** Notwithstanding any other provision of this section, the Board of Trustees is authorized to approve and issue an amending plat that is signed by the applicants only, and that is for solely one or more of the following purposes, and such approval and issuance shall not require notice, hearing, or approval of other lot owners.
  - (1) The purpose of the amendment is to correct an error in any course or distance shown on the prior plat.
  - (2) The purpose of the amendment is to add any course or distance that was omitted on the prior plat.
  - (3) The purpose of the amendment is to correct an error in the description of the real property shown on the prior plat.
  - (4) The purpose of the amendment is to indicate monuments set after death, disability, or retirement from practice of the engineer or charged with responsibilities for setting monuments.
  - (5) The purpose of the amendment is to show the proper location or character of any monument that has been changed in location or character or that originally was shown at the wrong location or incorrectly as to its character on the prior plat.
  - (6) The purpose of the amendment is to correct any other type of clerical error or omission as previously approved by the Planning and Zoning Commission and the Board of Trustees; such errors and omissions may include, but are not limited to, lot numbers, acreage, street names, and identification of adjacent recorded plats.
  - (7) The purpose of the amendment is to correct an error in courses and distances of lot lines between two (2) adjacent lots where both lot owners join in the application for plat amendment and neither lot is abolished, provided that such amendment does not attempt to remove recorded covenants or restrictions and does not have a material adverse effect on the property rights of the owners in the plat.
  - (8) The purpose of the amendment is to relocate a lot line in order to cure an inadvertent encroachment of a building or improvement on a lot line or on an easement.
  - (9) The purpose of the amendment is to relocate or vacate one or more lot lines between one or more adjacent lots where the owner or owners of all such lots join in the application for the plat amendment, provided that such amendment does not:

- a. Attempt to remove recorded covenants or restrictions;
- b. Increase the number of lots; or
- c. Result in the creation of a lot with less the minimum lot area in the required in underlying zone district (see Sec. 3.06).

#### § 6.10 Conditional Use Permits

- (a) General. A conditional use is a use that may or may not be appropriate in a given location depending upon the specific characteristics of the proposed use and\or the conditions imposed upon the approval of the use. Conditional use permits may be approved for the uses indicated in the use regulations of the zoning district of the property for which the conditional use permit is requested.
- **(b) Pre-application conference.** Prior to the filing of a conditional use permit application, the applicant shall meet with the Town Planner or his or her designated agent to acquaint himself or herself with the requirements of the Town. As such meeting, the application contents, referral agencies, review procedures, use and area standards, and the general character of the development may be discussed. At the pre-application conference, the applicant may be represented by a land planner, engineer or surveyor.
- (c) Submittal requirements. The applicant shall file ten (10) copies of an application requesting a conditional use permit and of a title certificate from a licensed title company or attorney listing the name of the property owner(s) and all liens, easements and judgments of record affecting the subject property. The application shall include signature of property owner. If the applicant is not the owner of the property, s/he shall also submit a letter from the property owner designating the applicant as his/her agent with the authority to represent the application. The application shall be accompanied by or show the following information:
  - (1) The street address and legal description of the property affected; and
  - (2) Site plan and description of proposed use and any and all plans, information, operating data and expert evaluation necessary to clearly explain the location, function and characteristics of any building or use proposed; and
  - (3) A report addressing the following standards:
    - a. Compatibility with surrounding land uses
    - b. Extent of noise, light, odor and activity and/or measures to mitigate intensity of
    - c. Circulation, traffic impact and parking
    - d. Number of customers/other (if applicable)
    - e. Ability to provide necessary utilities
    - f. Provisions for providing services and maintaining the site
    - g. Provisions for screening, lights and other items potentially causing off-site impacts
  - (4) A filing fee to cover the cost of review in accordance with the fee schedule adopted by resolution of the board of trustees.
- (d) Review by Planning and Zoning Commission. Before taking action on any proposed conditional use, the Board of Trustees shall submit the same to the Planning and Zoning Commission for its recommendation and report.

- (1) Public hearing required. The Planning and Zoning Commission shall hold a public hearing on any application for conditional use permit prior to making its recommendation to the Board of Trustees.
- (2) Notification requirements.
  - a. The applicant shall mail the written notice of public hearing before the Planning and Zoning Commission on the proposed amendment or change, after obtaining a copy of the notice from the Town Staff, to all owners of real property lying within one hundred feet (100') of the property on which the change is requested. Notice shall also be mailed to all owners of mineral estates of record in the records of the San Miguel county clerk and recorder. The notice shall be given not less than fifteen (15) days before the date set for hearing by depositing in the mail such notice properly addressed and postage paid to each such owner as the ownership appears on the last approved County assessor records. Applicants shall provide proof of mailing of notice to the Town prior to the public meeting or hearing that is the subject of the mailing of a notice.
  - b. The Town Clerk shall cause notice of the public hearing of the Planning and Zoning Commission to be given by one publication in a newspaper of general circulation in the Town of Norwood without the necessity of notifying property owners by mail. Such notice shall state the time and place of such hearing and the nature of the subject to be considered, which time shall not be earlier than fifteen (15) days from the date of publication.
- **(e) Action by Board of Trustees.** A public hearing shall be held by the governing body before approving a conditional use permit.

Notification requirements. The Town Clerk shall cause notice of the public hearing of the Board of Trustees to be given by publication in a newspaper of general circulation in the Town of Norwood without the necessity of notifying property owners by mail. Such notice shall state the time and place of such hearing and the nature of the subject to be considered, which time shall not be earlier than fifteen (15) days from the date of publication.

(f) Standards for Approval. The Board of Trustees shall use the following standards to evaluate an application for conditional use. The Board of Trustees may, in the interest of the public welfare and to assure compliance of this Ordinance, establish conditions of operation, location, arrangement and construction of any use for which a permit is authorized. In authorizing the location of any use listed as a Conditional Use Permit, the Board of Trustees may impose such development standards and safeguards as the conditions and location indicate important to the welfare and protection of adjacent property from noise, vibration, dust, dirt, smoke, fumes, gas, odor, explosion, glare, traffic circulation or other undesirable or hazardous conditions.

Standards for approval of conditional uses include:

(1) Sufficiency of water pressure and supply, availability of sewer service and other utilities as determined by the utility provider;

- (2) Adequacy of roads to accommodate proposed use and ability to ensure fire protection, snow removal and road maintenance;
- (3) Suitability of soils and subsurface geologic conditions to support proposed use;
- (4) The proposed use will not have a deleterious effect on air quality in the surrounding area and the town;
- (5) The design and location of any proposed structure, access or other characteristics of the use will not adversely affect surrounding land uses or the character of the town, nor cause harmful disturbance to critical natural features of the site or surrounding area.; and
- **(6)** Compatibility with surrounding area.
- (g) A file containing all documents relevant to the application and disposition of such conditional use permit shall be maintained by the Town Clerk.

#### § 6.11 Vested Property Rights

#### (a) General.

- (1) Pursuant to the provisions of Article 68 of Title 24, C.R.S., a property right shall be deemed vested with respect to any property, following notice and public hearing, when required, upon the approval or conditional approval, of a final plat by the Board of Trustees.
- (2) A vested property right shall attach to and run with the applicable property and shall confer upon the landowner the right to undertake and complete the development and use of said property under the terms and conditions of the approval.
- (3) The Board of Trustees may approve a subdivision plat or grant other final approval upon such terms and conditions as may reasonably be necessary to protect the public health, safety and welfare. Such conditional approval shall result in a vested property right, although failure to abide by such terms and conditions will result in a forfeiture of vested property rights.
- (b) Vested Property Right Term. A property right that has been vested as provided in subsection 6.11(a) shall remain vested for a period of three years. However, the Board of Trustees may enter into development agreements with land owners specifying that property rights shall be vested for a period exceeding three years when warranted in light of all relevant circumstances, including but not limited to the size and phasing of development, economic cycles and market conditions. Such development agreements shall be adopted as legislative acts subject to referendum.
  - (1) Issuance of a building permit or a development permit shall guarantee vested rights to use the property in compliance with the terms and conditions of final plat, although failure to comply with such terms and conditions shall result in forfeiture of vested property rights.
  - (2) Should no building permit or development permit be issued within said three years, the plan shall be terminated and the vested property right shall automatically expire.
- **(c) Extension of Vested Property Right Term.** The affected landowner may request that the Board of Trustees grant an extension of the final plat for up to three years, provided that:
  - (1) A written request for an extension is submitted by the affected landowner no less than sixty (60) days prior to the date of termination of the vested property right;
  - (2) Such extension request shall be considered by the Board in a public hearing, notice of which shall be advertised not less than 30 days prior to such hearing in a newspaper of general circulation within the County;
  - (3) There is no conflict with the Land Use Code or that any conflict may be corrected by an

- amendment to the final plat, which shall be presented with the request for extension;
- (4) The applicant has demonstrated that the final plat continues to be compatible with adjacent properties and the surrounding area, or that compatibility may be established by an amendment to the final plat, which shall be presented with the request for extension;
- (5) The applicant has demonstrated that the final plat is consistent with the Comprehensive Development Plan; and
- (6) Such extension, if granted, shall be valid only for the period approved by the Board of Trustees.
- (d) Further Reviews. Following approval or conditional approval of a subdivision plat or other final approval, nothing in this section shall exempt such a plan or plat from subsequent reviews and approvals, including, but not limited to, construction drawings, drainage plans, building permit and certificate of occupancy to ensure compliance with the terms and conditions of the original approval, provided that such reviews and approvals are not inconsistent with said original approval.
- **(e) New Regulations**. The establishment of a vested property right shall not preclude the application of land use regulations which are general in nature and are applicable to all property subject to land use regulation by the Town, including, but not limited to, building, fire, plumbing, electrical, mechanical codes, and other public health, safety and welfare codes.
- **(f) Natural or Man-made Hazards**. A vested property right shall automatically terminate upon the discovery on or in the immediate vicinity of the subject property of natural or man-made hazards which could not reasonably have been discovered at the time of site-specific development plan approval, and which, if uncorrected, would pose a serious threat to the public health, safety and welfare.
- **(g) Public Improvements.** The vested property rights provided herein shall in no way diminish or alter the requirement for public improvements, or other requirements, as provided in Town regulations.
- (h) Effective Date of Final Plat Approval. The effective date of the approval of a final plat shall be the date of approval or grant by the Board of Trustees. In the event amendments to a final plat are proposed and approved, the effective date of such amendments, for purposes of duration of a vested property right, shall be the date of approval or granting of the original final plat, unless the Board of Trustees finds to the contrary and incorporates such finding in its approval of the amendment.
- (i) Vested Rights Language. Each final plat shall contain the following language: "approval of this plan may create a vested property right pursuant to article 68 of Title 24, C.R.S., as amended." Failure to contain this statement shall invalidate the creation of the vested property right. In addition, a notice describing generally the type and intensity of use approved and the specific parcel or parcels of property affected and stating that a vested property right has been created shall be published once, not more than fourteen (14) days after approval of the final plat, in a newspaper of general circulation within the County.
- (j) Other Town Rules. Approval of a final plat shall not constitute an exemption from or waiver of

any other provisions of the Town's regulations pertaining to the development and use of property.

#### § 6.12 Annexations

#### (a) Authority.

The provisions of the *Colorado Municipal Annexation Act of 1965* ("Act"), as the same may from time to time be amended, shall govern the filing and processing of all petitions for annexation of land to the Town of Norwood. The provisions herein contained are intended to be complementary and in addition to the Act and not to be construed as altering, modifying, eliminating or replacing any requirements set forth in that Act, or any requirements set forth in other sections of this Land Use Code.

#### (b) Conditions on Annexation; Policies

The Town will impose terms and conditions on annexation to protect the public interest and to that end shall ensure that the following policies are met:

- (1) All annexations shall be consistent with the Norwood Comprehensive Plan and the Norwood Major Streets and Future Land Use Plan.
- (2) The Norwood Major Streets and Future Land Use plan identifies a Master Plan Boundary area surrounding the Town as an area appropriate for the future residential, commercial and industrial growth of the Town. Consent to annexation by petitioning landowners within the Master Plan Boundary and conformance with the Comprehensive Plan as amended by the Norwood Major Streets and Future Land Use Plan and the implementation zoning shall be a condition of extension of municipal utility service.
- (3) There will be no annexations of areas outside the boundaries of the Master Plan Boundary identified by the Norwood Major Streets and Future Land Use Plan, with the possible exception of annexation of any acquired open space.
- (4) The Town is very concerned about long term maintenance costs of annexed land. Petitioners and developers shall identify revenues adequate to pay the long term costs of maintenance of their development, and the Town shall ensure that revenues will be adequate prior to approval of a petition for annexation.

#### (c) Pre-submission Conference.

Prior to filing of an annexation submission, the prospective petitioner shall meet with the Town Planner or his or her designated agent. The purpose of the meeting is to allow the petitioner an opportunity to obtain information regarding the general annexation policies and requirements of the Town, the relationship of the proposed annexation to the comprehensive plan, and to prevent unnecessary expenditures for preparation of annexation submission which may not be consistent with Town Policies and requirements. At such meeting the annexation submission contents, referral agencies, review procedures, density standards, use and area standards, street requirements, utility service, long term maintenance costs and the general character of the land proposed to be annexed may be discussed. At the pre-submission conference, the petitioner may be represented by an attorney, land planner, engineer or

surveyor. A prospective petitioner may request a pre-submission conference with the Planning and Zoning Commission and/or Board of Trustees. All pre-submission conference discussions shall be non-binding.

#### (d) Annexation Submission Requirements

The petitioner shall supply one original and fifteen (15) copies of the annexation submission which shall include all of the following:

- (1) Petition for Annexation: A petition for annexation meeting the requirements of Section 31-12-107(1) (a), C.R.S.
- (2) Annexation Map. An annexation map drawn at a scale of one inch (1") equals twenty feet (20') with a north point, date and other pertinent data and meeting the requirements of Section 31-12-107(1)(d), C.R.S.
- (3) Vicinity Map. A vicinity map on a smaller scale showing the relationship of the area to be annexed and the existing town boundaries including existing and proposed land uses, residential densities, streets, watercourses, utilities, easements and any other major feature in and adjacent to the tract of land proposed for annexation, zoning of all adjacent properties, special districts within the area proposed to be annexed and contour intervals of not more than ten (10) feet.
- (4) Master Plan Report. A written report describing the proposed land use and requested zoning of the area proposed to be annexed, including:
  - a. A statement addressing consistency with the Town of Norwood Comprehensive Plan as amended by the Norwood Major Streets and Future Land Use Plan.
  - b. A copy of a draft annexation agreement, unless waived by the Town, which shall address all issues relating to the annexation and/or development of the property.
  - c. Estimates of the current population, assessed property value and costs of providing public services including but not limited to such services as fire protection, trash removal, law enforcement, school, utilities and street maintenance.
  - d. The name(s) of special district(s) providing services that would be affected by the annexation. An analysis of the cost of providing services to the area to be annexed. If the unincorporated area to be annexed is part of a special district or county service area whose responsibilities are to be assumed by the Town, a statement shall be included indicating what steps will be taken to ensure a smooth transition in service delivery.
  - e. A description of any and all water rights appurtenant to the area proposed for annexation.
  - f. A description including a time table demonstrating how petitioner will develop

and finance the extension and undergrounding, where necessary, of utilities and services including, but not limited to, water and sewer, electricity, gas, cable television, and telephone service.

- g. A statement and description of what land areas are to be dedicated for public use, or what equivalent cash will be paid, and what other types of public benefit will be provided within an agreed to period of time specifically addressing affordable housing, recreational trails and facilities and conveyance of water rights.
- h. A description of how the extension of municipal services, other than utilities, will be financed.
- i. A statement of zoning requested for the property, including a complete application for a zoning change.
- (5) Tax Payment Certification. Tax certificates showing payment of taxes then due on the property proposed for annexation.
- (6) Annexation Impact Report. An annexation impact report shall be submitted for all proposed annexations unless waived, in whole or in part, by the Board of Trustees. The Town may, in its discretion, prepare the report or arrange for the preparation of the report by a party of its choosing. If the Town prepares or arranges for the preparation of the report by a third party, the petitioner shall reimburse the Town for the cost of the report. The report shall meet all the informational and submission requirements of Section 31-12-108.5, C.R.S. as well as include:
  - a. General information including gross acreage of the area proposed for annexation, approximate number and type of units, acreage of streets and parking, acreage and percentage of open space to be created (analyzed as to the amount to be deeded or dedicated to the Town, the amount to be retained in public ownership and the estimated Town maintenance costs), density ratio (acres to be developed with lots and unit compared to acres in streets and open apace) and a statement of effect upon the R-2 School district including estimated number of students generated and capital construction required to serve.
  - b. Traffic impacts including projected vehicle trips to enter or depart the site (calculated on a peak and daily basis).
  - c. Utilities impacts including projected demand for municipal water (year-round average and seasonal peak demand and consumptive use), impact on existing water and sewer capacity and the potential need for additional facilities, and potential solid waste disposal costs.
  - d. Environmental analysis including soil types and bearing capacities, geologic hazard areas, high groundwater tables, potential erosion problems, flood-prone areas,

effects on wildlife and vegetation, aesthetic considerations and wetland designation.

(7) Filing Fee. A filing fee shall be submitted to cover the cost of review and processing the petition in accordance with a fee schedule adopted by resolution of the Board of Trustees.

#### (e) Development Applications

Simultaneously with the filing of the annexation submission, the petitioner may file additional development applications for the property proposed for annexation. Any development applications and all proceedings taken thereon shall conform to the requirements of this Land Use Code and shall be in addition to the requirements of this Section 6.12. The processing of the annexation submission, rezoning and other development applications may proceed concurrently, but applications and approvals must be sequenced and processed according to the requirements of the Norwood Land Use Code and the Act. See also Section 4.01(c)(3).

#### (f) Referrals; Staff Evaluation

Upon receipt of a complete annexation submission and payment of the filing fee, the original of the submittal shall be retained by the Town Clerk and copies shall be distributed to:

- (1) The Town Planner/Consultant (one copy)
- (2) Town Engineer/consultant (one copy)
- (3) Town Public Works Director (one copy)
- (4) Town Attorney (one copy)
- (5) San Miguel Power Association or other power company (one copy)
- (6) Natural Gas company (one copy)
- (7) Telephone company (one copy)
- (8) R-2 School District (one copy)
- (9) Norwood Water Commission (one copy)
- (10) Norwood Sanitation District (one copy)
- (11) Norwood Town Clerk (one copy for public record)
- (12) Norwood Fire District (one copy)
- (13) US Postal Service (one copy)
- (14) Any other agencies as determined appropriate by the Town Planner

No later than ten (10) days before the meeting of the Planning and Zoning Commission as provided for in subsection (h) each person or agency listed herein shall submit its comments and/or recommendation on the proposed annexation.

The Town Planner shall review the annexation submission; comments received from referral agencies, the Standards for Annexation, and all other relevant information and shall provide a written report of his or her evaluation and recommendations to the Planning and Zoning Commission prior to the meeting provided for in subsection (h).

#### (g) Reimbursement of Costs

The petitioner shall reimburse all of the Town's actual costs of reviewing and processing any annexation request. At any time during the review process, the Town may employ professional consultants to assist the Planning Commission and Board of Trustees in their review of the annexation request. As used in this section, "actual costs" shall include all direct and indirect costs incurred by the Town to review and process an annexation submission, the costs of consultants, publication, mailing, recording, engineering, legal, town personnel expenses, supplied, material and equipment. In addition, the petitioner shall assume the cost of updating the Town boundary map upon the successful completion of an annexation proceeding.

#### (h) Planning Commission Review and Recommendation

The Planning and Zoning Commission shall consider the annexation submission at a regular or special meeting and prepare a written recommendation on issues related to annexation to the Board of Trustees prior to the hearing established pursuant to Section 31-12-108, C.R.S. In making its recommendation on the annexation, the Planning Commission shall consider all relevant information including, but not limited to, the annexation submission, the Town Planner Report, referral comments, public comments and the Standards for Annexation established in subsection (i).

#### (i) Standards for Annexation

It shall be the policy of the Town with respect to annexation and the consideration of annexation submissions that all the following criteria have been met before annexation of any lands:

- (1) Use. That the master plan for the area to be annexed is consistent with the Norwood Comprehensive Plan and the Norwood Major Streets and Future Land Use Plan, is in harmony with the intent of Town zoning and policies of the Town, and compatible with adjacent neighborhoods. The annexation of lots that have less than the minimum area required by the respective zone district that implement the Comprehensive Plan will be nonconforming and shall be subject to the provisions of subsection 1.06 (b).
- Open Space. That the open spaces have a workable program established for maintenance and up-keep.
- (3) Necessity. That the proposed annexation is necessary or desirable and will contribute to the general well being of the community.
- (4) Health, safety and general welfare. That the proposed annexation will in no way be detrimental to the health, safety, or general welfare of person residing within the corporate boundaries or injurious to property or improvements in the vicinity of the annexed area.
- (5) Logical Road System. That the area has incorporated in its design, if a design has been developed, a logical extension of roads.
- (6) Utilities and Roads. That the extension of services is feasible and will be financed totally

- by the petitioner; and that the petitioners will post performance guarantees to assure the completion of public improvements.
- (7) Water Rights. That the petitioner has or will convey all right, title and interest in any water rights associated with the property or in any and all water located beneath the property to be annexed if the annexed area is to be served by the municipal water system or water is needed for the maintenance of open space or parks.
- (8) Revenues. That the revenues and/or public benefit to be gained from the Town's portion of the increased tax base are equal to or greater than the cost of services to be provided.
- (9) Public Lands and Trails. That a minimum of eight percent (8%) of the gross land area approved for annexation is dedicated to the Town in fee simple, or equivalent cash or consideration given; and that trails shall be dedicated and constructed consistent with an adopted trails plan. Land area dedicated for trails may be credited toward the land dedication required herein.

#### § 6.13 Board of Adjustments

- (a) Creation and organization. There is hereby created a board known as the Board of Adjustment, which shall be organized as follows.
  - (1) The Board of Adjustment shall consist of five (5) regular members, including one member of the Planning and Zoning Commission or Board of Trustees, appointed by the Mayor, subject to ratification by a majority of the Board of Trustees.
  - In addition to the five (5) regular members of the Board of Adjustment, two (2) alternate members of the Board of Adjustment, who shall serve in the absence of one (1) or more regular members when requested to do so by the Chairperson of the Board or Mayor, may be appointed by the Mayor, subject to ratification by majority of the Board of Trustees, so that all cases heard by the Board of Adjustment will always be heard by a minimum of four (4) members.
  - (3) Each member of the Board shall be a resident citizen and qualified voter of the Town of Norwood, Colorado, at the time of his or her appointment. A member that ceases to reside in the Town during his or her term of office shall immediately forfeit their office.
  - (4) The term of office of members of the Board shall be for two (2) years. Two (2) members shall be appointed for two (2) year terms beginning September 30, in even years and three (3) members shall be appointed for two (2) year terms beginning September 30, in odd years. Alternates shall serve the same term of office as regular members; one (1) member shall be appointed beginning September 30, in even years and one (1) member beginning September 30, in odd years. Regular and alternate members may be removed before their term of office expires.
    - (5) Board members shall serve without compensation.

#### (b) Officers and procedures.

- (1) The Mayor shall designate, and the Board of Trustees confirm, a chairperson as well as a vice chairperson from the regular members of the Board each year. The designation and confirmation shall be made no less than fifteen (15) days before the terms of the current members expire.
- (2) The chairperson shall preside over meetings. In the event a question over procedures arise, Robert's Rules of Order shall apply.
- (3) The vice chairperson shall assist the chairperson in directing the affairs of the Board of Adjustment. In the absence of the chairperson the vice-chairperson shall assume the duties of the chairperson.

- (4) The recording of the minutes of the Board of Adjustment meetings shall be the responsibility of the staff.
- (5) The Board of Adjustment may adopt rules to govern its proceedings and conduct of the business before the Board provided, however, that such rules are not inconsistent with this Title or Statutes of the State of Colorado.
- (6) Meetings of the Board shall be held at the call of the chairperson, and at such other times as the Board may determine. Such chairperson or, in his or her absence, the vice chairperson shall administer oaths and compel attendance of witnesses. All meetings of the Board shall be open to the public. Minutes of its proceedings shall be kept by the staff showing the vote of each member upon such question or, if absent or failing to vote, indicating such fact, and shall keep records of its examinations and other official actions, all of which shall be filed in the office of the Board and shall be a public record.
- (7) The concurring vote of four (4) members of the Board shall be necessary to reverse any order, requirement, decision or determination of the Town Staff, or to decide in favor of the applicant on any matter upon which the Board is required to act under this chapter or to grant any variance authorized by this Title.
- (8) The filing fee for every petition to be submitted to the Board of Adjustment shall be calculated to cover the cost of legal publications, accumulation of data, administrative cost and all other costs incurred by the Town in the review and processing of the petition or as provided for in the fee schedule adopted by resolution of the Board of Trustees.
- **(c) Jurisdiction.** The Board of Adjustment shall have the following powers, and shall have the power to impose reasonable conditions to insure compliance and protect adjacent property:
  - (1) To hear and decide appeals where it is alleged there is error in any order, requirement, decision or determination made by an administrative official of the Town in the enforcement of this Title, pursuant to Sec. 6.14, Appeals.
  - (2) To permit variance or modifications of the height, yard setback, area, coverage and parking regulations, pursuant to Sec. 6.15, Variances.

#### § 6.14 Appeals

- (a) Authority of Board. In exercising its powers, the Board of Adjustment may, in conformity with the provisions of the Statutes of the State of Colorado as existing or hereafter amended, reverse or affirm, wholly or partly, or may modify the order, requirements, decision, or determination appealed from and make such order, requirement, decision or determination in the Board's opinion, as ought to be made and shall have all the powers of the officer from whom the appeal is taken.
- (b) Application for appeal. Appeals to the Board of Adjustment can be taken by any person aggrieved or by any officer, department or board of the municipality affected by the decision of the Building Official. Such appeal shall be taken within thirty (30) days after the decision has been rendered by the administrative officer, by filing with the officer from whom the appeal is taken and with the Board of Adjustment, a notice of appeal specifying the grounds thereof. The officer from whom the appeal is taken shall forthwith transmit to the Board of Adjustment all the papers constituting the record upon which the action appealed from was taken.
- **(c) Filing fee.** A filing fee shall be submitted to cover the cost of review and processing with every appeal in accordance with the fee schedule adopted by resolution of the Board of Trustees.
- (d) Stay of proceedings. An appeal shall stay all proceedings of the action appealed from unless the officer from whom the appeal is taken certifies to the Board of Adjustment after the notice of appeal shall have been filed with him or her that by reason of facts stated in the certificate, a stay would, in his or her opinion, cause imminent peril to life or property. In such case, proceedings shall not be stayed, otherwise than by a restraining order that may be granted by the Board of Adjustment or by a court of record on application on notice to the officer from whom the appeal is taken and on due cause shown.
- **Hearing and notice.** The Board of Adjustment shall fix a reasonable time for the hearing of an appeal, give the public notice as follows:
  - (1) The applicant shall mail the written notice of public hearing before the Board of Adjustment on the appeal, after obtaining a copy of the notice from the Town Staff, to all owners of real property lying within one hundred feet (100') of the property on which the appeal is made. Notice shall also be mailed to all owners of mineral estates of record in the records of the San Miguel county clerk and recorder. The notice shall be given not less than fifteen (15) days before the date set for hearing by depositing in the mail such notice properly addressed and postage paid to each such owner as the ownership appears on the last approved County assessor records. Applicants shall provide proof of mailing of notice to the Town prior to the public meeting or hearing that is the subject of the notice.
  - (2) The Town Clerk shall cause notice of the public hearing of the Board of Adjustment to be given by one publication in a newspaper of general circulation in the Town of Norwood without the necessity of notifying property owners by mail. Such notice shall state the time and place of such hearing and the nature of the subject to be considered, which time shall not be earlier than fifteen (15) days from the date of publication.

(f) Appeals to court. Any person or persons, jointly or severally, aggrieved by any decision of the Board, or any taxpayer or any officer, department or board of the municipality may present to a court of record a complaint, duly verified, setting forth that such decision is illegal, in whole or in part, specifying the grounds of the illegality. Such petition shall be presented to the court within thirty (30) days after the filing of the decision in the office of the Board and not thereafter.

#### § 6.15 Variances

- (a) Authority of Board. The Board of Adjustment shall be authorized to permit such variance or modifications of the height, yard setback, area, coverage and parking regulations as may be necessary to secure appropriate development of a parcel of land that differs from other parcels in the district by being of such restricted area, shape or slope that it cannot be appropriately developed without such modification.
- **(b) Submittal requirements.** The applicant shall file three (3) copies of an application requesting a variance. The application shall include signature of property owner. If the applicant is not the owner of the property, s/he shall also submit a letter from the property owner designating the applicant as his/her agent with the authority to represent the application. The application shall be accompanied by or show the following information:
  - (1) The street address and legal description of the property affected;
  - (2) A site plan and any and all other information necessary to clearly demonstrate eligibility for the requested variance based upon the required findings in subsection 6.15(c), below; and
  - (3) A filing fee to cover the cost of review in accordance with the fee schedule adopted by resolution of the Board of Adjustment.
- **(c) Required findings.** In exercising its power to grant a variance in accordance with this Title, the Board of Adjustment shall make finding and show in its minutes that:
  - (1) There are special circumstances existing on the property on which the application is made related to size, shape, area, topography, surrounding conditions and location that do not apply generally to other property in the same area and the same zoning district;
  - (2) That a variance is necessary to permit the applicant the same rights in the use of this property that are presently enjoyed under this Title, by other properties in the vicinity and zone, but which rights are denied to the property on which the application is made;
  - (3) That the granting of the variance on the specific property will not adversely affect the land use pattern as outlined by the Land Use Plan and will not adversely affect any other feature of the Comprehensive Plan of the Town of Norwood;
  - (4) That the variance, if granted, will be no material detriment to the public welfare or injury to the use, enjoyment or value of property in the vicinity;

- (5) That such unnecessary hardship has not been created by the applicant; and
- (6) That the proposed use is a Permitted Use in the underlying zone district.

#### § 6.16 Special Exceptions

The Planning and Zoning Commission shall have the power to hear and decide the following special exceptions to the terms of this Title.

- (a) Reconstruction of non-conforming use. The Planning and Zoning Commission shall be authorized to permit the reconstruction of a building occupied by a nonconforming use, or permit resumption of a nonconforming use on the lot or tract occupied by such building, or the reconstruction of a structure destroyed by fire or the elements on the tract occupied by such building, provided an application is submitted to the town clerk within six months of the date the structure was destroyed. All new construction shall conform to the applicable International building code, current fire code and any other applicable safety and construction codes which are in effect at the time the rebuilding is proposed.
- **(b) Discontinuance of nonconforming use.** The Planning and Zoning Commission shall be authorized to require the discontinuance of nonconforming uses of land or structures under any plan whereby the full value of the structure and facilities can be amortized within a definite period of time, taking into consideration the general character of the neighborhood and the necessity of all property to conform to the regulations of this Title. All actions to discontinue a nonconforming use of land or structure shall be taken with due regard to the property rights of the persons affected when considered in the light of the public welfare and the character of the area surrounding the designated nonconforming use and the conservation and preservation of the property.
- (c) Status of nonconforming uses. The Planning and Zoning Commission shall, from time to time, on its own motion or upon cause presented by interested property owners or on request of the Board of Trustees or Planning and Zoning Commission inquire into the existence, continuation or maintenance of any nonconforming use within the Town.

#### § 6.17 Temporary Use Permits

Temporary use permits shall be issued by the Planning and Zoning Commission, subject to the following provisions.

- (a) **Zoning.** The use for which the permit is requested shall be authorized as a temporary use in the district in which the use is to be located.
- **Conditions.** The applicant shall meet all conditions for such temporary use permit set forth in this Title.
- **(c) Time limit.** A time limit for the discontinuance of the temporary use shall be specified on the temporary use permit.

Title 7: Land Use Code Section 6.18, Sign Permits

#### § 6.18 Sign Permits

(a) **Permits.** It shall be unlawful to erect, construct, reconstruct, alter, paint, or repaint, or change the use of any sign as defined in this section without first obtaining a sign permit; however, a sign permit shall not be required to repaint a sign exactly as it was permitted for the purpose of maintenance.

- (1) For all sign permits required, a fee of one dollar per square foot of surface area shall be charged to cover the costs of administration.
- (2) Off-premise advertising signs require a permit from the Town.
- (b) Sign permit application. Application for a sign permit, where such permit is required by Sec. 5.06, Signs, shall be made upon forms provided by the Town of Norwood and shall include the following information:
  - (1) A drawing to scale of the proposed sign.
  - (2) A drawing to scale of the site plan or building facade showing the proposed location of the sign.
  - (3) Name, address and telephone number of the applicant.
  - (4) Name, address and telephone number of the owner.
  - (5) Name, address and telephone number of the person or firm responsible for the erection of the sign.
  - (6) Location of the building, structure or tract to which, or upon which, the sign is to be attached or erected.
- (c) Sign permit fees and procedures. Every sign permit application shall be accompanied by a permit application fee payable to the Town of Norwood. The application fee for pennants, banners and posters that are erected pursuant to subsection 5.06(c) a fee of \$5 dollars shall be charged. For all other sigh permits required, a fee of one dollar per square foot of surface area shall be charged to cover the costs of administration.

Exception. No application or permit fee shall be required for special event or public announcement signs for bona fide charitable and other non-profit organizations.

#### § 6.19 Zoning Development Permits

No building permit may be issued and no person(s) may engage in any development within the incorporated area of the Town of Norwood without obtaining a development permit (Appendix "A" to this Code). Failure to comply with any condition(s) of approval, as determined by the Board of Trustees, shall result in inability to obtain any rights granted conditionally thereunder and Town revocation of the development permit. Upon thirty days' notice by Developer, the Town shall set a hearing for determination of non-compliance with conditions.

#### § 6.20 Certificates of Occupancy

- (a) Applicability. No building hereafter erected, converted or structurally altered shall be used or occupied and no land or nonresidential building may be changed in use unless or until a Certificate of Occupancy shall have been issued by the Building Official of the Town of Norwood stating that the building or proposed use of land or building complies with the provisions of this Title and other building and health laws of the Town of Norwood.
- **(b) Application.** A Certificate of Occupancy shall be applied for coincident with the application for a Building Permit and will be issued within ten (10) days after the completion of the erection, alteration or conversion of such building or land provided such construction or change has been made in complete conformity to the provisions of this Title.
- **(c) Record.** A record of all certificates shall be kept on file in the office of the Building Official, and copies shall be furnished, on request, to any person having a proprietary or tenancy interest in the land or building affected.

#### § 6.21 Administrative Officials

- **Town Planner.** There is hereby created a official administrative position know as the Town Planner. The Town Planner shall be the Director of the Town of Norwood Planning Department and shall be appointed by, directed by, and serve at the pleasure of the Board of Trustees. The Town Planner shall have the following jurisdiction, authority and duties:
  - (1) Administer the Land Use Code. The Town Planner administers the Land Use Code and has those powers and duties expressly assigned to him/her under this Code and otherwise delegated or assigned to him/her in accordance with Town policy.
  - (2) Issue Zoning Development Permits. The Town Planner shall issue zoning development permits as required by Sec. 6.19 prior to the issuance of any building permits for construction activities pursuant to the procedures in the Code and deny development permits that are inconsistent with the Code.
  - (3) **Enforce the Land Use Code.** The Town Planner shall enforce the provisions of this Title.
  - (4) Interpret the Land Use Code. The Town Planner shall have the authority to make all

interpretations of the Land Use Code and the boundaries of the Official Zoning Map.

- **(b) Building Official.** There is hereby created an official administrative position know as the Building Official. The Building Official shall be the Director of the Town of Norwood Building Department and shall be appointed by, directed by, and serve at the pleasure of the Board of Trustees. In addition to the jurisdiction, authority and duties which may be conferred on the Building Official by the International Building Code, the Building Official shall have the following jurisdiction, authority and duties:
  - (1) **Issue Building Permits**. The Building Official shall issue applicable permits to allow construction activities which have received a zoning development permit pursuant to the procedures in the Code and deny building permits for activities which have not received a zoning development permit.
  - (2) Interpret and enforce the International Building Code. The Building Official shall interpret and enforce the applicable International Building Code.
  - (3) Enforce the Land Use Code. The Building Official shall enforce the provisions of this Title.

# Chapter 7.00 RESIDENTIAL USE, GROWING, CULTIVATING AND PROCESSING OF MARIJUANA

#### **§7.01:** FINDINGS AND PURPOSE:

- A. The growing and processing of marijuana plants in a residential setting can affect the health, safety, and welfare of both the occupants of the residential structure within which the marijuana is grown or processed, and other persons.
- B. Unregulated residential growing or processing of marijuana may result in noncompliance with the town's building and other technical codes, has the potential to result in serious problems including, without limitation, traffic congestion, water pollution, light spillage, causing a fire to ignite, burn and emanate from the residential structure within which the marijuana is grown or processed, and to cause other public nuisances, all that would affect the health, safety, and welfare of both the occupants of the residential structure within which the marijuana is grown or processed, and other persons.
- C. Therefore, it is the purpose of this Chapter to require that persons using, growing, cultivating or processing marijuana in a residential setting within the Town pursuant to sections 14 or 16 of Article XVIII of the Colorado Constitution do so in a safe manner that does not endanger the public health, safety, and welfare, or create a public nuisance. In accordance with Colorado legislature house bill H.B.17-1220, effective January 1, 2018, the Town is setting plant count limits for both Medical and Personal Use/Recreational Marijuana cultivation to reduce the amount of legally grown marijuana that is being diverted to the illegal market. (Ord. 0510 Series 2018)

#### §7.02: **DEFINITIONS:**

- A. The definitions contained in the Medical Marijuana Code and in Article XVIII, § 14 and Article XVIII, § 16 of the Colorado Constitution are incorporated into this Chapter by reference.
- B. As used in this Chapter the following words, whether capitalized or not, have the following meanings, unless the context clearly requires otherwise:

ACCESSORY STRUCTURE: shall have the same meaning as set forth in LUC Chapter 2, section 2.02, except that for purposes of this Chapter 7, the definition of Accessory Structure shall be limited further to a Structure that (i) is capable of being and is an Enclosed and Locked Space, or is capable of including and does so include a separate Enclosed and Locked Space, (ii) fully complies with this Chapter 7, including without limitation subsection 7.03(K), is constructed in such a manner and with such materials that it is impenetrable and consistent with the definition of an Enclosed and Locked Space; and (iii) otherwise complies with the LUC. Further, for purposes of this Chapter, a solid, secured "greenhouse" structure constructed to the design requirements for imposed loads (e.g. wind, snow, seismic activity) as required by the International Building Code with exterior wall and roof coverings constructed with materials such as concrete, glass, metal, wood, polycarbonates or other such material that is tested and approved for such use, and that otherwise complies with the provisions of this Chapter, is considered an Accessory Structure. (Ord. 0510 Series 2018)

AUTHORIZED PERSON: means an individual human being that is twenty-one (21) years of age or older, and is either the Owner of the Premises or has obtained prior written permission from the Owner of the Premises to possess, use, grow, cultivate, or process marijuana within a Residential Structure on the Premises.

ENCLOSED AND LOCKED SPACE: means the Residential Structure or the area within the Residential Structure, as the case may be in compliance with this Chapter 7, where marijuana is grown, cultivated or processed pursuant to Article XVIII, §14 and Article XVIII, §16 of the Colorado constitution, that is covered and surrounded on all sides; secured at all points of ingress or egress with a locking mechanism such as a key or combination lock designed to limit access. Temporary opening of windows and doors or the temporary removal of a wall or ceiling panels does not convert the area into an unenclosed space. A hoop house, high tunnel or other similar structure that is covered or uses a membrane or a soft pliable sheet, i.e. plastic sheeting, visqueen, tarps, canvas, polyethylene films or similar materials is NOT an Enclosed and Locked Space. (Ord. 0510 Series 2018)

LOT: shall have the same meaning as set forth in LUC, Chapter 2, section 2.02.

MARIJUANA: shall have the same meaning as in section 16(2)(f) of article XVIII of the Colorado constitution.

MARSHAL: means the Town Marshal, the Town Marshal's designee, or other law enforcement officer including without limitation the San Miguel County Sheriff, undersheriff or deputy.

MATURE MARIJUANA PLANT: a "mature" marijuana plant means a marijuana plant other than an "immature" marijuana plant. For purposes of this ordinance, an "immature" marijuana plant means a nonflowering marijuana plant that is no taller than eight inches and no wider than eight inches produced from a cutting, clipping, or seedling and that is in a growing container that is no larger than two inches wide and two inches tall that is sealed on the sides and bottom. This definition is consistent with C.R.S. § 12-43.3-104(1.5). For purposes of this Chapter, a "flowering" marijuana plant would be a cannabis plant in which there are physical signs of flower budding out of the nodes in the stem. This definition is consistent with C.R.S. § 18-18-406((3)(c)(I). (Ord. 0510 Series 2018)

MISDEMEANOR: shall have the same meaning as set forth in subsection 1-3-2 of the Norwood Town Code.

NUISANCE: shall have the same meaning as set forth in subsection 1-3-2 of the Norwood Town Code.

OCCUPANT: shall have the same meaning as set forth in subsection 1-3-2 of the Norwood Town Code.

OFFENSE: shall have the same meaning as set forth in subsection 1-3-2 of the Norwood Town Code.

OPENLY: means the area within the Residential Structure where the marijuana is used, grown, cultivated or processed is not protected from unaided observation lawfully made from outside the perimeter of the Premises or the Residential Structure not involving physical intrusion into the Premises or Residential Structure.

OTHER CANNABINOIDS: means those parts of the plant of the genus cannabis whether growing or not, the seeds thereof, the resin extracted from any part of the plant, and every compound, manufacture, salt, derivative, mixture, or preparation of the plant, its seeds, or its resin, including marijuana concentrate.

OWNER: shall have the same meaning as set forth in LUC Chapter 2, section 2.02.

PARCEL: shall have the same meaning as set forth in LUC Chapter 2, section 2.02.

PATIENT: shall have the same meaning as in section 14(1)(d) in conjunction with section 14(1)(a) of Article XVIII of the Colorado Constitution. (Ord. 0510 Series 2018)

PRIMARY CAREGIVER: shall have the same meaning as in section 14(1)(f) of Article XVIII of the Colorado Constitution. Primary Caregivers who cultivate medical marijuana for their patients must register with CDPHE and the DOR-MED. A Primary Caregiver who cultivates medical marijuana for his or her patients or transports medical marijuana for his or her patients shall also register with the state licensing authority and comply with all local laws, regulations, and zoning and use restrictions. (Ord. 0510 Series 2018)

PUBLIC RIGHT-OF-WAY: shall have the same meaning as set forth in LUC Chapter 2, section 2.02.

PREMISES: The Lot and all Structures included thereon including without limitation the Residential Dwelling, Accessory Structure(s) and other Residential Structures. If there are Multiple Residential Dwellings on the Premises each is entitled to the plant count restrictions of §7.03.C. (Ord. 0510 Series 2018)

PUBLICLY: The Residential Structure or the area within the Residential Structure, as the case may be in compliance with this Chapter 7, where the marijuana is used, grown, cultivated or processed is open to general access without restriction.

REAL PROPERTY: shall have the same meaning as set forth in LUC Chapter 2, section 2.02.

RESIDENTIAL DWELLING: A lawful structure that is designed and occupied as living quarters and includes facilities for cooking, sleeping and sanitation. A Residential Dwelling may include a Mobile Home Dwelling, a Dwelling unit, Duplex dwelling, Single-family dwelling, Multi-family dwelling, Cottage Dwelling Unit, or residential Condominium, as such terms are defined in LUC Chapter 2, section 2.02, or otherwise described or regulated in this Chapter 7. Further, for purposes of this Chapter 7, the definition of Residential Dwelling shall be limited further to a Structure that (i) is capable of being and is an Enclosed and Locked Space, or is capable of including and does so include a separate Enclosed and Locked Space, (ii) fully complies with this Chapter 7, including without limitation subsection 7.03(K), and (iii) otherwise complies with the LUC.

RESIDENTIAL STRUCTURE: The Residential Dwelling or Accessory Structure(s) to the Residential Dwelling, as further defined herein, located on the Lot that are part of the Premises.

STREET: shall have the same meaning as set forth in LUC Chapter 2, section 2.02.

STRUCTURE: That which is lawfully built or constructed, an edifice or building or any kind or any piece of work artificially built up or composed of parts joined together in some definitive manner. For purposes of this Chapter 7, the term Structure shall not include: signs, fences, or paved surfaces. Further, for purposes of this Chapter 7, not all Structures are suitable for the use, growing, cultivating and/or processing of marijuana, as only Residential Dwellings and Accessory Structure(s) to Residential Dwellings, as further defined in this Chapter 7, shall be deemed suitable for such purposes.

THC: Tetrahydrocannabinol.

TOWN: shall have the same meaning as set forth in subsection 1-3-2 of the Norwood Town Code.

# §7.03 REGULATIONS FOR THE GROWING, CULTIVATING, AND PROCESSING OF MARIJUANA IN A RESIDENTIAL STRUCTURE:

Marijuana shall not be possessed, used, grown, cultivated, processed, or transported in or around any Residential Structure within the Town except in compliance with the following regulations:

- A. The possession, use, growing, cultivating and processing of marijuana within a Residential Structure shall be done in full compliance with:
  - (1) this Chapter 7; and
  - (2) all applicable provisions of Article XVIII, § 14; or
- (3) all applicable provisions of Article XVIII, § 16 of the Colorado constitution, the Medical Marijuana Code, and the Medical Marijuana Program.
- B. Marijuana may be possessed, used, grown, cultivated, or processed within a Residential Structure only by a Primary Care-Giver, a Patient, or Authorized Persons.
- C. Notwithstanding and in addition to all other provisions of this Chapter 7, the following restrictions apply:
- (1) Regardless of the number of Authorized Persons, Patients or Primary Caregivers on the Premises, at any one time, it shall be unlawful to possess, grow, cultivate, or process more than twelve (12) marijuana plants within a Residential Structure on the Premises, with six (6) or fewer being mature flowering plants. (Ord. 0510 Series 2018)
- (2) Authorized Persons may not (i) possess, use or transport more than one ounce of marijuana off the Premises.
- (3) One Patient or one Primary Caregiver may not (i) possess, use or transport more than two ounces of marijuana off the Premises.
- (4) Any possession, use, transportation, growing, cultivating, or processing of medical marijuana by a Patient or Primary Caregiver shall be for medical use only.
  - (5) Patients or Primary Caregivers possessing, using, transporting, growing, cultivating or

processing medical marijuana shall possess a registration identification card obtained pursuant to the provisions of Article XVIII, § 14 of the Colorado Constitution that reflects the same address as the subject Premises. . (Ord. 0510 Series 2018)

- D. Authorized Persons who own the marijuana plants described in subsection 7.03 C may lawfully possess the marijuana produced by the marijuana plants described in 7.03 C within a Residential Structure on the Premises where the plants were grown and otherwise in full compliance with Article XVIII, § 16 of the Colorado Constitution.
- E. A Primary Care Giver or a Patient who owns the medical marijuana plants described in 7.03 C may lawfully possess the marijuana produced by the medical marijuana plants described in 7.03 C for medical purposes only and in full compliance with Article XVIII, § 14 of the Colorado Constitution, the Medical Marijuana Code and the Medical Marijuana Program.
- F. None of the marijuana plants or the marijuana described in this Chapter 7, including without limitation within subsection C or D of this section may be sold or offered for sale.
- G. Marijuana may not be possessed, transported, used, grown, cultivated, or processed Openly or Publicly.
- H. Marijuana may not be grown, cultivated, or processed in any area that is located outside of the exterior walls of a Residential Structure.
- I. All marijuana plants shall be grown, cultivated or processed in an Enclosed and Locked Space.
- J. If the Residential Structure within which the marijuana is possessed, grown, cultivated or processed is larger than the respective square footage limitations set forth in subsection 7.03(K)(1) of this Chapter 7, then such growing, cultivation or processing must be done within a separate Enclosed and Locked Space within such Residential Structure.
- K. The possession, growing, cultivation or processing of marijuana plants shall be limited to the following areas within a Residential Structure:
- (1) Within any Residential Structure only within a defined and contiguous area not exceeding one hundred (150) square foot; and
- (2) Marijuana shall not be used, grown or processed within the common area of any Condominium, whether attached or detached.
- L. The possession, use, growing, cultivation, processing and transportation of marijuana shall not be perceptible from the exterior of the Residential Structure where the marijuana is possessed, used, grown, cultivated, processed or transported, including, but not limited to:
  - (1) Common visual observation;
  - (2) Light pollution, glare, or brightness that disturbs the repose of another reasonable person;

- (3) Undue vehicular or foot traffic, including unusually heavy parking at or near the Premises;
- (4) Noise from an exhaust fan that disturbs a reasonable person; and
- (5) Smell or odor of marijuana growing or being cultivated or processed within a Residential Structure shall not be capable of being detected by a reasonable person with a normal sense of smell from any adjoining Lot or Parcel not part of the Premises, or from any adjoining Public Right-of-way or Street.
- M. The Residential Structure within which the marijuana is grown, cultivated, or processed shall meet all applicable requirements of the Town's building and technical codes.
- N. If an Authorized Person, Patient, or Primary Care Giver grows, cultivates, or processes marijuana within a Residential Structure located on a Premises that he or she does not own, he or she shall obtain the written consent of the Real Property Owner before commencing to grow, cultivate or process marijuana in the Residential Structure.
- O. No chemical substance that causes transformation in composition, structure and properties, specifically including those prohibited under the Colorado Retail Marijuana Code, 1 CCR 212-2, Series R 504.D, shall be used to enhance or extract THC or other cannabinoids from marijuana that is grown in a Residential Structure.
- P. Pursuant to Article XVIII, § 16, and C.R.S. § 9-7-113, the use of a compressed, flammable gas as a solvent in the extraction of THC or other cannabinoids, as such terms are defined herein, including, but not limited to the production of "Hash Oil" concentrate from the marijuana plant, is prohibited anywhere in within the Town including in any Residential Structure or on any other portion of the Premises.
- Q. Waste Disposal. Marijuana waste must be stored, secured, locked, and managed. Marijuana waste must be made unusable and unrecognizable prior to leaving the Residential Structure by grinding and incorporating the marijuana waste with non-consumable, solid wastes such that the resulting mixture is at least 50 percent non-marijuana waste including paper waste, plastic waste, cardboard waste, food waste, soil. After Marijuana waste is made unrecognizable as described above, it only shall be deposited in a secure waste receptacle owned by the Authorized Person, with final disposal at a solid waste site or facility. If composted, Marijuana waste shall never exceed the 50 percent non-marijuana ratio and may only be composted using Bokash or other compost activators.
- R. The growing, cultivation, or processing of marijuana by Authorized Persons, Patients, or Primary Caregivers for personal use and/or recreational or medical purposes is prohibited outside of an Enclosed and Locked Space within a Residential Structure. (Ord. 0510 Series 2018))

#### §7.04: INSPECTION; INSPECTION WARRANT:

A. Subject to the requirements and limitations of this section, the Marshal shall have the right to enter upon any Premises within the Town where marijuana is being grown, cultivated, or processed during reasonable hours for the purpose of conducting a physical inspection of the Premises to determine

compliance with the requirements of this Chapter. However, no agent or employee of the Town shall enter upon any Premises to conduct such an inspection without either the permission of the Owner or occupant, or without an inspection warrant issued pursuant to this section.

- B. If verbal permission to inspect the Premises from the affected Owner or occupant is not obtained, or if the Residential Structure is locked and the Marshal has been unable to obtain permission of the Owners or occupant, the Marshal may request that an inspection warrant be issued by the municipal court judge pursuant to rule 241 of the Colorado municipal court rules of procedure.
- C. n case of an emergency involving imminent danger to public health, safety, or welfare, the Marshal may enter any Premises within the Town to conduct an emergency inspection for the growing, cultivation, or processing of marijuana without a warrant and without complying with the requirements of this Section.
- D. The Board of Trustees declares that violation of this Article 7 involves a serious threat to the public safety or order within the meaning of rule 241(a)(1) of the Colorado municipal court rules of procedure.
- E. The municipal court judge may issue an inspection warrant authorizing the inspection of a Premises for the growing, cultivation, or processing of marijuana in accordance with rule 241(b) of the Colorado municipal court rules of procedure. Any inspection warrant issued pursuant to this section shall fully comply with the applicable provisions of rule 241 of the Colorado municipal court rules of procedure.
- F. The municipal judge may impose such conditions on an inspection warrant as may be necessary in the judge's opinion to protect the private property rights of the Owner of the Premises to be inspected, or to otherwise make the warrant comply with applicable law.
- G. It shall be unlawful and a misdemeanor offense for any Owner or occupant to deny the Marshal or other authorized person access to the Premises owned or occupied by such Owner or occupant if the Marshal or other authorized person presents an inspection warrant issued pursuant to this section.

#### §7.05: VIOLATIONS AND ENFORCEMENT REMEDIES.

- A. It is unlawful and a misdemeanor offense for a person to violate any provision of this Chapter. In accordance with subsection 1-4-2 of the Norwood Town Code, a person shall be guilty of a separate offense for each and every day during any portion of which any violation of the requirements of this Chapter is committed, continued, or permitted by such person. Any violation of this Chapter by any person may be prosecuted or referred for prosecution by the Town of Norwood for appropriate prosecution, criminal prosecution or otherwise, including without limitation pursuant to the Offenses Code at Title 5 of the Norwood Town Code and the General Penalty provisions at Title 1, Chapter 4, or other punishments, penalties, fine and/or punitive measures available at law or equity or as the Board of Trustees may establish by resolution or ordinance. In addition to any other lawful method of enforcement, which methods shall not be deemed exclusive, a violation of this Chapter is declared to be a public nuisance subject to abatement.
- B. The maximum fine for violation of Sections 7.03 A, B, C (2), (3), (4), (5) and D through R shall be \$500.00. For violation of Sections 7.03 C (1) or (4) the fine shall be assessed as follows:

Offense		Fine
7.03 C(1)	13 to 16	
	plants	\$500.00
7.03 C(1)	17-21	
	plants	\$1,000.00
7.03 C (1)		
	22-23	\$2,650.00
Plant counts in excess of 24 shall be referred to		
the District Attorney for prosecution		

(Ord. 0510 Series 2018)

C. Marijuana production prior to the adoption of these regulations that exceeded the maximum plant count limits set forth herein shall NOT be considered to be a legal pre-existing non-conforming use for purposes of C.R.S. § 30-28-120, this Chapter or otherwise. There will be no "grandfathering" of the number of plants that personal use/recreational or medical grows, Authorized Persons, Patients or Primary Caregivers already possess prior to the adoption of these regulations. (Ord. 0510 Series 2018)

#### APPENDIX "A"

# TOWN OF NORWOOD ZONING DEVELOPMENT PERMIT

Compliance with applicable Town Land Use Code criteria for each of the categories listed in Sections 3 and 4 must be verified by the Town Planner and Building Official prior to consideration of applications for building permits.

### 1. APPLICANT SHALL COMPLETE:

Property Owner	Mailing Address	Phone
Subject Property Street	Address	
Legal Description; or	Attached	
Existing Use(s)		
Proposed Use(s)	Squa	are Footage
Lot Size Proposed	d Bldg. Height No. Off-Stre	eet Parking Spaces
Proposed Setbacks:	Front Side	Rear
Water Commission App	proval? Sanitation District	Approval?
knowledge and that I undevelopment, any and a Commissioners and all applicable provisions of	derstand all provisions of C ll conditions placed upon the information requested by the County and/or State codes is imposed by the County, in	he above is true and accurate to the best of my County and State codes applicable to the proposed he proposed development by the Board of County his document. I also understand that if I violate any s, I may be required to remedy such violation(s) through including moving or removing structures and ceasing of
Signature of Applicant	Date	

#### 2. APPLICANT SHALL ATTACH COMPLETE SITE AND ACCESS PLAN

# 3. TOWN PLANNER SHALL VERIFY COMPLIANCE WITH: Application Completeness \_\_\_\_\_ (If incomplete, return to applicant) Zoning \_\_\_\_\_ Cond. Use Permit #\_\_\_\_\_ List Conditions of Approval or Attached: Should Bldg. Permit Application Be Considered? YES\_\_\_\_\_\_ NO\_\_\_\_ Comments Signature of Town Planner Date 4. BUILDING OFFICIAL SHALL VERIFY COMPLIANCE WITH: Use \_\_\_\_\_ Height \_\_\_\_ Setbacks \_\_\_\_ Access \_\_\_\_\_ Driveway width \_\_\_\_\_ Parking Spaces \_\_\_\_\_ Water Commission Approval \_\_\_\_\_ Sanitation District Approval \_\_\_\_\_ Can Building Permit Be Applied For? YES\_\_\_\_\_\_ NO\_\_\_\_\_ Comments

5. IF THE BUILDING OFFICIAL SO AUTHORIZES, APPLICATION(S) FOR A BUILDING PERMIT THAT CONFORM MATERIALLY WITH ALL INFORMATION PROVIDED ABOVE MAY BE SUBMITTED TO THE BUILDING DEPARTMENT.

Signature of Building Official Date